

University of North Dakota UND Scholarly Commons

Theses and Dissertations

Theses, Dissertations, and Senior Projects

January 2014

Statistical Analysis Of The Effectiveness Of The 'You're Hired!' Program At Changing Students' Attitudes Towards Engineering

Kristin Michelle Brevik

Follow this and additional works at: https://commons.und.edu/theses

Recommended Citation

Brevik, Kristin Michelle, "Statistical Analysis Of The Effectiveness Of The 'You're Hired!' Program At Changing Students' Attitudes Towards Engineering" (2014). *Theses and Dissertations*. 1624. https://commons.und.edu/theses/1624

This Thesis is brought to you for free and open access by the Theses, Dissertations, and Senior Projects at UND Scholarly Commons. It has been accepted for inclusion in Theses and Dissertations by an authorized administrator of UND Scholarly Commons. For more information, please contact zeinebyousif@library.und.edu.

STATISTICAL ANALYSIS OF THE EFFECTIVENESS OF THE 'YOU'RE HIRED!' PROGRAM AT CHANGING STUDENTS' ATTITUDES TOWARDS ENGINEERING

By

Kristin Michelle Brevik Bachelor of Science, Minnesota State University Moorhead, 2011

A Thesis

Submitted to the Graduate Faculty

of the

University of North Dakota

In partial fulfillment of the requirements

for the degree of

Master of Science

Grand Forks, North Dakota

December 2014

This thesis, submitted by Kristin Brevik in partial fulfillment of the requirements for the Degree of Master of Science in Chemical Engineering from the University of North Dakota, has been read by the Faculty Advisory Committee under whom the work has been done and is hereby approved.

Dr. Frank Bowman, Chairperson

Dr. Robert Wills

Dr. Brian Tande

This thesis is being submitted by the appointed advisory committee as having met all of the requirements of the School of Graduate Studies at the University of North Dakota and is hereby approved

ne

Dr. Wayne Swisher Dean of the School of Graduate Studies

December 8, 2014

Date

PERMISSION

Title	Statistical Analysis of the Effectiveness of the 'You're Hired!' Program at Changing Students' Attitudes towards Engineering	
Department	Chemical Engineering	
Degree	Master of Science	

In presenting this thesis in partial fulfillment of the requirements for a graduate degree from the University of North Dakota, I agree that the library of this University shall make it freely available for inspection. I further agree that permission for extensive copying for scholarly purposes may be granted by the professor who supervised my dissertation work, or, in his absence, by the Chairperson of the department or the dean of the School of Graduate Studies. It is understood that any copying or publication or other use of this dissertation or part thereof for financial gain shall not be allowed without my written permission. It is also understood that due recognition shall be given to me and to the University of North Dakota in any scholarly use which may be made of any material in my thesis.

Kristin Michelle Brevik

November 20, 2014

TABLE OF CONTENTS

LIST OF FIGURESix
LIST TABLESxi
ACKNOWLEDGEMENTS xiv
ABSTRACTxvi
CHAPTER
I. INTRODUCTION1
1.1 Scope of this Research
1.2 Thesis Overview4
II. RESEARCH-BASED DESIGN OF K-COLLEGE STEM PROGRAMS
2.1 Why a Program is Needed5
Changing Attitudes with Authentic Experience9
2.2 Engineering Education Program Design10
Limitations of Current Engineering Education Programs10
Learning from Effective Programs11
Common Attributes of Effective Engineering Education Programs14
Active Learning/Inquiry-Based Learning14
Curriculum Standards Aligned15
Engaged Teachers/Role Models16

17
18
18
19
20
20
21
22
24
25
25
26
27
27
28
29
31
31
36
36
37
37

	Construct a Prototype	38
	Test and Evaluate Solution	38
	Redesign if Necessary	39
	Communicate the Solution	39
IV.	RESEARCH METHODOLOGY	47
	4.1 Institutional Review Board (IRB)	48
	4.2 2012-2013 You're Hired! Program	49
	Data Collection	49
	4.3 2013-2014 You're Hired Program	53
	Redesign	54
	4.4 Data Analysis	56
	Removing Identifiers	56
	Data Analysis	56
V.	RESULTS & DISCUSSION	59
	5.1 School A – Analysis of Experimental and Control Group Responses	60
	5.1.1 Engineer Question	61
	Engineer Question Discussion	65
	5.1.2 Interest Question	67
	Interest Question Discussion	71
	5.1.3 Attitude and Skills Question	73
	Attitude and Skills Question Discussion	75
	5.2 Experimental Group Results for the Five Participating vi	

	Schools
	Engineer Question Discussion
	5.2.2 Interests Question79
	Interest Question Discussion80
	5.2.3 Attitude and Skills Question
	Attitude and Skills Question Discussion82
	5.3 Experimental Group Responses When Comparing Each of the Five Schools to Each Other
	Section 5.3 Discussion
	5.4 Analysis Based on Gender
	5.4.1 Engineer Question90
	Engineer Question Discussion
	5.4.2 Future Work Question92
	Future Work Question Discussion
	5.4.3 Interest Question95
	Interest Question Discussion
	5.4.4 Attitude and Skills Question
	Attitude and Skills Question Discussion103
	5.5 Impact of the 'You're Hired!' Program104
VI.	CONCLUSIONS & FUTURE WORK108
	6.1 Conclusions108
	6.2 Future Work

	Redesign of the Pre-Survey and Post-Survey	112
	Professional Development for Teachers	120
	Revisiting Program Design Key Features	122
	Preparing for Scaling, Sustaining and Evolution of the Program	123
	Final Thoughts	125
APPENDICES		126
BIBLIOGRAPHY		174

LIST OF FIGURES

Figure	Page
4.1.	Oil Spill Problem Statement
4.2	The Engineering Design Process
4.3.	The Boardroom Sign-Up Sheet42
4.4.	Peer/Self-Assessment Feedback
5.1.	Comparison of Mean Values for School A's Experimental and Control Group Responses for Engineer Statements on Pre-and Post-Surveys
5.2.	Comparing Control vs. Experimental Responses for Interest Questions68
5.3.	Comparing Control vs. Experimental Responses for Attitude and Skills Questions
5.4.	Comparing All-School Experimental Responses for Engineer Questions77
5.5.	Comparing All-School Experimental Group's Responses for an Interest Statement, In10 – Science is a difficult Subject
5.6.	Comparing All-School Experimental Group's Responses for an Attitude and Skills Statement, A2 – I can use what I know to Design and Build Something Mechanical that Works
5.7.	Comparing Experimental Responses for Questions that were not Significant for Both School A and School B – School E Experimental Groups
5.8.	Comparing Female and vs. Male Responses for Engineer Statements
5.9.	Comparing Female and vs. Male Responses for Future Work Statements
5.10.	Comparing Female and vs. Male Responses for Interest Statements

5.11.	Comparing Female and vs. Male Responses for Attitude and Skills Statements	
5.12.	Student Levels of Agreement to the Statement, 'How much did Participating in the Activity Impact Each of the Following?	

LIST OF TABLES

Table	Page
4.1. Electronic Pre-Survey ar Response Options	nd Post-Survey Questions by Category with
5.1. Statements Showing Sta	tistically Significant Differences in Responses
between either Experime	ental and Control Groups for either Survey or
between Pre- and Post- F	Responses for Either Group for School A61
5.2. Statistical Comparison o	f Pre- and Post-Survey Responses to Engineer
Questions for Control an	d Experimental Groups at School A63
5.3. Statistical Comparison o	f Control and Experimental Responses to
Engineer Questions for I	Pre- and Post-Surveys at School A64
5.4. Statistical Comparison o	f Pre- and Post-Survey Responses to Interest
Questions for Control an	d Experimental Groups at School A69
5.5. Statistical Comparison o	f Control and Experimental Responses to
Interest Questions for Pr	e- and Post-Surveys at School A70
5.6. Statistical Comparison o	f Pre- and Post-Survey Responses to Attitude
and Skills Questions for	Control and Experimental Groups at School A74
5.7. Statistical Comparison o	f Control and Experimental Responses to
Attitude and Skills Ques	tions for Pre- and Post-Surveys at School A74
5.8. Statements showing Sign	nificantly Different Responses for the All-
Schools Experimental G	roup between Pre-Survey and Post-Survey76
5.9. Statistical Comparison o	f All-School Experimental Group's Responses
to Engineer Questions for	or Pre- and Post-Surveys at all Participating
Schools	
5.10. Statistical Comparison o	f All-School Experimental Group's Responses
to Interest Questions for	Pre- and Post-Surveys at all Participating
Schools	

5.11.	Statistical Comparison of Experimental Group's Responses to Attitude and Skills Questions for Pre- and Post-Surveys at all Participating Schools
5.12.	Statements showing Significantly Different Responses between Pre- Survey and Post-Survey for Section 5.1 Experimental Group Compared and Section 5.2 Experimental Groups
5.13.	Statistical Comparison of Experimental Responses for Questions that were not significant for Both School A and School B – School E Experimental groups
5.14.	Statements Showing Statistically Significant Different Responses between either Female and Male Responses for Either Survey or between Pre- and Post- Responses for Either Gender for the All-School Experimental Group
5.15.	Statistical Comparison of Female and Male Responses to Engineer Statements for all Participating Schools90
5.16.	Statistical Comparison of Pre- and Post-Survey Responses to Engineer Statements for Females and Males at All Participating Schools
5.17.	Statistical Comparison of Female and Male Responses to Future Work Statements for All Participating Schools
5.18.	Statistical Comparison of Pre- and Post-Survey Responses to Future Work Statements for Females and Males at All Participating Schools
5.19.	Statistical Comparison of Female and Male Responses to Interest Statements for All Participating Schools
5.20.	Statistical Comparison of Pre- and Post-Survey Responses to Interest Statements for Females and Males at All Participating Schools
5.21.	Statistical Comparison of Female and Male Responses to Interest Statements for All Participating Schools
5.22.	Statistical Comparison of Pre- and Post-Survey Responses to Attitude and Skills Statements for Females and Males at All Participating Schools
5.23.	Descriptive Statistics of Responses to Impact Statements for Females and Males

5.24.	Statistical Comparison of Responses to Impact Statements for Females and Males	106
5.25.	Breakdown of how School A – School E Responded to the Question, 'Would You Recommend that your Friends Participate in this Activity?'	107
6.1.	Original Categories of Questions asked by the AWE Pre-Survey for Engineering Activities	114
6.2.	Redesign of the Categories of Questions asked by the AWE Pre-Survey for Engineering Activities	117
6.3.	Redesigned Categories for New Pre-Survey with Revised as well as Additional Questions/Statements	117

ACKNOWLEDGEMENTS

I am thankful for my committee members Dr. Frank Bowman, Dr. Robert Wills, and Dr. Brian Tande for their support of my research topic for this thesis. In particular, I would like to thank Dr. Bowman for agreeing to be my advisor, allowing me to research a topic that I am passionate about, for always supporting the quality and value of work being done by this research, and for all of the constructive comments and remarks he has provided throughout this process. I truly am grateful for the guidance, support and encouragement Dr. Bowman has given me throughout my time as a student at UND, especially throughout the thesis process.

In addition, I would like to thank Dr. Wills for not only agreeing to serve as a member of my committee, but for investing himself in this research in such a way that he strives to advance the work I have done to improve on this research topic. I would also like to thank Dr. Tande for his ongoing support of the research project over the past few years.

I am very thankful to have had such a great mentor throughout this process. Dr. Kristi Jean not only has provided support and guidance, but has challenged my ideas and approaches. It is through this mentorship that I have been able to truly understand the scope and significance of the work within this research and it has made me confident what I have completed. I would like to particularly thank my family and fellow classmates at UND who have supported me throughout this entire academic process. Thank you to my family for providing continuous love, support, and the encouragement to just keep trying. To my classmates who spent countless hours with me working to solve problems, studying for exams, and taking the time for a cup of coffee and conversations, all I can say is thank you. To my family, this journey would have been nearly impossible without your constant love, encouragement, and support.

ABSTRACT

With the growing need for qualified employees in STEM-based careers, it is critical to develop activities for middle and high school students to increase their awareness of opportunities in these areas. With proper design, increasing awareness of STEM-based careers in conjunction with overcoming current stereotypes can lead to a change in attitudes towards these various careers. As part of this research project I have developed 'You're Hired!', a program providing middle and high school students a hands-on, authentic experience in various engineering roles while assessing the change in a student's attitude towards the engineering profession. Project design also incorporates an opportunity in which students can hone their 21st Century Skills such as collaboration, critical thinking and time management, also known as engineering mindset or workforce skills.

'You're Hired!' is a series of three STEM-based projects, given over the course of a school year, that requires students to work as a 'company' for an entire school day to find a solution to a relevant, present-day problem. At the end of each project, the students communicate their solution to a community-led boardroom, comprised of school board members, community stakeholders and local industry representatives. The 'You're Hired!' program is designed to immerse students in an authentic real-world experience that incorporates the engineering design process and 21st Century Skills. The program also tracks student progress in these areas throughout the year using peer- and selfassessments.

This research project used both quantitative and qualitative data collection methods to measure the impact of the 'You're Hired!' program. The methodology includes comparing a control group to an experimental group to further understand the benefit of the program. Other factors, such as gender and school setting, were also analyzed to determine program impact. The results of the statistical analysis show there is a significant difference in the change in a student's attitude toward engineering when participating in the program.

CHAPTER I

INTRODUCTION

For some students the closest thing to engineering they are introduced to in K-12 schools is physics class, and even that may not be offered until the student is a junior or senior. The lack of understanding of what engineering was, and association between the word engineering and extremely hard were the key factors in what made me decide to pursue a bachelor's degree in physics, not engineering. The goal of this research is to determine if the 'You're Hired!' program is effective at changing students' attitudes towards engineering. 'You're Hired!' is an engineering education outreach program for middle and high school students that was designed to change students attitudes about engineering so that students could better understand the type of work, the variation of possible jobs and the impact on the world that engineers have.

One key piece of information I believe to lacking from most schools in the immediate area of this research is not only that students can choose engineering as a career opportunity, but the diversity of engineering that can be chosen from.

William Wulf states that his favorite definition of engineering is "design under constraint." (Wulf, 2006) Wulf goes on to state that engineers design things to solve problems, but not just any design will work. The design must satisfy a long list of constraints such as cost, size, manufacturability, environmental impact, ect. (Wulf, 2006) Engineers must be creative in the way that they solve problems to provide the best possible solution. My goal for the 'You're Hired!' program is that it provides the opportunity for students to understand engineering as this definition; not that engineering is too hard. This project will measure the effectiveness of the 'You're Hired!' program doing just that.

For some schools the 'You're Hired!' program is a supplementary educational tool to an already STEM-focused curriculum. To other schools this program is its STEMcurriculum. For a number of schools that have been a part of the program, 'You're Hired!' may have provided the first opportunity in which students have talked about engineering at school.

In addition to providing exposure to engineering, other STEM careers such as technicians, web designers, etc. will also be highlighted. Not every student will want to be an engineer. The program is not designed to ensure all students decide to be engineers, it is designed to increase attitudes and understanding of a career that currently still has students thinking that engineers only drive trains or work on cars. If the program could provide understanding to more people, then when an individual is interested in pursuing a career option like engineering, their support system of friends/family would be able to understand and encourage.

One outcome I am aiming for within the scope of this project long term is to assist in creating more diversity in local engineering bachelor's degree programs. I want the program to reach students who enjoy math and science so that they see engineering as a combination of their interests in a fun, engaging way. This program isn't necessarily aimed at the students who already think they want to be engineers, instead it is focused

on the 'out-of-the-box' thinkers that don't know, because they haven't been reached, that engineering is a great option for them.

1.1 Scope of this Research

This project consists of two main parts: 1) development of a program that is designed to change students' attitudes and understanding of engineering, and 2) a research study to determine the effectiveness of the program based on data from participants. The 'You're Hired!' program that was developed is designed achieve the following goals: 1) introduce a variety of engineering careers possible with an engineering degree, 2) increase students' attitudes and understanding of engineering by incorporating features that have been previously researched and are proven to show impact, 3) introduce teachers to incorporating STEM and 21st Century Skills into their classrooms with minimal additions to their already busy workload, 4) hone students' 21st Century Skills also known as engineering mindset skills or workforce skills, and 5) engage industry professionals to participate and provide feedback as well as being a positive role model for students.

For this project there are two items the research study assessed: 1) change in students' attitudes and understanding of engineering, and 2) impact of the program on students' self-assessment about their own abilities and ambitions. The study did not assess improvement of engineering design mindset or skills, influence on teachers, students changing career choices, or the amount of impact the industry professionals had on students. These items were beyond the scope of the current research and could be assessed in future studies. The research study used student survey data to assess program effectiveness. Data analysis considered two primary factors:, 1) comparing data based on

school, and 2) comparing data based on gender. Information on these two factors, gender and school, is readily accessible from the pre- and post-surveys the students were given. Ethnicity was briefly looked into for analysis, however with the region that this project took place in, 85% of individuals participating in the activities classified themselves as 'White Americans'.

1.2 Thesis Overview

This thesis describes both the 'You're Hired!' program I helped develop and a research study into whether or not student attitudes and understanding of engineers shifted from pre- to post-survey as well as program impact.

Chapter 2 reviews the research based design of past/current K-college level STEM programs. An overview of a variety of programs as well as discussion of why these programs are beneficial to students can be found in this chapter. Chapter 3 describes the program design of 'You're Hired!'. Explanation of how 'You're Hired!' fits into the school-year calendar as well as what happens during a typical activity is discussed. Program features of 'You're Hired!' are explained and linked back to research based design programs from Chapter 2. Chapter 4 covers the research methodology behind the project. This chapter explains the methods used to gather information during the study and the analysis used to develop conclusions based on data. Chapter 5 explains the results of the program and the discussion of those results. In each section, statistical analysis is presented along with discussion of the results. Chapter 6 finalizes the document with concluding statements as well as proposed future work topics and considerations.

CHAPTER II

RESEARCH-BASED DESIGN OF K-COLLEGE STEM PROGRAMS 2.1 Why a Program is Needed

In the 21st Century workforce employers are looking for potential employees that can solve a range of intellectual and technical problems regardless of the position. (Marzano & Heflebower, 2012) Employers will be looking for employees that are willing to use an engineering mindset in which they problem solve, think critically, collaborate with others, and are able to solve elementary issues as well as more complex problems that may arise. In other terms, employers are looking for employees that have well developed 21st Century Skills. Today's workplace is very different than it was 10-20 years ago. Mastery of the basic skills in education such as reading, writing, and math is no longer enough. (Marzano & Heflebower, 2012) Schools need to place more emphasis on the 21st Century Skills such as critical thinking, problem solving, time management, etc. in addition to the basics.

While not always pictured this way, engineering is a very creative profession. It is important that engineers be able to design a solution while meeting constraints such as time, cost, manufacturability, size, etc. (Wulf, 2006) Without diversity within the profession, the same types of solutions will continue to be the only solutions. It takes a variety of people from various backgrounds to help solve the larger problems like renewable energy sources that work in all areas of the world, or a process that can inexpensively bring clean water to those countries that are less fortunate, or even an

individualized identification system that would make identity theft nearly impossible. The engineering design team needs to be diverse and creative in order to continue to excel and meet the needs of the society they are impacting. Engineering touches almost every piece of the world we live in, from an online video chat to an artificial heart valve. It takes unique ideas and brainstorming to design and redesign products for the future.

While engineering careers are in high-demand within a wide array of industries, the general public has only a limited perspective of what these careers really are. (Kimrey, 2013) The images/stereotypes students have about engineering frequently identify engineers as car mechanics, construction workers or train operators. (Knight & Cunningham, 2004) Other stereotypes include: engineering is boring, engineers work alone or at a computer, and engineers help society. (Knight & Cunningham, 2004) (Fussell Policastro, 2009) Research has linked K-12 students' limited knowledge and/or negative image of engineering careers to the shortage in the number of college graduates receiving degrees in engineering. (Committee on Public Understanding of Engineering Messages and National Academy of Engineering, 2008) (Hill, Corbett, & St. Rose, 2010)

Not only is there a limited knowledge of what engineering is, there is limited knowledge in general in terms of students being prepared to take the classes needed to pursue a degree like engineering. In one study, roughly 300 2-year and 4-year college instructors were interviewed and stated that high school did not prepare their students to comprehend complex materials, think analytically, or apply what they learned to solve problems. (Marzano & Heflebower, 2012) It may not come as a shock to these instructors that 40% of all students that enter college need to take remedial courses in addition to their major requirements. (Marzano & Heflebower, 2012)

In order to properly prepare the students of the United States of America for a successful future, President Obama has launched an initiative called Educate to Innovate which calls out two things that need to happen in order to keep the United States as a leader of innovations. The first point is that students, including women and minorities, need to be *prepared* and proficient in STEM (Science, Technology, Engineering, and Math) subjects in order to be ready for the future workforce. The second point is that students need to be *inspired* and motivated in STEM activities so that they can be encouraged to pursue a STEM career. (The President's Council of Advisors on Science and Technology, 2010)

Whether students decide to pursue a STEM career or not, they should have the knowledge, conceptual understandings, and critical-thinking skills that come from studying STEM subjects. (The President's Council of Advisors on Science and Technology, 2010) While the 21st Century workforce cannot function without the appropriate advances made in science and technology, it also cannot function without advances in education, business practices etc. There are numerous careers that do not require the use of math and science specifically.

Research has been done regarding the importance of implementing more STEM as well as experiences in which students can hone their 21st Century Skills into K-12 classrooms.

STEM learning environments provide opportunities for students to focus on discovery and the ability to solidify educational concepts in their minds by being able to understand a concept in a variety of contexts. A STEM learning environment has been described as one in which there are no barriers between the various classes the students

take. All subjects become pieces to a bigger picture in terms of student understanding. Students that are actively learning in a STEM environment would not ask why they need to know about citations because they would be using them not only in English, but also understanding the need to use them in science, math and history. Students in geography would not ask why they have to know about Africa, because in science they are creating water purification prototypes, in gym they are walking a mile carrying full jugs of water and in English they are reading 'A Long Walk to Water' by Linda Sue Park. A true STEM environment has student engaged in all aspects of a topic that include all core classes that are standard across the nation.

In most schools, the E in STEM is left out completely. All schools have science and math teachers and many have technology education teachers (tech ed.), however engineering tends not to make it into most schools consistently. This leads to students' images of engineering stemming from a variety of other outside sources such as: knowing engineers in their community, career expositions, friends or family members, class trips, guest speakers or media. (Yurtseven, 2002) (Young, 2007) Unfortunately, these current methods of educating K-12 students about engineering and engineering careers have shown little impact on improving the number of students pursuing an engineering related degree, with the percentage of U.S. bachelor's degrees awarded in engineering remaining constant around 4.5% over the past decade, down from an average of 7.1% during the 1980's. (Science and Engineering Degrees: 1966-2010, 2013) Of the students who took the ACT college readiness assessment in 2013, only 23% expressed an interest in STEM careers. (The Condition of STEM 2013 National, 2014) The ACT assessment categorized STEM as four key areas: science, computer science & mathematics, medical & health, and engineering & technology. (The Condition of STEM 2013 National, 2014) A series of questions about STEM related activities was also asked. The ACT defines *expressed interest* as students stating that they are interested in STEM. *Measured interest* was defined as interest shown regarding STEM ideas and activities, while when the student was asked specifically about STEM careers did not express interest. Of the 54% of students that took the ACTs, 9% answered the STEM related questions in such a way that it is possible that the student is interested in a potential STEM career, even though when asked specifically the student said no. (The Condition of STEM 2013 National, 2014)

Changing Attitudes with Authentic Experience

One of the goals of the 'You're Hired!' program is to introduce a variety of engineering careers possible with an engineering degree. Another goal is to increase students' attitudes and understanding of engineering. My theory is that while not every student will want to be an engineer, after participating in the 'You're Hired!' program more students will have a better understanding of the work engineers do. In order to meet the national engineering demand with more students choosing the engineering profession, students must first have a positive image of engineers and the engineering profession. 'If you want to change how people behave, you first have to change how they think.' (Patterson, Grenny, Maxfield, McMillan, & Switzler, 2007) Recognizing that the engineering profession has had the same image for numerous years, innovative approaches must be taken to change this persistent and resistant view by providing personal experiences in which participating students can see first-hand the variety of work that is engineering. (Bandura, 1977) As Benjamin Franklin stated, "Tell me and I forget, teach me and I may remember, involve me and I learn." (Franklin) By giving students the opportunity to learn about engineering through hands-on learning, the program is increasing the likelihood that a student will pursue a degree in engineering or related STEM field.

2.2 Engineering Education Program Design

Limitations of Current Engineering Education Programs

There are numerous programs aimed at educating K-12 students about what engineers do. Most traditional programs fit into the following categories: competitions, summer schools/camps, speaker programs, site visits/tours, career advisers and teacher resources. (Dawes & Rasmussen, 2006) While these traditional programs have high impact rates with the participating students, there are some major limitations that need to be considered. One of the largest limitations with current methods of educating students about engineering has to do with the students that are impacted. Sometimes the students that could greatly learn and benefit from an activity or experience are students that are not exposed. One major limitation in the current programs to educate students is that every student is not personally invested in the program.

While it may appear beneficial to bring an engineer into the classroom to spend time with the students and talk about their profession, many students are often left with only a slight understanding of the guest engineer's career and fail to see themselves in the same role as that engineer. (Eniola-Adefeso, Fall 2010) Another current method of introducing students to engineering is field trips to see the kind of environment an engineer works in. Similar to the classroom visit, this type of exposure gives students a look into only one of the many types of jobs engineers have. Students could leave an industry tour thinking that all engineering is dirty or they could leave thinking that engineers only work on roads and bridges. It is also possible that engineering is not mentioned in their classroom after that one visit or field trip.

Summer camp is another current tool to get students interested in engineering. Summer camps tend to have a higher impact on students' understanding of engineers and engineering careers which would suggest that they are effective, however given the number of students camps reach, this type of approach, while greatly beneficial, only confirms that an interested student is indeed wanting to become an engineer. The summer camp method of outreach completely misses the students that don't understand what engineering is, meaning that it can potentially miss the students that could benefit most.

More effective methods of generating student interest in and understanding of engineering are needed as well as a tool that will assist students in the ability to hone their 21st Century Skills which are closely aligned with engineering attributes – qualities of an engineer.

Learning from Effective Programs

Research into K-college engineering programs has identified a variety of elements that contribute to effective programs. For example, a review by researchers from Old Dominion University and Colorado State University found that effective pre-engineering activities, such as robotics club and engineering camp, include what they term 'key attributes' that positively influence students to successfully pursue and complete an engineering degree: self-motivation, use of problem-solving strategies, use of computer applications, and immediate feedback on success of efforts. (Fantz, Siller, & DeMiranda, 2011) Other researchers have highlighted 'key design features' that offer engagement

with engineering: the 'wow' factor, simple yet effective ideas, social responsibility, potential for world-wide impact, and personal relevance. (Marshall, McClymont, & Joyce, 2007) For programs that impact all students, not just a select group that joins a club or camp, additional common 'themes' regarding engineering education arise including: active learning/inquiry-based learning through hands on activities, interdisciplinary approaches which add a technological component, teacher engagement, curriculum supplements that connect to standards, and creating mentorships and partnerships that make engineers "cool". (Dawes & Rasmussen, 2006) (Jeffers, Safferman, & Safferman, 2004) Work done by one successful program, Engineering is Elementary, has led to a list of inclusive 'principles' as a starting point for resource design. The Engineering is Elementary principles include many of the attributes listed above for other programs. (Cunningham & Lachapelle, 2012)

Including these key attributes, factors, themes, and principles should theoretically increase the likelihood of positively impacting a broad range of students, as different individual students will respond best to different program elements. It is thought that changing attitudes towards engineering on a large scale will require that multiple influence strategies and project attributes be incorporated.

More recently, a greater emphasis has been put on program designs that offer a tool for schools to use in which the program is integrated into current curriculum and every student participates. (Dawes & Rasmussen, 2006) (Cunningham & Lachapelle, 2012) (Clarke & Dede, 2009) (Mooney & Laubach, 2002)

One innovative approach directly involves students in a personal, authentic experience where they take the role of an engineer. Such an experience will change a student's attitude and behavior towards engineering if the student can answer 'yes' to two questions: 1) can I do it? and 2) what is in it for me? (Patterson, Grenny, Maxfield, McMillan, & Switzler, 2007) (Bandura, 1977) (Fantz, Siller, & DeMiranda, 2011)

Some programs are designed to be implemented locally, while others are designed for scaling and sustaining across broader regions to impact a greater number of students. Regardless of how the program is implemented there are attributes or common approaches that have been adopted by a majority of these programs, at least in some combination. The adoption of these approaches is supported by the statistically significant effect the previous programs have had regarding the attitudes, understanding and behaviors of students towards engineering.

In applying effective research design from the literature review to the design of the 'You're Hired!' program, the many proven attributes, factors, themes and principles were first summarized as a set of common program attributes and common individual activity attributes. Common program attributes describe the key features that are incorporated into the overall program, which is comprised of a variety of individual activities. Common activity attributes are the key features that assist in getting participants engaged in the individual activities and allow for the best learning environment. Common activity attributes do not look past the individual activity in terms of projected impact. However in the design, building upon the skills acquired during the individual activities plays into the overall program attributes.

Common Attributes of Effective Engineering Education Programs

The common attributes of effective K-12 engineering education programs can be broken down into 5 categories: active learning/inquiry-based learning, curriculum standards aligned, engaged teachers/role models, appropriate range of student ages, and partnerships/community involvement. (Jeffers, Safferman, & Safferman, 2004) (Davis, Yeary, & Sluss, 2012) (Cunningham & Lachapelle, 2012) (Dawes & Rasmussen, 2006) *Active Learning/Inquiry-Based Learning*

Engineering is a hands-on field, so in order to teach students what engineering is, students must do some sort of engineering - and do so in a hands-on, inspiring, active learning environment. (Jeffers, Safferman, & Safferman, 2004) (The President's Council of Advisors on Science and Technology, 2010) It is not enough to be lectured about engineering. Most outreach programs recognize this and have incorporated active learning into their programs. These active learning/inquiry-based learning environments allow students to explore a problem using previously learned principles. These types of learning environments focus less on theory and more on context-based materials that students can relate to on a personal level. (Dawes & Rasmussen, 2006) Activities that focus on and emphasize observation as well as an exploration through experimentation can facilitate the inquiry that is necessary for students to be able to focus their thinking in addition to the ability to form new ideas and pose new questions. (Jeffers, Safferman, & Safferman, 2004)

There are a number of students that will thrive in a hands-on, active learning environment; some of these students may be individuals that do not perform well in a traditional classroom setting. It is important to understand and recognize that students as a whole do not conform to one type of learning style. Recent research has highlighted that in order to reach the greatest number of students, multiple learning environments need to be incorporated in a classroom to accommodate the various learning styles students within that classroom have. (Dawes & Rasmussen, 2006) While some students thrive in a traditional classroom in which the teacher lectures and students complete in class worksheets, homework and tests where the students reiterate information in a similar context to what was taught. Other students thrive in situations where they can try out principles they have learned and see first-hand the laws of gravity or the accuracy of probability calculations. While the best method of teaching relies heavily on the students within that classroom, in order to facilitate the most learning possible, it is important for teachers to incorporate these various learning styles within their teaching.

Numerous schools across the United States have incorporated active learning/inquiry-based learning into their classrooms as an additional style of learning/teaching. More emphasis still needs to be placed on this type of instruction if students want to be ready for the 21st Century workforce. As explained by Wagner et al, in the 21st Century, it will not be enough for students to master skills in reading, writing and math. Almost any job that pays more than minimum wage – blue or white color – will require employees that can solve a range of problems and think intellectually to do so. (Wagner, 2008) (Marzano & Heflebower, 2012)

Curriculum Standards Aligned

It is important for engineering education programs to provide curriculum supplements so that teachers can integrate fun and exciting examples of engineering while tying back to curriculum standards. (Jeffers, Safferman, & Safferman, 2004) (Dawes & Rasmussen, 2006) The goal of engineering education programs is not only to supplement current math and science curricula, but to also include other curricula subjects such as history, geography, writing, and reading - to provide a true integrated STEM learning experience for both the teachers and the students. (Jeffers, Safferman, & Safferman, 2004)

Including teachers in the development process of new engineering education programs and allowing them to help with curriculum alignment can be very beneficial. Teachers can quickly see that they can be engaged and incorporate these engineering lessons while still meeting other curriculum standards from the basic courses such as math and science. (Dawes & Rasmussen, 2006) Because engineering generally blends math and science together, teachers quickly realize that one engineering lesson can meet the criteria of multiple curriculum requirements from a variety of subjects.

Engaged Teachers/Role Models

One way that teachers are included and engaged in engineering education is the partnership with engaged role models such as engineering professors from local universities. Due to time constraints, it is often difficult for professors to be present in multiple classrooms. Engineering students have been used as professor replacements in which engineering students work with a classroom including the teacher on activities and get the students excited about engineering. (Jeffers, Safferman, & Safferman, 2004)

Engineering students make great role models for children in schools because it is often a way for them to see a diverse group of people with different backgrounds all passionate about the same thing – engineering. In one successful engineering education program, Engineering is Elementary, researchers found that it is best to have a diverse group of role models for students to relate to. A variety of role models – of both genders from a variety of races and ethnicities, with different strengths and disabilities, and that engage in a variety of hobbies and extracurricular activities create a broad picture of the types of people that make great engineers. (Cunningham & Lachapelle, 2012)

Appropriate Range of Student Ages

Most engineering education programs understand that it is important to get engineering exposure to students as soon as possible. One researcher goes as far to state, 'young children are inherently active with strong impulses to investigate, share with others what they have discovered, to construct things, and to create. In other words, young children are natural engineers.' (Genalo, Bruning, & Adams, 2000) (Jeffers, Safferman, & Safferman, 2004) It is not until children reach school where they split into specific classes such as English, math, science, and history that students start to lose interest in math and science because it becomes difficult to see how the 'facts' relate to their everyday life.

While many original engineering education programs were designed to work with high school students in hopes to increase enrollment numbers at engineering universities, the current focus is on younger students. Some programs choose to focus on elementary level students; others focus on middle school to early high school aged students. Exposing younger students to engineering will give the students an early opportunity to try out engineering concepts. If the young students enjoy the experience they will have a reason and motivation for taking more difficult and intimidating math and science classes in high school and in college. (Price, 2000) (Jeffers, Safferman, & Safferman, 2004)
Partnerships/Community Involvement

In addition to local universities supplying support to teachers in the form of engineering professors and engineering students, Dawes expresses in his research that partnerships in which the community is involved in the students' education are important. (Dawes & Rasmussen, 2006)

In some situations, partnerships can create a valuable incentive for businesses within a community to take interest in educational content. If teachers could build an educational project that revolves around a specific business such as a power company or a water purification plant, it provides an opportunity for teacher, students, and families to understand more about the businesses within their community. The students will also benefit greatly by the ability to make the connection between what they are learning in school and how it applies not only to a real world application, but one that specifically impacts their immediate community.

Common Attributes of Activities that Lead to a Change in Student Attitude and Understanding

The attributes of individual engineering education activities affect how well students perceive and accept the information presented. This can vary from program to program. There are a variety of primary attributes that appear to enhance program effectiveness. Those primary attributes are real-world problems, hands-on active learning, application of problem solving skills, computer/technology applications, and feedback on student efforts. (Dawes & Rasmussen, 2006) (Jeffers, Safferman, & Safferman, 2004) (Cunningham & Lachapelle, 2012) (Fantz, Siller, & DeMiranda, 2011)

Real-World Problems

In order to increase the number of students that choose to pursue engineering as a future career opportunity, students must first see the connection between engineering and the world around them. "Many students who are academically competent in the school subject matter ultimately view school's knowledge and skills as irrelevant for their future career and/or everyday lives." (Cunningham & Lachapelle, 2012) Therefore to increase the number of students pursuing engineering and the appreciation for engineering by all students, not only must students see the relevance of what they are learning in context to the real-world, but they need to be given the opportunity to explore and to discover the possibility of them personally filling such roles in the future.

Research indicates that students learn concepts and skills through experience. (Cunningham & Lachapelle, 2012) "Children's learning is more profound when they engage in realistic disciplinary practices." (Cunningham & Lachapelle, 2012) As an adult applying for a job position, previous experience can be the deciding factor of whether or not the job is offered. Employers are looking for prospective employees that can utilize knowledge from past experiences in order to learn new information quickly in addition to being able to problem solve when needed. When students learn new information, rarely do they have the proper past experiences to connect the information to a real-life application which makes it more difficult to understand and remember. If students are given opportunities in school to learn through real-world applications, all of a sudden the concept of volume calculations they are using to relate the amount of oil that was spilt in the Gulf Oil Spill to the volume of the classroom they are in, or the swimming pool in the park become more understandable and provide a better foundation of information for future problems. The open-ended structure to a real-world problem provides students the ability to make connections with their interests and prior experiences. (Cunningham & Lachapelle, 2012) In the real world, the engineering career is filled with open-ended problems.

Hands-on, Active Learning

Engineering is a hands-on field. (Jeffers, Safferman, & Safferman, 2004) It is difficult to teach fundamental concepts such as the law of gravity or calculating volume of non-uniform objects by water displacement without actually doing hands-on learning. Students focus their effort on being engaged in the hands-on learning that is taking place and don't even realize the valuable information that they are gaining in the process.

Not all hands-on experiences have an impact on student achievement. Activelearning that emphasizes student engagement in analyzing and making sense of the presented information is clearly superior to traditional student learning in terms of increasing student conceptual understanding. (Cunningham & Lachapelle, 2012)

Application of Problem Solving Skills

The use of problem solving skills proved to be important in one study that was based on pre-engineering attributes that not only had short term effects on the interest of students, but long term effects of the pre-engineering experience on a student's selfefficacy when compared to students that did not have the same experience. (Fantz, Siller, & DeMiranda, 2011) Problem solving skills are the skills that students must use to solve an open-ended problem. Sometimes students will need to hone their problem solving skills when working on a project by themselves, other times the students will be given a larger problem to solve and will need to do so in a group environment. More often than not, when students are placed into a problem solving environment they are grouped with other students. Part of the learning process then becomes a small group of students working together and trying to explain their different ideas and viewpoints to each other – students must collaborate to solve the problems in a team environment.

Computer/Technology Applications

Technology has become a resource that most careers cannot do without. While basic technology skills such as Microsoft Word, Excel, Power Point, etc. are not only used by teachers, but are taught in computer classes across the United States, few students have the opportunity to take a technology class that would gear them up for the 21st Century workforce.

Technology education should include more than learning basic skills at this point in the 21st Century. Technology education should include: basic skills (at an earlier level), various applications and examples of how and when to use the different types of technology (What is the probability of your basketball team making it to the finals? How to determine randomly within your group of friends who gives a Christmas gift to whom?), and last but most importantly how to trouble shoot that technology.

Research has shown that when schools effectively integrate technology within an educational setting in addition to traditional instructor-led teaching strategies, it improves effective knowledge access and transfer, encourages students to critically think, enhances students' academic achievement, and overall improves the quantity and quality of teaching and learning within a classroom. (Safar & AlKhezzi, 2013) (Ololube, 2009) This integration of technology could include both teachers using technology to explain new

material as well as teachers assign projects for students to complete that requires the use of technology.

Access to educational technology tools has greatly improved in numerous schools across the United States as a result of millions of dollars spent through political and institutional support. (Keengwe & Onchwari, 2008) While the technology per student ratio has increased, the technology has not been utilized or integrated to its full intended potential. (Keengwe & Onchwari, 2008) One study found that fewer than 20% of teachers utilized technology in their classrooms several times a week, and up to 50% of teachers did not use technology at all. (Keengwe & Onchwari, 2008) Some teachers become very hesitant to incorporate technology into their classes due to the amount of change required. Overall teachers are not only asked to change the way they teach their students, but to also change the role they play within their classroom and the way the classroom time is arranged. Teachers need to hone their technology and troubleshooting skills, however teachers will be most successful in utilizing technology within their classroom when they do not act as the experts, but as instead as guides to facilitate the students learning. (Keengwe & Onchwari, 2008)

Feedback on Student Efforts

Feedback is a very beneficial teaching and learning strategy. Feedback comes in many forms, indirect and direct, immediate and delayed, as well as a number of types of feedback that can be delivered (correct response, try again, elaborated, error flagging, etc.) (Shute, 2008) In addition to student feedback influencing achievement, it is also shown to be a significant factor in motivating learning. (Shute, 2008) Studies have been done around the topic of feedback, and have found that good feedback contains the following: (Ertmer, 2007)

- 1. Clarifies what good performance is (goals, criteria, standards).
- 2. Facilitates the development of self-assessment and reflection.
- 3. Delivers high quality information to the students about their learning.
- 4. Encourages teacher and peer dialogue around learning.
- 5. Encourages positive motivational beliefs and self-esteem.
- 6. Provides opportunities to close the gap between current and desired performance.
- 7. Provides information to teachers that can help shape teaching.

Feedback from both teachers as well as peers is very beneficial to the student's learning. Peer feedback allows students to take a different role and participate in each other's learning. This type of learning environment creates on in which students have understanding and appreciation for their peers. (Ertmer, 2007) Work has been done to propose that when students give peer feedback, the amount of learning is greater because the student must step out of the mindset of completing a task and move into one that allows them to read, compare or questions a peers efforts. (Ertmer, 2007)

There is an abundance of articles that have been written about the role that feedback has in student learning. (Shute, 2008) While most articles convey that the feedback is an important piece to the learning process, there are quite a few articles that have reported either no effect or debilitating effects on learning. (Shute, 2008) Feedback can have a negative effect on learning if it is students see the teacher feedback as being critical or controlling or if the students are interrupted by feedback from an external source. (Shute, 2008) There are also a couple of challenges to be aware of when students provide peer feedback. One challenge is students overcoming the anxiety about giving feedback to their peers, especially negative feedback. Another challenge is ensuring that students are giving accurate feedback.

Local Programs versus Scaled Programs

Learning is a complicated phenomenon with numerous, often contradictory, approaches that try to explain when and where learning happens. (Clarke & Dede, 2009) Too many times programs are designed in a particular location and expected to be able to scale, or expand, to impact a greater number of students, but in very different settings. (Clarke & Dede, 2009) One-size-fits-all models for teaching and learning that are abundant in policies and accountability measures are not always realistic in every situation. One-size-fits-all educational programs do not work because they ignore factors that determine a program's efficacy in a particular local situation. (Clarke & Dede, 2009)

A properly scaled project could be used in a variety of locations and situations. In the design of a project that is intended to scale is a 'ruggedness' that needs to be attained so that the project can succeed at multiple locations. (Clarke & Dede, 2009) Clarke and Dede explain that while there are key design features for scaling a project, the robustdesign that is created in order to scale is not expected to outperform a program that is designed to work locally for a specific group of students. (Clarke & Dede, 2009) The design of a project that can scale versus a project that is meant to stay local is different. It is important that program designers decide if it is more important to have a tool reach a variety of people from various backgrounds, or if it is intended to greatly and specifically impact a select group of people. It was decided for the 'You're Hired!' program that the goal is a program that can reach a variety of individuals, including both urban and rural schools.

Coburn defined scale as four interrelated dimensions: depth, sustainability, spread, and shift. Clarke and Dede took Corburn's work one step further by adding evolution to the end of the list of dimensions to bring the total to five. (Clarke & Dede, 2009)

Depth

Depth can be related to the deep and substantial change in a classroom practice, changing teacher's beliefs, norms within the classroom and classroom interactions, and pedagogical principles as enacted in the curriculum. (Clarke & Dede, 2009) (Coburn, 2003) Coburn defines teacher's beliefs as the underlying assumptions that teachers have about the way that students learn, the nature of the subject material, expectations they have for their students, and what constitutes effective instruction. (Coburn, 2003) Uprooting a teacher's beliefs can prove to be very difficult given that teachers tend to 'gravitate' or stay close to approaches of teaching and learning that are similar to that of their previous classroom situations. (Coburn, 2003)

Because depth is a critical element of scale, it is important that the degree of depth is able to be assessed correctly. It is probable that depth cannot be assessed by survey alone. Instead, capturing depth correctly could possibly require in-depth interviewing, classroom observation, instructional support, etc. (Coburn, 2003)

Sustainability

Sustainability relates to the ability of a classroom practice to continue over a substantial period of time given the practice's depth. It is possible and common that

reforms can be adopted without being implemented and can be implemented superficially only to fail shortly after they start. (Coburn, 2003) In order to reach a level of long term sustainability, researchers must use a robust design that is not only optimal for many different locations, but also for a variety of different types of learners. (Clarke & Dede, 2009)

When researchers are preparing a program that has the necessary attributes to be able to scale and sustain, that enough of the original requirements remain in order to make the program still valuable. (Clarke & Dede, 2009) One way to do this is by developing a list of attributes that define a program during design. Then ensure that while the project is implemented and slightly altered (adapted) to meet the need of a variety of locations, that the main attributes remain assuring that the project is still carrying the same significance as the original design.

Spread

Spread involves the diffusion of the innovation to a large number of locations. In order to achieve spread, the design of the project needs to be modified and altered to reduce the amount of resources and expertise needed for success. (Clarke & Dede, 2009) If a project can run off of minimal expertise from researchers, it is more likely to spread seamlessly. However, if the researchers continue to play a large role in the project as in providing a feedback loop, customizing data and projects for each location, etc. It is less likely that the project will successfully be able to spread.

Websites are a great tool to encourage spread more efficiently. Websites provide information needed by each location, and in the design leaves it up to the individual locations to utilize that resource with minimal help from the research team. In the case of some schools, all information to successfully run the project could be available online with specific teacher sections being password protected. This type of a design tool allows teachers and students to share the same resource which makes it very clear where the important resources can be found.

Shift

Shift is the exciting point in the project's design where the ownership is passed from researchers/designers to the individual sites that are implementing the project. During shift, teachers and schools take ownership in the sense that instead of following a strict curriculum, it is encouraged that the focus be on adapting to meet the needs of their specific location. (Clarke & Dede, 2009) (Coburn, 2003)

In the case of an educational project, teachers become co-evaluators, and codesigners at every stage of implementation. (Clarke & Dede, 2009) Teachers provide researchers with a very important perspective in regards to the overall design of an educational project. Integrating teacher feedback into redesign is critical in creating a project that can truly scale up and sustain.

Evolution

Evolution is the process of adopters taking a project and revising it in such a way that project designers have reshaped their thinking in terms of the innovation of the project. (Clarke & Dede, 2009) Evolution goes further than shift in terms of users taking ownership of a project. With evolution, users not only take ownership, but they incorporate their ideas into the evolution of the project.(Clarke & Dede, 2009) Having a continuous feedback loop between designers and teachers will help to ensure appropriate evolution of the project.

CHAPTER III

YOU'RE HIRED! PROGRAM DESIGN

I am proposing that a successful engineering education outreach program can be designed that includes many of the attributes described in Chapter 2 that have been proven to be effective in changing K-12 students' attitudes and understanding of engineering. The program design not only includes the attributes for successful programs, but it also has been specifically designed to scale and sustain across a broader region. This ability to scale will ensure that schools over a greater area will have access to positive engineering experiences for their students. The program is named 'You're Hired!' to symbolize that this program can help students hone the skills necessary for being successful in their future workplace, not only in an engineering career, but in a broad range of non-engineering careers that also require these important skills.

The 'You're Hired!' program is designed to engage students in an intense, coherent set of STEM-focused experiences, which requires the use of the engineering design process and infuses 21st Century Skills to solve real-world problems. The program was designed to be an innovative way to meet school's needs of (1) implementing interdisciplinary STEM experiences for all their students and teachers without burdening existing resources and (2) providing a method to assess and hone all students' 21st Century Skills. The program also is designed to addresses industry's need for more students to enter STEM-related careers by providing the students with a positive, authentic experience of engineering. Ultimately, 'You're Hired!' is intended to be a proven piloted program which can be replicated, adapted to other contexts, and nationally scaled using Dede's dimensions of scale that were covered in Chapter 2.

3.1 You're Hired! Program Design

The 'You're Hired!' program design has small groups of students, ideally 5-8, work together as a 'company' for an entire school day to research a real-world problem, experiment, determine a reasonable solution given the problem constraints, and then communicate, advertise, and market their discoveries/solution in front of a boardroom that is deciding which company has done the best job of meeting the constraints of the problem. In each activity the students participate in, there are five possible careers to choose from. Having 5-8 students in each group allows there to be at least one student per job, but there isn't too many students in which an individual feels unneeded/left out.

This program provides an experience that incorporates many of the key attributes for successful pre-engineering activities. Students must be *self-motivated* to complete their tasks in a timely manner. Students also have to *problem-solve* in an *active learning* environment to come up with a solution to the given *real-world problem*. Students use *computer applications* to make graphs and presentations for communicating their discovery/solution to a community-led boardroom, comprised of school administrators, school board members and local industry representatives including professional engineers. The boardroom gives students immediate *feedback on students' efforts* after each activity.

The 'You're Hired!' program is not just a onetime event; it is a series of three day-long activities that occur over the course of a school year to provide multiple

29

authentic experiences of possible engineering careers. Individual 'You're Hired!' activities set learning in a real-world context by posing different problems with worldwide impact that require the expertise of different engineering disciplines. For example, one project has students in the role of chemical/environmental engineers that are helping to clean up an oil spill using a nano-based coating on sand. Another project has students diving into electrical engineering in which they have the opportunity to build a nightlight circuit and determine through a series of calculations the amount of power that can be saved by switching from incandescent to LED lighting in a home. Problems are developed that will be of personal relevance to students (e.g., determining the effect texting has on reaction times and relating the experimental results to the dangers of texting while driving) and that can be described with simple engineering ideas. Introducing various engineering careers within the construct of solving socially relevant problems helps convey the diversity within the engineering profession. The variety of problems is also intended to increase the likelihood of individual students experiencing the 'wow' factor at least once during the year.

The age of students participating in 'You're Hired!' range from 6th grade to 12th grade. While engineering outreach programs are being implemented throughout the entire K-12 range nationwide, many of these programs focus on middle to high school students because this is the age when students begin thinking about and planning their own paths in education. (Dawes & Rasmussen, 2006) (Cunningham & Lachapelle, 2012) (Cunningham & Lachapelle, The Impact of Engineering is Elementary (EiE) on Students' Attitudes Towards Engineering and Science, 2010) (Lachapelle, Phadnis, Hertel, & Cunningham, 2012) (Habash & Suurtamm, 2010) (Mooney & Laubach, 2002) (Davis,

Yeary, & Sluss, 2012) Students can do this by choosing classes to take in high school and thinking about options to pursue math or science related fields in college. (Dawes & Rasmussen, 2006) (Mooney & Laubach, 2002) (Habash & Suurtamm, 2010) (Davis, Yeary, & Sluss, 2012)

'You're Hired!' Over the Course of a School Year

The 'You're Hired!' program begins at a participating school prior to the first day-long activity with teachers organizing companies that consist of 5-8 students. Depending on the size of the school, if multiple grades are participating teachers can choose to mix students from the various grades in a company. It is important that teachers divide up the groups because they have spent the most time with the students in their regular classrooms and know which students should not be in the same group.

The 'You're Hired!' team also works with a school contact to coordinate scheduling days, students, teachers, boardroom participants and other necessary approvals well in advance to ensure smooth project execution. In the future, in order to scale, the organizational information will be available online. The school then completes a total of three separate 'You're Hired!' activities throughout the school year. During and after these activities, students receive multiple forms of feedback on their success and efforts.

Breakdown of a 'You're Hired!' Activity

The basic framework of a day-long 'You're Hired!' activity ensures that schools are able to easily conduct the activity with minimal resources required, a key aspect when designing for scalability. The following breakdown is offered as a template, with flexibility to modify as needed. Each activity starts with an introduction of the problem statement given by either a teacher, an online video from the 'You're Hired!' website or myself personally. The problem statements are designed to be simple, yet open-ended so that students can dive into a creative solution. In the future, all introductions will be available on the 'You're Hired!' website so that the school/classroom can start the day without any additional support.

For the pilot year, 2012-2013, the three problem statements used with the schools were:

- 1. Using nano-technology coated sand to clean up an oil spill
- Experimenting with reaction times and creating a prototype to end texting while driving
- 3. Creating a school lunch menu that follows all new nutrition standards.

For the 2013-2014 school year the problem statements were:

- 1. Using Glo-Germ powder to show how quickly viruses can spread in a school
- 2. Building a nightlight circuit to determine the amount of energy that can be saved by switching every home from incandescent to LED lighting
- Experimenting with non-Newtonian fluids to try to find a mixture to replace the synovial fluid in the human knee that can meet both viscosity and force requirements.

During the introduction to the activity student groups are given a packet of paper information along with an experiment kit. The paper packet includes: A written version of the introduction, a company information sheet that students will need to turn in a guide/list of objectives and questions to answer in the form of the engineering design process, a copy of the rubric that the boardroom members will use to score the

presentations, and access to three separate 5-minute question sessions with teachers. For

projects that involve more difficult experimentation, an additional experimental

procedure document is given to students as well as online videos are available on the

'You're Hired!' website to assist in student discovery.

For the oil spill design challenge, students were given a written introduction as

shown in Figure 4.1. The experiment kit included an oil dropper, vegetable oil, a plastic

cup, a spoon, and Magic Sand.

An environmentalist group known as **Greener Ways** has decided to undertake the large task of safely cleaning up oil spills. It's estimated that millions of gallons of oil enter the oceans each year, from ship motors, natural leaking, and oil spills. The oil waste then reaches the coastal shoreline and contaminates beach vegetation, wildlife, and swimming environments.

Greener Ways has developed a break-through product that is predicted to clean oil spills easily and effectively. They have developed a product known as Magic Sand that they would like to use to clean up such incidences.

Your company is a state of the art company that is very good at cutting edge technology. Your company has been chosen to be one of the finalists in our search for a company that will be able to use the Magic Sand to clean up an oil spill.

Your company's tasks will be to:

- Research
- Experiment
- Advertise
- Market the product
- Come up with a presentation to convince us to use your company over other leading companies.

Figure 4.1. Oil Spill Problem Statement.

Once the introduction is completed and the students break into their predetermined companies the students must define their company including a name, mission statement, logo and organizational roles. A company profile sheet within the student information packets calls out specific roles, requiring the students to take on specific tasks during the day. In the design of the individual activities, groups of students should preferably not be larger than 8 students in order keep students feeling necessary and engaged in the activity. With this design there are enough students in the group so that more than one student can take on a given role for the day, however if one student is not participating or has to leave school for some reason, the group can still manage and be successful. Typical roles include manager, advertising specialist, website designer, engineer, and technician. Each 'You're Hired!' activity tailors the engineering and technology-related roles to the specific careers that would actually be working on the problem. This level of explicit detail provides the practical application of role responsibility in a STEM career and draws a strong connection between their one-day experience and a possible future career.

After the company is defined, the students work on developing their solution in a problem-based/active learning environment for a majority of the school day (4-5 hours). Students are guided in what they should be doing in the form of an engineering design process sheet which is included in their paper packets. This engineering design process document as shown in Figure 4.2 guides the student companies through an engineering mindset of: identify the problem, research the problem, develop possible solutions, select best solution, construct a prototype, test and evaluate the solution, redesign if necessary, and communicate the solution. As students go through the engineering design process

34

there are a series of 'questions to consider' that are asked in order to get students to think about and find a solution that meets the objectives of the given activity.



Figure 4.2. The Engineering Design Process.

Identify the Problem

The students were initially made aware of the problem during the introduction. The company also has a written form of the introduction/problem statement available to them in their company packet and should utilize that document in order to clearly identify the objectives for the project. For the example of the oil spill project the objectives are stated that the students should first determine if the nano-based coated sand, Magic Sand, works to clean up oil spills, and second the companies should come up with a prototype for using the sand to clean up the 2010 Gulf Oil Spill.

Research the Problem

Once students have turned in their company profile sheets, it means that students know what their role in the activity will be and can start working on the next step of the engineering design process. In researching the problem, the engineer and technician will generally start by looking over the experimental supplies given. The engineers and technicians should research and learn the basics of the nano-based coating on the sand. The sand coating is hydrophobic which means it repels water. The sand and oil do not repel one another, in fact the combination of the nano-based coated sand and the oil creates a somewhat viscous mixture which potentially could make it easier to clean. The advertising marketer along with the web designer should research facts about past oil spill clean ups, methods that are utilized versus not utilized, how using this sand would give the company a leading edge over other methods, etc. Advertising marketers along with web designers will also use this step to find images and facts about the 2010 Gulf Oil Spill.

36

Develop Possible Solutions

The companies may look into other methods of cleaning up oil spills for guidance. For example, when researching oil spills, a group may find that booms are used to contain the oil to a specific area. Students might choose to use a design similar to this to contain the oil spill before using the sand. When developing possible solutions, students are encouraged as part of the engineering design process document to create sketches or construct make-shift prototypes out of materials found in the classroom to better explain their ideas. Students work through the details of their proposed solutions in order to determine the feasibility of them.

One of the tasks students have to complete is to prepare a cost estimate for the clean-up. This cost estimate keeps students from finding solutions that involve major equipment and excess resources. Students have to think about how many helicopters or ships they would use. What is the cost of the sand and how much do you need for your design? It is important that students think through their solutions and are able to justify their decisions in the boardroom.

Select the Best Solution

Once the students have thought through a couple possible solutions, the company will select its best solution. This solution should be explained in detail including rough cost estimations and drawings.

Sometimes the solution the company presents in the boardroom is the only solution they thought of, the students did not consider a redesign. When this happens, generally the presentations are weak in detail and explanation. Teachers are encouraged to take advantage of this situation and talk to the students about the importance of the engineering design process and making sure brainstorm and redesign are part of that process. The teacher input generally takes place during the 'Teacher Feedback' portion of the boardroom hour or the next school day.

Construct a Prototype

Constructing a prototype is the part of the project where students can be creative. At this point, students know the experiment and what their solution is and can create a visual for the boardroom. For the oil spill project this could include using a paper towel as an example of the fine-holed net that is suggested to keep the oil/sand combination from sinking to the bottom of the ocean after distribution of the sand, or the bucket and fish tank pump that is designed to circulate the ocean water and capture the oil/sand mixture in a filter before releasing clean water back into the ocean.

In some 'You're Hired!' projects, a prototype isn't as straight forward. For example the Energy activity doesn't have a prototype the students can build necessarily, so instead students must present their final solution in the form of a mathematical model. In regards to saving energy, does switching light bulbs really increase savings, or would it be more beneficial to unplug unused appliances? Students must do the math and then determine what they think the best solution is, and how much energy their company can save if their solution is implemented.

Test and Evaluate Solution

Students should test their prototype before presenting in front of the boardroom. This point of the activity might be the first time other group members have seen the solution the engineers and technicians have come up with first hand. The engineers and technicians should have been communicating their ideas to the advertising specialist, the web designer and the marketing team so that those individuals could tailor their work, but this may be the first time everyone has seen an actual prototype/product. Pictures and/or videos are usually taken at this point in order to best convey the solution to the boardroom.

Redesign if Necessary

If a small fix cannot solve the design issues with the prototype, then students should try out other options as part of the redesign process. It is important to both the students and the boardroom members that the presentations include something that is tested and that works.

Out of all the engineering design process steps the students go through during the 'You're Hired!' activity, this is the step that is overlooked the most. There is an abundance of time and effort that goes into getting to a solution for these projects. When a solution does not work the way the students think it should, the students may decide to keep it due to a timing issue, or because they still think it is a good idea.

Communicate the Solution

Once students have prepared a solution, they must present that solution to the boardroom. Students must put together a presentation including key elements stated in the problem statement, the engineering design process worksheet, and the boardroom scoring rubric. Students should understand their presenting time limit and have practiced to ensure they could meet that time limit.

Most importantly, students need to make sure they are prepared to answer questions based on what they worked on throughout the day. They are the experts in their role, and the boardroom members may ask them questions based on what they did during the activity.

The program is designed so that during the 4-5 hours the students are working to develop a solution, the amount of interaction and collaboration between students is ample, whereas the amount of interaction with the teachers is minimal.

While teachers do not interact with the students as much as they would on an average school day, the teachers play a key role during the day by providing content knowledge on a just-in-time basis through a system of question vouchers. For the day-long activity each student group is given three vouchers that can be used for 5-minute question sessions with a teacher. These question sessions provide direct teacher instruction on the specific topics of immediate interest to students. Because these mini-lessons are student-instigated at a time of need, rather than force-fed by teachers at the beginning of the day before their importance is clear, students are in a much better position to understand the necessity of the information provided and its relevance to their problem-solving and investigating practices. Additionally, 'such just-in-time instruction promotes knowledge construction in a way that makes knowledge more available for future use in relevant contexts.' (Hmelo-Silver, Golan Duncan, & Chinn, 2007) (Edelson, 2001) The number of vouchers is limited to encourage students to problem solve and collaborate together instead of immediately looking to a teacher for the "correct" answer.

Because the 'You're Hired!' activities build on knowledge students already have, all students are able to participate and they can do so independently from teachers. For example, with the oil spill activity, students are told that millions of gallons of oil are introduced to the oceans each year. The students are then told they need to be able to convey this information to the boardroom to make them care about not only the issue, but the clean-up efforts. To do that, students need to first understand for themselves the volume of oil that was spilt. Students are guided on the engineering design process sheet to convert the huge unimaginable number of 'millions of gallons' to a more approachable value such as 'x' number of classrooms full of oil is spilt. Students determine for themselves what they would like to compare the volume of oil spilt to (classrooms, swimming pools, bathtubs, etc.), and then use their knowledge of volume calculations and unit conversions to solve. The Common Core Standards Initiative states that by the 5th grade, students are to have been introduced to volume calculations and unit conversions. (Grade 5 Measurement & Data, 2014)

Due to the problem based nature of this program, students are encouraged to utilize technology and other resources such as the library to guide them to useful information that can assist them in finding a solution. Students will quickly see that 'You're Hired!' is a project that merges all their different classes together in a real-world application. If students cannot find the answer they are looking for, they have the chance to use one of their three 5-minute questions sessions to ask a teacher for guidance.

Students are not told when they are supposed to present to the boardroom, instead students take responsibility and sign themselves up for an appropriate time on the Boardroom Sign-Up Sheet as shown in Figure 4.3, which also includes times for students to take the post-activity survey, clean-up, and talk with the teacher from their classroom to receive immediate feedback on their efforts and success. Students are also not instructed on what type of a presentation they should give. Some students will choose to present via PowerPoint, others through a Prezi. Some students might decide to present

41

through the website they designed throughout the day. The boardroom presentation is broke into two parts, a six minute presentation followed by four minutes of questions to total ten minutes in the boardroom. Students should know what the boardroom is expecting because they were given a copy of the boardroom scoring rubric as one of the documents in the paper packet. The boardroom rubric states that companies will be scored on different aspects including the engineering design process, teamwork, and creativity of the solution, etc. Whether the students chose to read through every document in that packet is up to each individual company.



Figure 4.3. The Boardroom Sign-Up Sheet.

The companies present their solution to the challenge in front of a boardroom comprised of school board members, school administrators, and local industry representatives including local engineers. Generally a boardroom is made up of 2-4 volunteers and can see 4 companies per hour. In the design, 2-4 volunteers consistently can generate quality questions and still have time to properly assess the student presentations. Company presentations involve all students that participated throughout the day. Understanding that one or two students might lead the presentation, all students must be available and ready for any questions the boardroom members may have.

The boardroom not only provides immediate feedback on the company's solution, marketing and presentation, but also includes recommendations for improvements, both of which enhance overall student learning by affirming that students' contributions and ideas have value. For students, this feedback from the boardroom is different than teacher feedback. Students see teachers every day and become accustomed to how teachers asses their work. The boardroom provides a fresh perspective to both the student efforts as well as the teacher efforts which is beneficial.

The boardroom is not the only time students get feedback. During the postactivity survey students are asked questions about the specific activity they just completed. The survey questions provide a chance for the individual students to reflect on how they thought they did throughout the activity in terms of 21st Century Skills. This survey also provides students the opportunity to assess how well their group members did regarding the same 21st Century Skills. There is an area on the survey for individuals to reflect on the things they can improve as well as an area in which students can set goals for the next activity. Peer/Self feedback is returned to the individuals shortly after the

43

activity is completed, within a school day or two. Students receive a hand-out similar to the one shown in Figure 4.4. This feedback provides an anonymous average of how a group rates an individual compared to how that individual rates themselves. The ability for students to give honest feedback to their peers comes with knowing that their assessment of their teammates is anonymous. This reduces the challenges that are associated with peer feedback.

Name Comp	any Name		
Comp	any Name		
comp			
	· 사망에 가장 알려 있다. 영국는 것은 사람이 바람이 있다. 한 것은 사람이 있는 것은 것을 하는 것을 수 있다.		
21 ST CENT	URY SKILLS ASSESSMENT (5-EXCELI	ENT JOB, 1-NOT ACCEPT.	ABLE)
	Skills Assessed	Self-Assessment	Company Members
Leadership		5	4.8
Participation	and cooperation	5	4.9
Contribution	s to the project	5	4.9
Attitude		4	4.6
Commitment	to the group's success	5	4.9
Ability to put	themselves in another person's shoes	4	4.5
Ability to ide	ntify and move onto the next task without	5	4,8
	and a second		27 C
DEDCOM	A REPORT OF A CONTRACT OF A DESCRIPTION OF		
LEVOIN	AL REFLECTION		
Where on the	AL REFLECTION scale below would you place your group's p	resentation and WHY? B	e sure to cite as evidence
Where on the the board roo	AL REFLECTION : scale below would you place your group's p om's reaction as well as your "gut feeling".	resentation and WHY? B	e sure to cite as evidence
Where on the the board roo Ok on	AL REFLECTION : scale below would you place your group's p im's reaction as well as your "gut feeling". : the day: We didn't have a product ready to	resentation and WHY? B show	e sure to cite as evidence
Where on the the board roo Ok on	AL REFLECTION escale below would you place your group's p om's reaction as well as your "gut feeling". the day: We didn't have a product ready to	resentation and WHY? B show	e sure to cite as evidence
Where on the the board roc Ok on Overall, for n	AL REFLECTION escale below would you place your group's p om's reaction as well as your "gut feeling". the day: We didn't have a product ready to ext timeWhat are <u>three</u> things you would d Make a product to sell first	resentation and WHY? B show o differently? Why?	e sure to cite as evidence
Where on the the board roo Ok on Overall, for n 1. 2.	AL REFLECTION escale below would you place your group's p om's reaction as well as your "gut feeling". whe day: We didn't have a product ready to ext timeWhat are <u>three</u> things you would d Make a product to sell first Different teams	resentation and WHY? B show o differently? Why?	e sure to cite as evidenci
Where on the the board roc Ok on Overall, for n 1, 2, 3,	AL REFLECTION escale below would you place your group's p om's reaction as well as your "gut feeling". whe day: We didn't have a product ready to ext timeWhat are <u>three</u> things you would d Make a product to sell first Different teams Have fun doing it so we have better attitud	resentation and WHY? B show o differently? Why? es	e sure to cite as evidenci
Where on the the board roc Ok on Overall, for n 1. 2. 3.	AL REFLECTION escale below would you place your group's p om's reaction as well as your "gut feeling". whe day: We didn't have a product ready to ext timeWhat are <u>three</u> things you would d Make a product to sell first Different teams Have fun doing it so we have better attitud	resentation and WHY? B show o differently? Why? es	e sure to cite as evidenco
Where on the the board roo Ok on Overall, for n 1, 2, 3, Please descri	AL REFLECTION escale below would you place your group's p om's reaction as well as your "gut feeling". The day: We didn't have a product ready to ext timeWhat are <u>three</u> things you would d Make a product to sell first Different teams Have fun doing it so we have better attitud be what you did today? How did you contribu	resentation and WHY? B show o differently? Why? les ute to the project?	e sure to cite as evidenci
Where on the the board roc Ok on Overall, for n 1, 2, 3, Please descri	AL REFLECTION escale below would you place your group's p om's reaction as well as your "gut feeling". The day: We didn't have a product ready to ext timeWhat are <u>three</u> things you would d Make a product to sell first Different teams Have fun doing it so we have better attitud be what you did today? How did you contribu- te scripts for everyone and helps with poster	resentation and WHY? B show o differently? Why? es ute to the project? and eo thines straight w	e sure to cite as evidenci
Where on the the board roc Ok on Overall, for m 1. 2. 3. Please descri I wrot gone.	AL REFLECTION escale below would you place your group's p om's reaction as well as your "gut feeling". The day: We didn't have a product ready to ext timeWhat are <u>three</u> things you would d Make a product to sell first Different teams Have fun doing it so we have better attitud be what you did today? How did you contribu- te scripts for everyone and helps with poster	resentation and WHY? B show o differently? Why? es ute to the project? and go things straight w	e sure to cite as evidenco /hen managers were
Where on the the board roc Ok on Overall, for n 1, 2, 3, Please descri I wrot gone. How can you	AL REFLECTION escale below would you place your group's p om's reaction as well as your "gut feeling". The day: We didn't have a product ready to ext timeWhat are <u>three</u> things you would d Make a product to sell first Different teams Have fun doing it so we have better attitud be what you did today? How did you contribu- te scripts for everyone and helps with poster use your reflection and peer evaluation to be	resentation and WHY? B show o differently? Why? es ate to the project? and go things straight w e more of a team player i	e sure to cite as evidence /hen managers were n solving conflict or

Figure 4.4. Peer/Self-Assessment Feedback.

Some of the 'You're Hired!' activities can get messy. Clean-up is simply that, a time for students to straighten up the room, hallways, area they had occupied throughout the day and ensure that the school is cleaner after the activity than it was before.

'You're Hired!' is designed to be a series of short term interventions that give students authentic experiences that can later be built on in the classroom. During the activity teachers are available for student support, but also have the opportunity to stand back and observe student learning. This allows teachers to see gaps within students' ability to utilize educational material and take notes. During the boardroom hour, teachers have the opportunity to sit down with each individual company for 15 minutes to give immediate feedback on what they observed throughout the activity.

The boardroom hour generally ends shortly before the end of the school day. After the final boardroom session, depending on how much time there is before the day ends, some schools choose to compile boardroom winners quickly and announce the winners of each boardroom before the students leave. It is suggested that each boardroom has its own winner due to the differences between boardroom members scoring. By keeping the boardrooms separate in terms of scoring, if a judge tends to give high scores, than that judge will give all the companies high scores which will average out as opposed to one boardroom scoring average presentations highly and another boardroom scoring presentations low.

In the following days in the classroom, teachers now have a personal experience for students to reflect on while revisiting key material. For example, an English teacher at one school realized that while students had learned about citations, many of the presentations did not use them correctly. That teacher was then able to revisit the topic in

45

the classroom, using the 'You're Hired!' activity as an example. During the next activity, the students did much better at understanding the importance of citing the information they found. As I continue to develop the 'You're Hired!' program, the addition of professional development for teachers will be a key component, including training on how to take 'You're Hired!' teacher observations to the next level and allowing teachers to be able to reflect with the students on their experiences in the regular classroom.

After the school year is over and data has been compiled and analyzed, a report is put together for the schools. Late summer/early fall the report is distributed to the school contacts with the intent that the report will provide motivation for the school to participate again the following year. In the report the specific school's student pre-survey and post-survey results are presented. Those results are then compared to an anonymous grouping of all the other participating schools. This allows schools administrators to see not only the progress that their students made, but also where their students stand in comparison to other schools doing the same project. A large motivator for the schools comes from the data about program impact on students. If a high percentage of students agree that the 'You're Hired!' program 'helped them understand problem solving better', that would provide a very positive result for the school officials to see, and ultimately would be a motivator for the schools to participate again.

CHAPTER IV

RESEARCH METHODOLOGY

The purpose of this research was to determine the success of the 'You're Hired!' engineering education program at changing students' attitudes and understating of engineering.

In order to research the effectiveness of the "You're Hired!' program at promoting positive attitudes towards engineering, pre- and post-surveys were used for summative evaluation of students' awareness and perceptions towards different aspects of engineering. The pre- and post-survey responses were studied using statistical analysis, considering factors such as program participation, gender and school. The influence of these factors are thought to be able to provide insight when designing a program that can work not only in one specific area, but that can be scaled and able to sustain to benefit a larger region. Data from these surveys has been used to address the following research question: Does the 'You're Hired!' program lead to a change in student attitudes towards engineering? The other key metrics for project effectiveness, increased engineering design skills and enhanced 21st Century skills, are not directly assessed in the student surveys. Separate assessment rubrics for measuring student growth in these areas are under development. Evaluation for this research, therefore, focuses on student attitudes toward and understanding of engineering careers. This research study provides a solid platform that can be used as motivation for more in depth future research work. This research is the first step of many in determining an effective engineering education experience for not only students, but for teachers, administrators, and the community.

4.1 Institutional Review Board (IRB)

My research on the 'You're Hired!' program involves humans as research subjects and thus the study had to be approved by an Institutional Review Board, IRB. The job of the IRB is to ensure the safety of all human subjects that are involved in research. This research project on the 'You're Hired!' program was approved through the University of North Dakota's IRB. In order to meet IRB requirements, every school participating in the research study needed to complete the following two items.

First, a school administrator had to write and sign a letter of willingness to participate in the project. The letter stated that the administrator would like the 'You're Hired!' program in their school for the upcoming year and included a list of deliverables the administrator agreed to provide. That list included deliverables such as: students would take a pre-survey, the school would complete 3 separate 'You're Hired!' activities, students would take a post-survey, and that teachers would provide feedback to the program design team.

Second, students had to provide assent to participate in the project. Because the IRB declared that the program was similar to what a normal classroom could potentially do, parental consent was not required for students under the age of 18 to participate. The program is part of a research project however, so assent from every student stating willingness to participate was necessary. The assent form was presented to the

48

participants as the first page of the pre-survey that every student took. If a student chose not to give assent, the student still participated in the program, however their data was not included in the research statistics.

Along with the two required pieces of documentation, the IRB also required that all documents that were used for the activities be accepted through the IRB approval process. That means that every document for each activity needed to be approved in order for it to be used as part of the research project.

4.2 2012-2013 You're Hired! Program

Prior to the 2012-2013 school year, there were only two schools that had experienced 'You're Hired!' as a proof of concept for an all-day engineering education activity. Both of the schools showed great interest and as a result work was done to pursue the program. The 2012-2013 school year was the pilot year for 'You're Hired! and included 17 schools. Of those 17 schools, three were included in the research study. Each of the three schools is considered suburban, however at varying degrees. Two schools were middle school grades (7-8) in which there was roughly 80 students per grade, and one school consisted of a cohort of high school students (grade 10) in which there was 18 participants.

Data Collection

Shortly after the start of the 2012-2013 academic year, schools administered a paper copy of the letter of assent to each student to read, sign and return. Only 10 of 17 of schools completed student assent forms. Without these forms, the individual student responses cannot be used as research data.

Schools then had students take either a paper or electronic pre-survey adopted from the Assessing Women and Men in Engineering (AWE) Core Middle School Immediate Pre-Participation Survey. (AWE instrument user guide - pre-college surveys, Accessed 2011) This survey established the baseline data of their initial exposure and knowledge about engineering and how this exposure and/or knowledge influences their decisions about future career paths. The survey also includes background questions such as age, gender, previous classes taken, etc. Because the survey tool was designed by AWE, there were questions asked that did not pertain directly to what I wanted to research, the change in attitudes towards engineering. From the survey, four main categories of questions were analyzed; Engineer, Future Work, Interests, and Attitude and Skills. The pre-survey provided an evaluation of these four question categories when compared to the end-of-the-year post-survey. Another category involving Impact of the program was also included in the post-survey analysis. In total 14 of the 17 of schools in the pilot year did not complete required documentation such as pre/ post-surveys or letters of willingness to participate from an administrator which were needed to be included in the research.

Both pre- and post-survey questions with the allowed categorical responses are listed in Table 4.1. Attitudes towards Engineer questions (E1-E8) relate to direct attitude and perception about the field of engineering and the interest in becoming an engineer. Future Work (F1-F9) and Interest (In1-In11) questions provided a background of the type of students that were participating in the program. Self-Efficacy, or Attitude and Skills questions (A1-A8) were chosen to measure the student's attitude in regards to his/her understanding of the concept of the engineering design process and ability to successfully work with or lead a team to an effective solution. Project Impact questions (Im1-Im9)

were used to gain insight into the students' perspective on the overall project. A full list

of all questions asked on the pre-/post-surveys, including demographic questions, can be

found in Appendix A.

Table 4.1. Electronic Pre-Survey and Post-Survey Questions by Category with Response Options.

IMPA	IMPACT: How much did participating in the activity impact each of the following:			
(Not at All, Slightly, Moderately, A Great Deal)				
Im1	Helped me to understand problem solving better			
Im2	Led me to a better understanding of my own career goals			
Im3	Increased my interests in studying engineering in college			
Im4	Increased my interests in studying a technical degree in college			
Im5	Made me think more about what I will do after graduating from high school			
Im6	Made me decide to work harder in school			
Im7	Made me decide to take different classes in school (including college) than I had planned to			
Im8	Made me more confident in my ability to succeed in engineering or a technical field			
Im9	Increased my confidence in my ability to participate in engineering projects or activities			
ENGINEERS: Read the following statements about what engineers might do and indicate your agreement or disagreement with each statement: (Agree, Disagree, Don't Know)				
E1	Mainly work on machines and computers			
E2	Mainly work with other people to solve problems			
E3	Work on things that help the world			
E4	Can choose to do many different kinds of jobs			
E5	Mainly work on things that have nothing to do with me			
E6	I don't know what engineers do			
E7	Pursue a career in an engineering-related field?			
E8	Do you think you want to be an engineer?			
FUTURE WORK: How important is it to you to do: (Not Important, Somewhat Important, Very Important)				
F1	Work that makes me think			
F2	Work that allows me to make lots of money			
F3	Work that allows me to use math, computer, engineering, or science skills			
F4	Work that allows me to tell other people what to do			
F5	Work that allows me to help solve problems and create solutions			
F6	Work that is fun to do			
F7	Work that allows me to have time with family			
F8	Work that allows me to help my community and/or society			

F8 Work that allows me to help my community and/or society

Table 4.1. cont.

FUTU	FUTURE WORK: How important is it to you to do: (Not Important, Somewhat			
Important, Very Important)				
F9	Work that makes people think highly of me			
F10	Work that is satisfying to me			
INTERESTS: Here is a list of statements. Tell us what you think about them. Select a response that indicates your level of agreement: (Strongly Disagree, Somewhat Disagree, Somewhat Agree, Strongly Agree)				
Jul	Llock forward to gaine alogg in school			
IIII In2	Lie als forward to science class in school			
Inz	I nook forward to main class in school			
I., 2	I would rather solve a problem by doing an experiment than be told the			
Ins	answer Mene time also alla answer			
In 1	More time should be spent on projects in science or technology activities at			
1114	School			
In5	a would like to (or already do) befolig to a science or technology activities			
mu	Last bored when I watch programs on channels like Discovery Channel			
In6	Animal Planet Nova Mythbusters etc			
In7	Llike to get science books or science experiment kits as presents			
In8	I like learning how things work			
In9	Science is too hard when it involves math			
In10	Science is a difficult subject			
In11	Doing experiments in science class is frustrating			
In12	I feel comfortable with using a computer to make graphs and tables			
In13	I am interested in learning more about how things work			
In14	I like to learn to use new technology			
ATTI worki us ho Alway	ATTITUDES AND SKILLS: The table lists things you can do when you are working on school activities or assignments. Check the appropriate box to tell us how often you do each of these things: (Never, Sometimes, Very Often, Always)			
A1	When I see a new math problem, I can use what I have learned to solve the problem			
A2	I can use what I know to design and build something mechanical that works			
	In lab activities, I can use what I have learned to design a solution to a			
A3	problem			
A4	I can effectively lead a team to design and build a hands-on project			
	I know where I can find the information that I need to solve difficult			
A5	problems			
A6	I can explain math or science to my friends to help them understand			
A7	I can get good grades in math			
A8	I can get good grades in science			

After students took the electronic pre-survey, the school completed three separate 'You're Hired!' activities throughout the school year. For the 2012-2013 school year the activities included: 1) using nano-technology coated sand to clean up an oil spill, 2) experimenting with reaction times and creating a prototype to end texting while driving, and 3) creating a school lunch menu that follows all new nutrition standards.

At the end of the academic year, roughly 2-4 weeks after the school had completed the third and final 'You're Hired!' activity, students completed the paper or electronic post-survey. Each participating school received a year-end report anonymously summarizing their students' responses. This report also included a school by school comparison with all student and school names and identifiers removed. An example copy of this year-end report can be found in Appendix B.

The pilot year went well, however there were gaps in the program design that made it difficult to gain all of the necessary information from the schools. To address these shortcomings, several elements of the program were redesigned for the 2013-2014 school year.

4.3 2013-2014 You're Hired Program

For the 2013-2014 school year, there were three schools that participated in the research study: School A, School B and School C. School A and School B had 7th grade participants whereas School C had both 7th and 8th grade participants. School A and School B have much larger class sizes than School C.

The class size in School A is large enough that students are broken down into pods, or sub-sections of students within a grade. The pods of students stay together and have their own set of primary teachers including: math, science, geography and English.
School A allowed the project to have an experimental versus control comparison by having one pod complete the surveys as well as the three separate 'You're Hired!' activities whereas the control group only completed the surveys. School B and School C provided experimental groups only.

Redesign

As part of redesign for the 2013-2014 school year, the student assent was included on the electronic pre-survey as the first question the students read. Students could choose to participate or to not participate in the research before continuing with the rest of the pre-survey. Redesign also included an improved system to make sure that all schools complete the necessary documentation in a timely manner, while improved, this system will need to continue to be redesigned.

Electronic pre- and post-surveys were also part of the pilot year redesign. In order to analyze the data, responses to the survey questions needed to be entered into the data analysis software. To do that, every paper version of the pre- or post-survey needed to be hand entered. This method of manually entering data was not only time consuming, but can lead to errors when typing the data in. In designing a program that could eventually scale across a broader region, it was important to move away from paper surveys towards electronic versions. In the 2013-2014 school year and future years to come, only electronic versions of the surveys will be available. The electronic documentation used for the 2013-2014 program allowed me the ability to quickly see and address schools that had outstanding deliverables needed for research.

Increased discussion about engineering and possible engineering careers throughout the activities was another point for redesign. Students could remember the activities they had previously done, but had a hard time remembering the role which they played during that activity. Company profile sheets were changed as part of redesign to include more specific roles for everyone. Originally engineering and technician roles were called out with great detail, while the other roles were more general, such as manager, researcher, and marketer. It is believed that the generalization of roles lead the whole group to not remember their job for the activity. Redesign for the 2013-2014 school year included the same specific engineering and technician roles aimed at the current project as well as manager, website designer, and advertising specialist, all careers that students could look into and find not only career information, but college information as well. Overall, more emphasis on roles and responsibilities throughout the day, including the boardroom was used during the 2013-2014 school year.

With the goal of increasing student engagement, the process of providing feedback was redesigned to ensure that after each activity, students received all forms of feedback within one to two days of school. This feedback included boardroom presentation results, peer/self - assessment results, and teacher feedback. This immediate feedback is something that was missing from the 2012-2013 school year, but that is one of the key attributes in the program design.

Although the survey questions were another item identified as needing redesign, in the process of scientific research, it was decided that for the 2013-2014 redesign only the structure of the project activity would be changed, while the survey questions would remain unchanged to provide a consistent basis for measuring if the changes could alter student responses. In future redesign, the survey will be looked at closely and changed to better fit the needs of the program.

4.4 Data Analysis

Removing Identifiers

In accordance to IRB requirements, all data analysis and data handling needed to be done without calling out individual student names. To do that, I created student identification numbers for all students and then paired their responses to their number. The student identification numbers can be traced back to original names by only a select group of approved researchers that have access to the research analysis storage folders.

Prior to the redesign that was done for the 2013-2014 school year in which all assent and pre-surveys were done electronically, all students signed a paper version of the assent and some schools had their students take a paper version of the pre-survey. All of these documents with student names need to be kept in a safe place. The documents are organized by school and are locked in a cabinet at North Dakota State College of Science's Fargo campus.

Data Analysis

The data from the pre- and post-surveys for the three 2012-2013 and three 2013-2014 schools was downloaded from the online survey site called Surveymonkey, formatted in excel and analyzed using the statistical software JMP 7. Over the two years, five different schools participated in 'You're Hired!' There was one school, School C that participated during both years. The 2012-2014 data was analyzed in four sets; results from School A's experimental and control groups, results from all students that participated in the project, results based on school and results based on gender in order to understand the project impact more clearly. Descriptive statistics including means, 95% confidence intervals and standard error were calculated as well as significance testing using p-value and chi².

The chi-square test along with its associated p-value was used to determine the significance of categorical data. Because the pre-and post-survey responses were categorical, this test is an appropriate tool for determining significance. The chi-square test measures the statistical difference between two or more independent sets of categorical data. The p-value calculated from the chi² result is the probability that the data sets are not different.

The categorical data was also converted into numerical data using a Likert Scale. Once numerical data was available, descriptive statistics such as means and 95% confidence intervals were calculated.

I used the categorical data to determine the p-values. The survey questions students answered were presented as categorical, and I believe that p-values based off of this analysis are the most accurate. When the data was converted to a Likert Scale there is deviation that may arise because the Likert Scale is not how the students perceived the response options. However, the Likert Scale allowed for an additional opportunity to see changes in responses based on the difference in means and confidence intervals between pre- and post-surveys. The converted numerical data was not used to determine significance.

The first set of data analysis consisted of primary significance tests to determine the program impact by comparing pre- to post-survey responses for the control vs. experimental group. For this comparison, only School A that had both a control and experimental group was included. All other schools that did not have a control group

57

associated with the experimental group were left out of the analysis in order to understand the project impact more clearly.

The second analysis set included experimental data from all schools to see if the same significance was present between pre- and post-survey responses from the full "All-School" experimental group for those same questions that showed significant differences between responses for the experimental and control groups at School A. This analysis also provided responses that were significant for the All School experimental only group that was not present in the School A experimental group versus control group analysis.

When analyzing the results for this research study, it is important to note that School A has been working over the last couple years to integrate STEM, especially the engineering design process into the various grades within the school. The experimental group at School A was part of the STEM track whereas the control group was part of the non-STEM, more traditional track. While overall the STEM track is still new, and the number of actual activities is still limited, it is possible that the increased exposure to STEM activities resulted in a difference for these students.

CHAPTER V

RESULTS & DISCUSSION

This chapter summarizes the statistically significant changes that were observed when completing a statistical analysis on every question that was asked of the students during the 'You're Hired!' pre- and post-surveys. In each section, the results are objectively stated, followed by a brief discussion of the results.

Section 5.1 looks at data from the school with separate experimental and control groups, School A. In this section, responses from an experimental group made up of students that completed both pre- and post-surveys as well as three separate 'You're Hired!' activities is compared to a control group made up of students that completed pre- and post-surveys, but no 'You're Hired!' activities. Section 5.2 looks at the experimental group. The responses from all 5 schools, also referred to as the all-school experimental group. The responses in this section are a representation of every student in every school that participated in the 'You're Hired!' program. Section 5.3 looks into differences that were seen between the all-school experimental group responses when comparing each of the five schools to each other. Section 5.4 presents an analysis of the questions by gender for every student and every school participating in the activities. Finally in Section 5.5 results on the impact of the 'You're Hired!' activities were given a post-survey that had extra 'Impact' questions on it. The

results of what the students thought about the program specifically can be found in this section.

The analysis in this chapter supports the hypothesis that 'You're Hired!' positively influences students' ideas and attitudes of engineering.

5.1 School A - Analysis of Experimental and Control Group Responses

Table 5.1 contains questions from the pre- and post-survey in which there was a statistically significant difference between experimental and control group responses for either survey or between pre and post responses for either group for School A. The questions listed in this table were present on both the pre-survey as well as the post-survey. If a significant difference was determined for any of the comparisons, then the question was included for analysis. The significance differences were determined by the values of the analysis having p-values less than or equal to 0.05. For each category of question, Engineer, Future Work, Interest and Attitudes and Skills there are two tables of data. One table is for the pre- and post-survey comparisons, the other table is for the comparisons between experimental group and control group. It is possible that for the comparisons in one table, the question is not significant, but for the comparisons in the other table the question is significant. Out of the four categories of questions; Engineers, Future Work, Interests, and Attitude and Skills, three categories contained questions that showed significant differences in the survey responses. The three categories that contained significant shifts were; Engineer, Interest, and Attitude and Skills. Statistical analysis of the different questions can be found in Figures

5.1-5.3 and Tables 5.2 – 5.7. Statistical analysis of every question, including

questions that were not significantly different, can be found in Appendix C.

Table 5.1. Statements Showing Statistically Significant Differences in Responses between either Experimental and Control Groups for either Survey or between Pre- and Post-Responses for either Group for School A.

Eng	ineer Questions					
E1	Mainly work on machines and computers					
E2	Mainly work with other people to solve problems					
E4	Can choose to do many different kinds of jobs					
Inte	Interest Questions					
In1	I look forward to science class in school					
In4	More time should be spent on hands-on projects in science or technology					
	activities at school					
In5	I would like to (or already do) belong to a science or technology activities club					
In8	I like learning how things work					
In9	Science is too hard when it involves math					
Atti	Attitude and Skills Questions					
A4	I can effectively lead a team to design and build a hands-on project					
A8	I can get good grades in science					

5.1.1 Engineer Question

Figure 5.1 and corresponding tables, Table 5.2 and Table 5.3, show a statistical comparison of responses for the Engineer statements that displayed statistically significant differences. The statement from the surveys for this category read, 'Read the following statements about what engineers might do and indicate your level of agreement or disagreement with each statement.' Response choices for these statements were converted to a Likert scale in order to further compare the responses, with 'Agree'=3, 'Don't Know'=2, and 'Disagree'=1. It was decided to put 'Don't Know' as an intermediate value on the Likert scale because it was more of a neutral response between the 'Yes' and 'No' responses.



Figure 5.1. Comparison of Mean Values for School A's Experimental and Control Group Responses for Engineer Statements on Pre- and Post-Surveys. A) Statement E1 – Mainly work on machines and computers, B) Statement E2 – Mainly work with other people to solve problems, C) Statement E4 – Can choose to do many different kinds of jobs.

Engineers	Group	N	P-Value	X ²	Pre-Survey Mean	Post-Survey Mean
E1 - Mainly work on	С	270	0.74	0.60	2.20	2.29
machines and computers	Е	271	0.30	2.42	2.54	2.42
E2 - Mainly work with other	С	267	0.30	2.40	2.31	2.39
people to solve problems	*Е	269	< 0.01	17.52	2.36	2.74
E4 - Can choose to do many	*C	266	0.05	6.21	2.37	2.57
different kinds of jobs	*Е	270	< 0.01	13.18	2.43	2.74

Table 5.2. Statistical Comparison of Pre- and Post-Survey Responses to Engineer Questions for Control and Experimental Groups at School A.

*Significance: p-value less than or equal to 0.05 C: 'Control Group' and E: 'Experimental Group' Degrees of Freedom (DF) =6

Figure 5.1 summarizes the data in numerical form in which the dots represent the possible answer choices. For example with this set of Engineer statements, there were three answers the students could choose from: 'Agree' (3), 'Don't Know' (2), and 'Disagree' (1). The middle line for each group is the mean for that particular group's responses. The green, horizontal outer lines for each set of responses represent the upper and lower 95% confidence intervals around the mean response. The x-axis contains the pre- and post-responses of each of the two groups, control and experimental. The y-axis displays the answer options in Likert scale format. The categorical data was converted to numerical data using the Likert Scale to provide another analytical method, comparison of means.

Tables 5.2 and Table 5.3 provide a statistical comparison of the categorical responses to the Engineer questions. P-values less than or equal to 0.05 are considered significant. The pre- and post-survey mean values are also included and represent the same value that is shown in the figures as the middle line for each group. Table 5.2 displays the analysis based on each group with p-value and chi² statistics representing the difference between pre- and post-survey

responses, for example comparing control pre-survey versus control post-survey for statement E4, the statistical analysis results in a p-value of 0.05. Table 5.3 displays the analysis based on survey, with statistics representing the difference between control and experimental group responses for the pre- or post-survey.

As shown in Table 5.2 for statement E1, 'Engineers mainly work on machines and computers', the p-value indicates that there was no significant shift in either group from pre- to post-survey responses. For statement E2, 'Engineers mainly work with other people to solve problems' there was a significant shift between the pre-survey response mean of 2.36 to the post-survey response mean of 2.74 for the experimental group, but not for the control group. The shift to higher values for the experimental group indicates that, on average the students in that group agreed more with the statement 'Engineers mainly work with other people to solve problems' when taking the post-survey than when they took the pre-survey. For statement E4, 'Engineers can choose to do many different kinds of jobs' there was a significant shift for both groups between survey responses. The control group's responses had a shift in the mean numerical value of 0.20 and the experimental group's responses shifted by 0.31.

Table 5.3. Statistical Comparison of Control and Experimental Responses to Engineer Questions for Pre- and Post-Surveys at School A.

					Control	Experimental
Engineers	Survey	Ν	P-Value	X ²	Mean	Mean
E1 - Mainly work on	*Pre	269	< 0.01	12.93	2.20	2.54
machines and computers	Post	272	0.38	1.93	2.29	2.42
E2 - Mainly work with other	Pre	265	0.89	0.24	2.31	2.36
people to solve problems	*Post	271	< 0.01	14.27	2.39	2.74
E4 - Can choose to do many	Pre	264	0.68	0.76	2.37	2.43
different kinds of jobs	Post	272	0.08	5.10	2.57	2.74

*Significance: p-value less than or equal to 0.05 DF = 6

Table 5.3 presents the same statements as shown in Figure 5.1 and Table 5.2, but analysis statistics look at the difference between control and experimental group responses. For statement E1, 'Engineers mainly work on machines and computers', the p-values indicate that there was a significant difference in the way that the experimental group and the control group responded to the pre-survey question. For all responses, the mean values were between the response options of 'Don't Know'=2 and 'Agree'=3. The experimental group agreed with a mean response of 2.54 for the question whereas the control group's mean response was 2.20. The statistical difference in the two group's responses was not observed in the post-survey question. For E2, 'Engineers mainly work on with other people to solve problems', there was not a significant difference between the two groups for the pre-survey, but there was a statistical difference for the post-survey responses. The experimental group's mean response increased from the pre-survey by 0.38 to result in a post-survey mean of 2.74. The control group's mean response for this question increased by only 0.08 resulting in a post-survey mean of 2.39. For statement E4, 'Engineers can choose to do many different kinds of jobs' there was neither a significant difference between groups for the pre-survey nor for the postsurvey responses. This statement was significant when looking at the data by group and not by survey.

Engineer Question Discussion

E1 - 'Engineers mainly work on machines and computers' did not have a significant shift for either group from pre- to post-survey, but there was a significant difference in how the two groups responded to the statement for the

pre-survey. The experimental group responded statistically higher than the control group. There is a common misconception that while engineers do in fact spend time working on machines and computers, it is not the only type of work environment that is available to them. The post-survey responses for the two groups resulted in less of a difference as the two groups' attitudes and understanding shifted slightly.

The statement E2 - 'Engineers mainly work with other people to solve problems' had only the experimental group showing a statistically significant shift from pre- to post-survey responses. When looking at the statement by survey, there was no statistical difference between the experimental and control groups when looking at pre-survey responses, but there was a statistically significant difference in how the two groups responded in the post-survey. This change in understanding and attitude is thought to be a result of the 'You're Hired!' program because throughout the program students are encouraged by program designers (if present), teachers, and boardroom members to collaborate and work with their teammates no matter what career choice the students choose for the day. This suggests that the 'You're Hired!' program made a positive impact on the participating students' perceptions of engineers and engineering careers.

E4 - 'Engineers can choose to do many different kinds of jobs' showed a statistically significant shift for both the experimental and the control groups from pre- to post-survey responses. When comparing the surveys based on group, there was no significant difference for either survey. The results for this statement would indicate that while students did have a better understanding of the variety of possible engineering careers at the end of the program, that increase cannot be linked to the 'You're Hired!' program due to both the control and the experimental groups' levels of agreement increasing from pre- to post-survey.

5.1.2 Interest Question

Figure 5.2 and corresponding tables, Table 5.4 and 5.5 show the statistical comparison of pre-and post-survey responses for the Interest questions. The statement for this category of questions read, 'Here is a list of statements. Tell us what you think about them. Select a response that indicates your level of agreement.' Responses for these statements were converted to a Likert scale in order to further compare the responses, with 'Strongly Agree'=4, 'Somewhat Agree'=3, 'Somewhat Disagree'=2, and 'Disagree'=1.



Figure 5.2. Comparing Control vs. Experimental responses for Interest Questions. A) In1 - I look forward to science class in school, B) In4 - M ore time should be spent on hands-on projects in science or technology activities at school, C) In5 - I would like to (or already do) belong to a science or technology activities club, D) In8 - I like learning how things work, E) In9 - Science is too hard when it involves math.

68

Interests	Group	Ν	P-Value	X ²	Pre-Survey Mean	Post-Survey Mean
In1 - I look forward to science class in school In4 - More time	C E	269 269	0.46 0.54	2.56 2.16	2.90 3.41	2.78 3.43
should be spent on hands-on projects in science or technology activities at school	C *E	268 268	0.84 <i>a</i>	0.83 <i>a</i>	3.29 3.63	3.23 3.64
In5 - I would like to (or already do) belong to a science or technology activities club	C E	266 268	0.69 0.82	1.45 0.93	2.02 2.71	1.90 2.64
In8 - I like learning how things work	C E	267 265	0.22 0.32	4.39 3.54	3.26 3.32	3.07 3.39
In9 - Science is too hard when it involves math	С *Е	264 270	0.88 0.01	0.69 11.52	2.21 1.98	2.17 2.33

Table 5.4. Statistical Comparison of Pre- and Post-Survey Responses to Interest Questions for Control and Experimental Groups at School A.

*Significance: p-value less than or equal to 0.05 C: 'Control Group' and E: 'Experimental Group' DF = 9

 a - A reliable chi² and p-value could not be calculated for In4 – 'More time should be spent on hands-on projects in science or technology activities at school' because one of the answer options received less than five responses.

For the statements In1, In5 and In8, 'I look forward to science class in school', 'I would like to (or already do) belong to a science or technology activities club' and 'I like to learn how things work' the p-value indicates that there was no difference for either group from pre- to post-survey responses. The significance for these questions will be observed in Table 5.5.

For statement In9, 'Science is too hard when it involves math', there was a significant difference between survey responses for the experimental group, but not for the control group. When looking at the experimental group's survey responses, the pre-survey mean was 1.98 and over the course of the school year, that mean responses increased by 0.35 resulting in a post-survey mean of 2.33. A statistically significant difference was not observed between the experimental and control group's pre- to post-survey responses but there was a significant difference in the pre-survey between how the two groups responded.

Table 5.5. Statistical Comparison of Control and Experimental Responses	to
Interest Questions for Pre- and Post-Surveys at School A.	

Interests	Survey	N	P-Value	X ²	Control Mean	Experimental Mean
In1 - I look forward to science class in school	*Pre *Post	270 268	<0.01 <0.01	30.88 44.15	2.90 2.78	3.41 3.43
In4 - More time should be spent on hands-on projects in science or technology activities at school	*Pre *Post	271 265	<0.01 <0.01	22.34 21.16	3.29 3.23	3.63 3.64
In5 - I would like to (or already do) belong to a science or technology activities club	*Pre *Post	267 267	<0.01 <0.01	30.71 35.04	2.02 1.90	2.71 2.64
In8 - I like learning how things work	Pre *Post	267 265	0.10 0.02	6.20 10.41	3.26 3.07	3.32 3.39
In9 - Science is too hard when it involves math	*Pre Post	268 266	0.03 0.24	8.66 4.20	2.21 2.17	1.98 2.33

*Significance: p-value less than or equal to 0.05

DF = 9

For the statements In1, In4, and In5, 'I look forward to science class in

school', 'More time should be spent on hands-on projects in science or

technology activities at school', and 'I would like to (or already do) belong to a science or technology activities club' there was a significant difference between the control group and the experimental group for both pre-survey and post-survey responses. For statement In8, 'I like learning how things work', there was not a significant difference between groups when looking at the pre-survey responses, but there was a significant difference between the control group's mean of 3.07 and the experimental group's mean of 3.39 for the post-survey responses. When looking at the statement In9, 'Science is too hard when it involves math', there was a significant difference between groups when looking at the pre-survey responses, but not the post-survey responses. For the In9 statement, the difference between the two group's mean values for the pre-survey was 0.23 and for the post-survey the difference had decreased to 0.16 which resulted in the group's mean responses no longer being significantly different.

When looking at the changes in responses between pre-survey and postsurvey by group, 1n4- 'More time should be spent on hands-on projects in science or technology activities at school' and 1n9- 'Science is too hard when it involves math' were significant for the experimental group, but not for the control group.

Interest Question Discussion

Out of the five statements that showed a statistically significant difference either between 1) the experimental and control group responses from pre- to postsurvey or 2) the pre- and post-survey responses when comparing the groups, four of the five statements had pre-survey responses that were statistically different. This differentiation from the pre-survey responses was not present for all categories of questions/statements asked of the students, but for this set of Interest statements it was noted that the two groups did not have similar responses at the beginning of the study indicating that in terms of interests, these two groups of students were originally statistically different. It is possible that the difference in the students' responses may be due to the pod/house the students are in throughout the school year. School A is working to develop a STEM track within their middle school. The students that participated in the program were part of that STEM track. While the track is still fairly new, it is possible that the type of students attracted by this particular way of learning may have caused the statistical differences between group responses.

For the statement, In9 - 'Science is too hard when it involves math', there was a significant increase in the numerical mean from pre- to post-survey for the experimental group. This increase would suggest a negative shift in attitude towards science that involves math, however it is possible that this shift is related more to the everyday classes the students take as opposed to the supplementary program of 'You're Hired!'. This statement talks about science specifically indicating that it might have been influenced by experiences in a formal science class.

For the statement, In8 - 'I like learning how things work', the pre-survey showed no significant difference between the two groups, however after the course of the year and the implementation of the 'You're Hired!' program, the post-survey response shows a significantly higher level of agreement for the experimental group. This shift may be a result of the 'You're Hired!' program providing the participating students with the opportunity to problem solve realworld examples that affect everyday life.

5.1.3 Attitude and Skills Question

Figure 5.3 and corresponding tables, Table 5.6 and 5.7 show the statistical comparison of pre-and post-survey responses for the Attitude and Skills questions. The statement for this category of questions read, 'The table lists things you can do when you are working on school activities or assignments. Check the appropriate box to tell us how often you do each of these things.' Responses for these statements were converted to a Likert scale in order to further compare the responses, with 'Always'=4, 'Very Often'=3, 'Sometimes'=2, and 'Never'=1.



Figure 5.3. Comparing Control vs. Experimental Responses for Attitude and Skills Questions. A) A4 - I can effectively lead a team to design and build a hands-on project, B) A8 - I can get good grades in science.

Table 5.6.	Statistical Comparison of Pr	e- and Post-Surve	y Responses to Att	itude
and Skills	Questions for Control and E	xperimental Group	os at School A.	

Attitude and Skills	Group	N	P-Value	X ²	Pre-Survey Mean	Post-Survey Mean
A4 - I can effectively lead a team to design and build a hands-on project	C E	268 267	0.45 0.15	2.66 5.35	2.39 2.85	2.53 2.85
A8 - I can get good grades in science	C E	265 266	0.71 0.73	1.36 1.30	2.92 3.35	2.89 3.25

*Significance: p-value less than or equal to 0.05

C: 'Control Group' and E: 'Experimental Group' DF = 9

As shown in Table 5.6, for the statements A4 and A8, 'I can effectively

lead a team to design and build a hands-on project' and 'I can get good grades in

science', there was not a significant shift between pre- and post-survey responses

for either control or experimental group responses.

Table 5.7. Statistical Comparison of Control and Experimental Responses to Attitude and Skills Questions for Pre- and Post-Surveys at School A.

Attitude and Skills	Survey	N	P-Value	X ²	Control Mean	Experimental Mean
A4 - I can effectively lead a team to design and build a hands-on project	*Pre *Post	269 266	0.00 0.03	14.23 8.71	2.39 2.53	2.85 2.85
A8 - I can get good grades in science	*Pre *Post	267 264	$\begin{array}{c} 0.00\\ 0.00 \end{array}$	19.10 13.59	2.92 2.89	3.35 3.25

*Significance: p-value less than or equal to 0.05 DF = 9

When comparing control and experimental groups by survey, there was a

significant difference in responses to the statements A4 and A8. For each

statement in each survey, the experimental group had a higher level of agreement when compared to the control group.

Attitude and Skills Question Discussion

When looking at the changes in responses between pre-survey and postsurvey by groups, neither statement was significant. For changes in responses between groups by survey, both A4 - 'I can effectively lead a team to design and build a hands-on project' and A8 - 'I can get good grades in science' were significant for both pre- and post-survey responses. These results indicate that there was a statically significant difference between the two groups at the beginning of the program as well as at the end of the program. This analysis indicates that the 'You're Hired!' program did not have an effect on students' perceptions of their skills.

5.2 Experimental Group Results for the Five Participating Schools

The significant differences between an experimental group and a control group described in Section 5.1 only include one school's responses, School A. There is a larger population of students that participated (School B – School E) in the program that did not have an associated control group. When looking at presurvey data for all students participating in 'You're Hired!', there are significant differences between how the various schools respond. These significant differences between schools are not consistent with all questions. One question can have no significant difference between how the schools respond and the next question could result with some schools being statistically different from the others. A full list of the differences between pre-survey responses can be found in Appendix D.

This section describes statistically significant shifts in responses from preto post-survey when considering all participating schools, School A – School E, as a single experimental group, referred to as the "all-schools experimental group". Table 5.8 contains questions that show a statistical difference between pre-and post-survey responses for the all-school experimental group. Of the four categories of questions, three had at least one question with responses that were significantly different between pre-and post-survey. The 'Future Work' category was the only category to not have a significant difference in responses to statement within the category. Analysis of the statistically different questions can be found in Figure 5.4 –5.6 and Table 5.9 –5.11. Further analysis of every question including questions that were not significantly different from pre-to postsurvey responses can be found in Appendix E.

Table 5.8. Statements Showing Significantly Different Responses for the All-Schools Experimental Group between Pre-Survey and Post-Survey.

Engi	Engineers					
E1	Mainly work on machines and computers					
E2	Mainly work with other people to solve problems					
E4	Can choose to do many different kinds of jobs					
Inter	est					
In10	Science is a difficult subject					
Attitude and Skills						
A2	I can use what I know to design and build something mechanical that					
	works					

5.2.1 Engineer Question

Figure 5.4 and corresponding Table 5.9 show a statistical comparison of the all-school experimental group responses between pre-survey and post-survey for the significant Engineer statements.



Figure 5.4. Comparing All-School Experimental Responses for Engineer Questions. A) E1 - Mainly work on machines and computers, B) E2 - Mainly work with other people to solve problems, C) E4 - Can choose to do many different kinds of jobs.

).01 9.9	93 2.40	2.23					
0.01 9.9	93 2.40	2.23					
0.01 36.2	22 2.36	2.62					
0.01 25.1	11 2.49	2.69					
*E4 - Can choose to do many different kinds of jobs 554 <0.01 25.11 2.49 2.69 *Significance: p-value less than or equal to 0.05							

Table 5.9. Statistical Comparison of All-School Experimental Group's Responses to Engineer Questions for Pre- and Post-Surveys at all Participating Schools.

DF = 2

For statement E1, 'Engineers mainly work on machines and computers', the shift was from a pre-survey numerical mean of 2.40 to a lower, less approving post-survey mean of 2.23. The statements E2 and E4, 'Engineers mainly work with other people to solve problems' and 'Engineers can choose to do many different kinds of jobs' both had positive shifts from the student's pre-survey mean to the post survey mean.

Engineer Question Discussion

The quantitative analysis of responses from all students indicates that statements E1 - 'Engineers mainly work on machines and computers', E2 -'Engineers mainly work with other people to solve problems' and E4 - 'Engineers can choose to do many different kinds of jobs' show statistically significant preto post-survey changes. Responses to these three statements suggest that students had a more accurate understanding of engineering when they took the post-survey versus the pre-survey. The significance of E2 and E4 for the all-school experimental group correlates with the significant shifts of the experimental group from School A as described in Section 5.1.

In Section 5.1 the experimental group for School A had positive shifts for statements E2 and E4. In Section 5.2 the all-school experimental group also had positive shifts for statements E2 and E4.

Statement E1 resulted in a statistically significant shift from pre- to postsurvey for the all-school experimental group, but not for the experimental group from School A. It is possible that due to the smaller sample size of students from School A, the change in students' response for this statement was not statistically different. The much larger sample size of the all-school experimental group suggests that these results could be an indication that 'You're Hired!' did in fact have an influence on students and that this impact was not seen in the analysis of only School A. A larger experimental/control sample size would help to clarify this difference and will be addressed in future work.

5.2.2 Interests Question

Figure 5.5 and corresponding Table 5.10 show the statistical comparison of the all-school experimental group responses between pre-survey and postsurvey for the one Interest statement with a significant pre- to post-survey difference, 'Science is a difficult subject.'



Figure 5.5. Comparing All-School Experimental Group's Responses for an Interest Statement, In10 – Science is a difficult Subject.

Table 5.10. Statistical Comparison of All-School Experimental Group's Responses to Interest Questions for Pre- and Post-Surveys at All Participating Schools.

	Interests	Ν	P-Value	X ²	Pre-Survey Mean	Post-Survey Mean	
	*In10 - Science is a						
	difficult subject	551	0.04	8.46	1.98	2.13	
*Significance: p-value less than or equal to 0.05							
Ľ	$\mathbf{DF} = 3$		-				

Interest Question Discussion

The significant increase shifted from a pre-survey mean response between 'Disagree' and 'Somewhat Disagree' to a post-survey mean response between 'Somewhat Disagree' and 'Somewhat agree'. This shift appears to be in a direction that is unfavorable after the students had participated in the 'You're Hired!' program.

When reflecting on this result as well as the results of School A from Section 5.1, there are some similarities I noticed. The statement, 'Science is too hard when it involves math' from Section 5.1 showed a significant increase in the level of agreement from pre-survey to post-survey responses. The all-school experimental group analysis in this section resulted in a different significant statement, 'Science is a difficult subject'. Similar to the shift observed in Section 5.1, this statement from Section 5.2 also had an increase in the level of agreement from the pre-survey to the post-survey. Both statements appear to show negative attitudes towards science class. The 'You're Hired!' activities were specifically designed to remove the various walls that are currently dividing the essential core classes to blend together science, technology, engineering and math as well as English, computer, and geography/history in such a way that students understood that the core classes are important and can be related to real-world applications. Because this statement about science specifically had a shift in the unfavorable direction it is possible that the shift in responses may have had nothing to do with the 'You're Hired!' program and instead is related more to the classroom experiences the students had throughout the year. Future work involving a panel of students giving feedback on the survey tool may help to clarify this change.

5.2.3 Attitude and Skills Question

Figure 5.6 and corresponding Table 5.11 show the statistical comparison of experimental group responses from all five schools between pre-survey and post-survey for the significant Attitude and Skills statement.



Figure 5.6. Comparing All-School Experimental Group's Responses for an Attitude and Skills Statement, A2 - I can use what I know to design and build something mechanical that works.

Table 5.11. Statistical Comparison of Experimental Group's Responses to Attitude and Skills Questions for Pre- and Post-Surveys at all Participating Schools.

Attitude and Skills	N	D Voluo	\mathbf{V}^2	Pre-Survey	Post-Survey
Attitude and Skins A2 - I can use what I	19	I - Value	Λ	Wiean	Ivicali
know to design and build something mechanical	554	< 0.01	17.24	2.30	2.43
that works					

*Significance: p-value less than or equal to 0.05 DF = 3

For the statements listed in Table 5.11 there was a statistically significant

shift from pre-to post-survey responses for the group of students that completed

the 'You're Hired!' program.

Attitude and Skills Question Discussion

Both pre- and post-survey mean responses were between 'Sometimes' and

'Very Often'. It is possible that this positive increase is a result of the 'You're

Hired!' program because a majority of the schools that were part of the project do

not have the opportunity to problem solve in a hands-on learning environment in

their everyday classes. Because this program allowed students to problem solve to find a possible solution to a given problem, it is possible that the students' level of confidence increased.

5.3 Experimental Group Responses When Comparing Each of the Five Schools to Each Other

For the questions in which the all-school experimental group analysis in Section 5.2 did not result in similar observations to that of the control vs experimental analysis in Section 5.1, the statements were further broken down by school to indicate if one or two school's responses were greatly different than School A's experimental responses. Results of further analysis by school can be found in this section. A full list of the significant differences between pre-survey responses for the experimental group from all five schools can be found in Appendix D.

Table 5.12 contains the questions that were either significant for the experimental group in School A (Section 5.1) but not significant for the all-school group (Section 5.2) or questions that were significant for the all-school group but not for School A. The only category that had statements that were significant for both School A's experimental group as well as the all-school experimental group was the Engineer statements E2 and E4. Statistical analysis of the significantly different responses can be found in Figure 5.7 and Table 5.13. Further analysis of every question can be found in Appendix F.



Figure 5.7. Comparing Experimental Responses for Questions that were not significant for both School A and School B – School E Experimental groups. A) E1 - 'Mainly work on machines and computers', B) In9 - 'Science is too hard when it involves math', C) In10 - 'Science is a difficult subject' and D) A2 - 'I can use what I know to design and build something mechanical that works'

Table 5.12. Statements Showing Significantly Different Responses between Pre-Survey and Post-Survey for Section 5.1 Experimental Group Compared and Section 5.2 Experimental Groups.

Engineers		
E1	Mainly work on machines and computers	
Inter	est	
In9	Science is too hard when it involves math	
In10	Science is a difficult subject	
Attitude and Skills		
A2	I can use what I know to design and build something mechanical that works	

Table 5.13. Statistical Comparison of Experimental Responses for Questions that were not significant for both School A and School B – School E Experimental groups.

				Pre-Survey	Post-Survey
Engineer	School	P-Value	X ²	Mean	Mean
	А	0.30	2.42	2.37	2.35
	В	0.56	1.15	2.46	2.31
E1 - Mainly work on machines	*C	0.04	6.33	2.34	2.15
	D	0.13	4.05	2.43	2.07
	Е	0.18	1.80	1.89	2.33
Interests					
	*A	0.01	11.52	1.98	2.33
	В	0.65	1.66	2.07	2.23
In9 - Science is too hard when it involves math	С	0.11	6.08	2.29	2.29
it involves math	D	0.85	0.80	2.17	2.10
	E	0.39	3.04	1.78	1.94
	А	0.13	5.59	1.78	2.01
In10 - Science is a difficult	В	0.32	3.54	1.96	2.23
subject	С	0.15	5.34	2.10	2.21
	D	0.81	0.97	2.06	1.89
	E	0.62	0.97	1.72	1.88
Attitude and Skills					
	*A	0.02	9.59	2.48	2.55
A2 - I can use what I know to	В	0.06	7.35	2.24	2.57
design and build something	С	0.07	7.13	2.08	2.23
	D	0.60	1.89	2.81	2.87
	Е	0.86	0.31	3.06	2.94

*Significance: p-value less than or equal to 0.05

Figure 5.7 displays information in the same format as the previous figures in this chapter, but with more groups. Figure 5.7 along with Table 5.13 contain the significant statements that were not consistent between the Section 5.1 analyses of the control School A compared to the Section 5.2 analysis of all participating schools, School A – School E.

For statement E1, School C contributed to the question being significant for the all-school experimental group. While most of the other schools exhibited similar changes in mean values, only School C's shift was significant. For this statement, all schools but one resulted in a decrease in the level of agreement from pre- to post-survey responses. School E responded with a mean that was 0.44 higher in the post-survey.

For statements In9 and A2, School A responded statistically different than the other participating schools. Statement In10 resulted in a significant shift from pre- to post-survey for the all-school experimental group from Section 5.2, however when breaking the responses down by individual school, there is not one school that consistently responds differently than the schools.

Section 5.3 Discussion

For three out of the five statements, there is a significant difference between School A's responses, but not any of the other schools' responses. This may be due to the fact that School A's experimental group is considered to be a STEM track. While overall the STEM track is still new, and the number of actual activities is still limited, it is possible that the increased exposure to STEM activities resulted in a difference for these students. School B is also very similar to School A in the way that it has started a STEM track and that STEM group is the ones that participated in the project. School B was not statistically different than any other school for the five statements in this section. It is possible that School A spent more time and resources on developing their STEM track which resulted in their students having statistically different responses in three of the five statements as compared to the other participating schools.

5.4 Analysis Based on Gender

It is important to not only understand the different perceptions of engineering based on schools, but also on gender. This section looks into significant shifts in responses from pre- to post-survey for the all-school experimental group involving School A – School E by gender.

Table 5.14 contains questions from the pre- and post-survey where there was a statistically significant difference between female and male responses for either survey or between pre and post responses for either gender. If a significant difference was determined for any of the comparisons, then the question was included for analysis. For each category of question, Engineer, Future Work, Interest and Attitudes and Skills there are two tables of data. One table is for the pre- and post-survey comparisons, the other table is for the comparisons between female and male responses. It is possible that for the comparisons in one table, the question is not significant, but for the comparisons in the other table the question is significant. Out of the four categories of questions; Engineers, Future Work, Interests, and Attitude and Skills, all four contained questions that showed significant differences in the survey responses. Statistical analysis of the different

questions can be found in Figures 5.8-5.11 and Tables 5.15 –5.22. Statistical

analysis of every question, including questions that were not significantly

different, can be found in Appendix G.

Table 5.14. Statements Showing Statistically Significant Different Responses between either Female and Male Responses for either Survey or between Pre- and Post- Responses for either Gender for the All-School Experimental Group.

Engir	Engineer Question			
E1	Mainly work on machines and computers			
E2	Mainly work with other people to solve problems			
E4	Can choose to do many different kinds of jobs			
E5	Mainly work on things that have nothing to do with me			
E6	I don't know what engineers do			
E7	Pursue a career in an engineering-related field			
E8	Do you think you want to be an engineer			
Futur	e Work Question			
F2	Work that allows me to make lots of money			
F8	Work that allows me to help my community and/or society			
Intere	ests Question			
In5	I would like to (or already do) belong to a science or technology activities			
	club			
In6	I get bored when I watch programs on channels like Discovery Channel,			
	Animal Planet, Nova, Mythbusters, etc.			
In7	I like to get science books or science experiment kits as presents			
In8	I like learning how things work			
In13	I am interested in learning more about how things work			
In14	I like to learn to use new technology			
Attitu	Attitude and Skills Question			
A2	I can use what I know to design and build something mechanical that			
	works			
A3	In lab activities, I can use what I have learned to design a solution			
A5	I know where I can find the information that I need to solve difficult			
	problems			
A7	I can get good grades in math			
Impa	ct Question			
Im3	Increased my interest in studying engineering in college			
Im4	Increased my interest in studying a technical degree in college			
Im8	Made me confident in my ability to succeed in an engineering or			
	technical field			




5.4.1 Engineer Question

Figure 5.8 and corresponding Table 5.15 and Table 5.16 show a statistical

comparison of pre- and post-survey responses for the significant Engineer

statements. The only statement that was not significant for the Engineer question

was, 'Work on things that help the world'. Both female and male students

responded highly for this statement in both the pre-survey and post-survey.

Table 5.15. Statistical Comparison of Female and Male Responses to Engineer Statements for All Participating Schools.

					Pre-Survey	Post-Survey
Engineers	Gender	Ν	P-Value	X ²	Mean	Mean
E1-Mainly work on						
machines and	Female	251	0.06	0.12	2.32	2.17
computers	Male	309	0.03	1.22	2.46	2.29
E2-Mainly work with						
other people to solve	*Female	250	< 0.01	18.23	2.39	2.62
problems	*Male	304	< 0.01	19.59	2.33	2.61
E4-Can choose to do	*Female	250	< 0.01	15 49	2.52	2.71
many different kinds of	1 emaie	200	-0.01	15.15	2.32	2.71
jobs	*Male	304	< 0.01	14.30	2.45	2.68
E5-Mainly work on	Famala	250	0.13	4 10	1.62	1.61
things that have nothing	*Mala	200	0.15	4.10	1.02	1.01
to do with me	Wale	302	0.02	1.97	1.59	1.70
E6-I don't know what	Female	242	0.94	5.56	1.58	1.55
engineers do	*Male	293	0.54	6.75	1.45	1.39
E7-Pursue a career in an						
engineering-related	Female	253	0.68	0.77	1.61	1.63
field?	Male	311	0.89	0.23	2.08	2.11
	F 1	252	0.46	1.50	1.45	1.50
E8-Do you think you	Female	253	0.46	1.56	1.45	1.50
want to be an engineer?	Male	311	0.92	0.16	1.97	1.99

*Significance: p-value less than or equal to 0.05DF = 6

For the statements, E2 - Engineers mainly work with other people to solve problems' and E4 - Engineers can choose to do many different kinds of jobs' there was a significant shift in both female and male responses from pre- to post-survey. For the statements, E6 - I don't know what engineers do' there was a significant decrease in the level of agreement for males that responded, for E5 - 'Engineers mainly work on things that have nothing to do with me' there was a

significant increase in the male responses.

Table 5.16. Statistical Comparison of Pre- and Post-Survey Responses to Engineer Statements for Females and Males at All Participating Schools.

					Female	Male
Engineers	Survey	Ν	P-Value	X ²	Mean	Mean
E1-Mainly work on						
machines and	*Pre	559	0.04	6.29	2.32	2.46
computers	Post	554	0.22	3.03	2.17	2.29
E2-Mainly work with						
other people to solve	Pre	535	0.27	2.62	2.39	2.33
problems	Post	560	0.86	0.30	2.62	2.61
E4-Can choose to do						
many different kinds	Pre	556	0.17	3.56	2.52	2.45
ofjobs	Post	559	0.25	2.74	2.71	2.68
E5-Mainly work on						
things that have	Pre	554	0.53	1.29	1.62	1.59
nothing to do with me	Post	564	0.27	2.59	1.61	1.70
E6-I don't know what	Pre	553	0.15	3.80	1.58	1.45
engineers do	*Post	556	0.02	7.56	1.55	1.39
E7-Pursue a career in						
an engineering-related	*Pre	552	< 0.01	48.16	1.61	2.08
field?	*Post	554	< 0.01	48.09	1.63	2.11
E8-Do you think you						
want to be an	*Pre	551	< 0.01	61.00	1.45	1.97
engineer?	*Post	558	< 0.01	54.88	1.50	1.99

*Significance: p-value less than or equal to 0.05 DF = 6

Statements E7 and E8, 'Pursue a career in an engineering-related field' and 'Do you think you want to be an engineer' showed significant differences in how females responded versus males for both the pre- and post-surveys. Neither of these questions resulted in a significant shift between pre- and post-survey for a specific gender. The significant difference remained between how females answered compared to how males answered. Statement E6, 'I don't know what engineers do' there was not a significant difference between gender responses in the pre-survey, but there was for the post-survey. For the statement, E1 – 'Engineers mainly work on machines and computers' there was a significant difference between how males responded versus the females in the pre-survey but not the post-survey.

Engineer Question Discussion

Responses to statement E2 - ' Engineers mainly work with other people to solve problems' and E4 - 'Engineers can choose to do many different kinds of jobs' appear to be a result of the 'You're Hired!' program. The change from preto post-survey response is consistent with both the experimental versus control results for School A as stated in Section 5.1 as well as the all-school experimental results as stated in Section 5.2.

Male responses to statement E7 - 'Pursue a career in an engineeringrelated field' and E8 - 'Do you think you want to be an engineer' were statistically higher than female responses for both pre-survey and post-survey. Work still needs to be done to address current stereotypes that result in fewer women wanting to pursue a career in an engineering-related field. The 'You're Hired!' program is designed to address those stereotypes by providing real-world topics that interest both female and male students.

5.4.2 Future Work Question

Figure 5.9 and corresponding Table 5.17 and Table 5.18 show a statistical comparison of pre- and post-survey responses for the significant Future Work statements. Both F2 - Work that allows me to make lots of money' and F8 -

'Work that allows me to help my community and/or society' had a significant difference in responses when looking at changes by gender.



Figure 5.9. Comparing Female and vs. Male Responses for Future Work Statements. A) F2 - 'Work that allows me to make lots of money' and B) F8 - Work that allows me to help my community and/or society'.

Table 5.17. Statistical Comparison of Female and Male Responses to Future Work Statements for All Participating Schools.

Future Work	Gender	Ν	P-Value	X ²	Pre-Survey Mean	Post-Survey Mean
F2-Work that allows me to make lots of	*Female	252	0.05	6.06	2.52	2.39
money	Male	308	0.78	0.51	2.55	2.54
F8-Work that allows me to help my community and/or society	Female Male	250 306	0.30 0.64	2.39 0.88	2.56 2.39	2.48 2.38

*Significance: p-value less than or equal to 0.05 DF = 6

For the statement F2 – 'Work that allows me to make lots of money', the

female responses dropped significantly from a pre-survey mean of 2.52 to a post-

survey mean of 2.39. Indicating that females decided at the end of the school year

money is less of a reason to choose a specific job.

Future Work	Survey	Ν	P-Value	X ²	Female Mean	Male Mean
F2-Work that allows me to make lots of	Pre	560	0.14	3.93	2.52	2.55
money	*Post	553	0.01	9.49	2.39	2.54
F8-Work that allows						
me to help my community and/or	*Pre	556	0.01	9.53	2.56	2.39
society	Post	551	0.18	3.46	2.48	2.38

Table 5.18. Statistical Comparison of Pre- and Post-Survey Responses to Future Work Statements for Females and Males at All Participating Schools.

*Significance: p-value less than or equal to 0.05DF = 6

Statement F2 – 'Work that allows me to make lots of money' did not show a significant difference in how females versus males responded in the pre-survey, but did in the post-survey. Post-survey responses for this statement had the female level of agreement dropping from pre- to post by 0.13 and the male level of agreement only dropping by 0.01. Statement F8 – 'Work that allows me to help my community and/or society' showed a significant difference between how the genders responded for the pre-survey. The female level of agreement dropped by 0.08 from pre-to post-survey resulting in a mean of 2.48 which was closer to the male level of agreement of 2.38 which was only a 0.01 decrease from the presurvey response.

Future Work Question Discussion

There was a significant change in how females looked at the importance of money from pre-to post-survey. This shift was not apparent in the male responses.

While there was not a significant shift in how the genders themselves responded to importance of helping the community and/or society from pre- to post-survey, the slight decrease in the female level of agreement lead to the change in significance from pre- to post-survey. In the pre-survey the females had a significantly higher level of agreement compared to males that was no longer apparent in the post-survey responses.

While it is not believed that these changes in responses are a result of the 'You're Hired!' program, it is important to keep the category of Future Work with all of its subsequent statements to provide a clearer understanding of the type of students that are participating in the program.

5.4.3 Interest Question

Figure 5.10 and corresponding Table 5.19 and Table 5.20 show a statistical comparison of pre- and post-survey responses for the significant Interest statements.

A) 유민장 inicratis - I would the to the for the advect do) being to a solarge or technology antivities e 4. Pact Pro-Surve 1. Fornale 3. Male Pre-Skry

Gendet-Stryay

o, Mala Pre-Stury

3, Mala Pre-Sury

Poet-Su Gender-Gurvey

Gander-Burysy

C)

briarcete - 1 liko to gat actence books or actence experiments kits as present

Ē)

lphoreste - 1 zm interestad i hing more about how things

1. Fémul Pre-Surv

T. Femalo Pre-Surve 2, Femala



3

Figure 5.10. Comparing Female and vs. Male Responses for Interest Statements. A) In5 - 'I would like to (or already do) belong to a science or technology activities club', B) In6 - 'I get bored when I watch programs on channels like Discovery Channel, Animal Planet, Nova, Mythbusters, etc.', C) In7-'I like to get science books or science experiments kits as presents', D) In8 - 'I like learning how things work', E) In13 - 'I am interested in learning more about how things work', and F) In14-'I like to learn to use new technology'

Table 5.19. Statistical Comparison of Female and Male Responses to Interest Statements for All Participating Schools.

					Pre-Survey	Post-Survey
Interests	Gender	Ν	P-Value	X ²	Mean	Mean
In5-I would like to (or already do) belong to a science or technology activities club	Female Male	248 303	0.11 0.74	6.11 1.26	2.35 2.58	2.20 2.62
In6-I get bored when I watch programs on channels like Discovery Channel, Animal Planet, Nova, Mythbusters, etc.	Female *Male	251 304	0.82 0.05	0.93 7.62	2.20 1.79	2.19 1.97
In7-I like to get science books or science experiments kits as presents	Female Male	249 304	0.98 0.62	0.21 1.79	2.10 2.38	2.13 2.33
In8-I like learning how things work	Female Male	247 300	0.39 0.14	3.03 5.56	3.11 3.39	3.14 3.37
In13-I am interested in learning more about how things work	Female Male	245 303	0.49 0.69	2.41 1.47	2.90 3.28	3.01 3.25
In14-I like to learn to use new technology	*Female Male	246 303	0.00 0.90	13.22 0.60	3.09 3.41	3.14 3.44

*Significance: p-value less than or equal to 0.05

DF = 9

The male responses for statement In6 – 'I get bored when I I watch programs on channels like Discovery Channel, Animal Planet, Nova, Mythbusters, etc.' were significantly different from pre- to post-survey. The level of agreement to the statement increased indicating that at the end of the school year, males were less interested in engaging in these types of programs.

Statement In14 – 'I like to learn to use new technology' had a small, but statistically significant shift from a pre-survey mean of 3.09 to a post-survey response of 3.14 for females. Male levels of agreement did not show a significant shift.

					Female	Male
Interests	Survey	Ν	P-Value	\mathbf{X}^2	Mean	Mean
In5-I would like to (or						
already do) belong to a	Pre	551	0.06	7.55	2.35	2.58
science or technology	*Post	553	< 0.01	21.43	2.20	2.62
activities club						
In6-I get bored when I						
watch programs on channels	*Dre	555	<0.01	29.70	2 20	1 79
like Discovery Channel,	*Post	551	0.01	8 99	2.20	1.75
Animal Planet, Nova,	1 050	551	0.05	0.77	2.17	1.77
Mythbusters, etc.						
In7-I like to get science	*Dre	553	0.01	11 42	2 10	2 38
books or science	Post	550	0.01	5 48	2.10	2.30
experiments kits as presents	1 031	550	0.14	5.40	2.13	2.55
In8-I like learning how	*Pre	547	< 0.01	15.27	3.11	3.39
things work	*Post	550	< 0.01	13.03	3.14	3.37
In13-I am interested in						
learning more about how	*Pre	548	< 0.01	25.64	2.90	3.28
things work	*Post	552	0.01	12.59	3.01	3.25
	*D	540	<0.01	10.15	2.00	2 41
In14-I like to learn to use	*Pre	549	< 0.01	18.15	5.09 2.14	5.41 2.44
new technology	*Post	549	<0.01	22.50	3.14	3.44

Table 5.20. Statistical Comparison of Pre- and Post-Survey Responses to Interest Statements for Females and Males at All Participating Schools.

*Significance: p-value less than or equal to 0.05

DF = 9

Statements In6, In7, In8, In13, and In14 all showed significant differences between how the females responded for each survey when compared to the male responses. For statement In6, 'I get bored when I watch programs on channels like Discovery Channel, Animal Planet, Nova, Mythbusters, etc.' the level of agreement for females was significantly higher for both surveys. For statement In7, 'I like to get science books or science experiments kits as presents' the male level of agreement was significantly higher. Males also had a significantly higher level of agreement to 'I like learning how things work', 'I am interested in learning more about how things work' and 'I like to learn to use new technology'. For statement In5, 'I would like to (or already do) belong to a science or technology activities club' there was not a significant difference between males and females in the pre-survey responses, but there was in the post-survey responses.

Interest Question Discussion

While female responses showed a significant increase in the number of females that stated In14 - 'I like to learn to use new technology', females still appear to have lower levels of interest when it comes to learning more about science and technology and how things work. With more exposure to hands-on activities in which both males and females can gain valuable experiences, it is thought that these levels of agreement will increase for both females as well as males. The other statements in the table are only included due to the fact that there is a significant difference between how males responded to these statements as compared to females for both the pre- and post-surveys.

5.4.4 Attitude and Skills Question

Figure 5.11 and corresponding Table 5.21 and Table 5.22 show a statistical comparison of pre- and post-survey responses for the significant Attitude and Skills statements.



Figure 5.11. Comparing Female and vs. Male Responses for Attitude and Skills Statements. A) A2 - I can use what I know to design and build something mechanical that works', B) A3 - I hab activities, I can use what I have learned to design a solution', C) A5 - I know where I can find the information that I need to solve difficult problems', and D) A7 - I can get good grades in math'.

					Pre-Survey	Post-Survey
Attitude and Skills	Gender	Ν	P-Value	X ²	Mean	Mean
A2-I can use what I						
know to design and build something mechanical that	*Female *Male	250 304	0.01 0.04	10.50 8.32	2.05 2.50	2.20 2.63
works						
A3-In lab activities, I can use what I have learned to design a solution	Female Male	251 306	0.45 0.51	2.63 2.32	2.55 2.84	2.55 2.76
A5-I know where I can find the information that I need to solve difficult problems	Female *Male	250 304	0.17 < 0.01	4.97 13.03	2.61 2.89	2.72 2.66
A7-I can get good grades in math	Female Male	251 305	0.45 0.21	2.64 4.51	3.07 2.99	3.15 2.93

Table 5.21. Statistical Comparison of Female and Male Responses to Interest Statements for All Participating Schools.

*Significance: p-value less than or equal to 0.05 DF = 9

The statement A2 – 'I can use what I know to design and build something mechanical that works' had a significant shift for both females and males from pre- to post-survey. The female level of agreement increased by 0.15 resulting in a post-survey mean of 2.20 and the male level of agreement increased by 0.13 resulting in a post-survey mean of 2.63.

For the statement A5 - iI know where I can find the information that I need to solve difficult problems' there was a significant shift in how the males responded from pre- to post survey. The level of agreement for the males decreased by 0.23. While not significant, the female responses to this question increased from pre-to post-by 0.11.

Attitude and Skills	Survey	Ν	P-Value	X ²	Female Mean	Male Mean
A2-I can use what I						
know to design and	*Pre	554	< 0.01	29.54	2.05	2.50
build something	*Post	544	< 0.01	30.92	2.20	2.63
mechanical that works						
A3-In lab activities, I						
can use what I have	*Pre	557	< 0.01	21.69	2.55	2.84
learned to design a	*Post	544	0.02	10.17	2.55	2.76
solution						
A5-I know where I can						
find the information that	*Pre	554	< 0.01	16.29	2.61	2.89
I need to solve difficult	Post	545	0.50	2.34	2.72	2.66
problems						
A7 I can get good	Pre	556	0.08	6.78	3.07	2.99
grades in math	*Post	545	0.02	9.95	3.15	2.93

Table 5.22. Statistical Comparison of Pre- and Post-Survey Responses to Attitude and Skills Statements for Females and Males at All Participating Schools.

*Significance: p-value less than or equal to 0.05 DF = 9

There was a significant difference in how females versus males responded in statements A2 – 'I can use what I know to design and build something mechanical that works' and A3 – 'In lab activities, I can use what I have learned to design a solution'. For statement A3, female mean levels stayed the same, while male levels slightly decreased. The male responses even after the decrease were still higher than the female responses.

For statement A5 - iI know where I can find the information that I need to solve difficult problems' there was a significant difference in how the genders responded on the pre-survey. Male responses were significantly higher than the female responses. Over the course of the school year, those levels of agreement shifted and while there was no longer a significant difference in the post-survey responses, the level of agreement for males not only decreased, but it resulted in a lower level of agreement than that of the females.

For statement A7 – 'I can get good grades in math' there was not a significant difference observed in the pre-survey results, but there was for the post-survey results. While male levels of agreement decreased by 0.06, the female level of agreement increased by 0.08 resulting in a final difference between the two levels of agreement of 0.22.

Attitude and Skills Question Discussion

I found it very interesting that there was a significant decrease in the male level of agreement to the statement A5 - 'I know where I can find the information that I need to solve difficult problems'. This decrease was opposite of the responses females had to this same statement, however that relationship cannot be concluded because the female response was not statistically significant.

It is thought that because the students participating in 'You're Hired!' choose their own careers and tasks for the individual activities throughout the program that females tend to take more of the 'research' roles and males tend to migrate towards the hands-on/find the solution roles. In multiple schools I observed first hand this theory where the female students would research facts and data to complement the experimentation done by the male students. I believe that more emphasis on the importance of not having the same 'job/career' each activity might help not only improve male responses for this question, but also female responses to other questions.

5.5 Impact of the 'You're Hired!' Program

While it is likely that outside influences also had an effect on the responses in Section 5.1 – Section 5.4, the information presented in this section can be directly attributed to the 'You're Hired!' program. Program impact was measured by asking, 'How much did participating in the activity impact each of the following?' on the post-survey with response options of: 'A Great Deal', 'Moderately', 'Slightly' and 'Not At All'. This question, unlike the other questions asked on the pre- and post-surveys, was directly focused on the 'You're Hired!' program. Figure 5.12 presents all Impact statements the students responded to. The y-axis lists the Impact statements along with the percentage of students that agreed, at least on some level, to the statement. The x-axis which is labeled 'Data' represents the percentage of agreement as a decimal number. In order to display the data from the Impact statements visually, the categorical data from the post-survey was summarized so that each response option had a percentage of students that had responded associated with it. The responses that showed a level of agreement ('A Great Deal', 'Moderately' and 'Slightly') were then added together to create the overall level of agreement shown in the figure. Descriptive statistics for project impact based on gender can be found in Table 5.23 and Table 5.24. In order to summarize the data for these tables, the data was converted using a Likert Scale in which 'A Great Deal' =4, 'Moderately' = 3, 'Slightly' = 2, and 'Not At All' =1. Once the categorical data was converted to numerical data using the Likert Scale, mean values as well as 95% confidence intervals could be determined.



Figure 5.12. Student levels of agreement to the statement, 'How much did participating in the activity impact each of the following?' The highest impacts were in 'Helped me understand problem solving better' at 88.8% and 'Increased my confidence in my ability to participate in engineering projects of activities' at 80.3%.

				95%	6 C.I.
Impact	Gender	Ν	Mean	Lower	Upper
Im1 - Helped me understand	Female	253	2.68	2.57	2.79
problem solving better	Male	308	2.57	2.47	2.67
Im2 - Led me to a better	Female	251	2.41	2.28	2.54
goals	Male	307	2.34	2.23	2.46
Im3 – Increased my interest in	Female	250	2.00	1.87	2.13
studying engineering in college	Male	308	2.41	2.30	2.53
Im4 - Increased my interest in	Female	253	1.93	1.82	2.05
college	Male	307	2.34	2.22	2.45
Im5 - Made me think more about	Female	253	2.49	2.35	2.62
high school	Male	308	2.44	2.32	2.56
Im6 - Made me decide to work	Female	253	2.55	2.41	2.68
harder in school	Male	305	2.46	2.34	2.58

Table 5.23. Descriptive Statistics of Responses to Impact Statements for Females and Males.

Table 5.23. cont.

				95% C.I.	
Impact	Gender	Ν	Mean	Lower	Upper
Im7 - Made me decide to take different classes in school (including college) that I had planned to	Female Male	251 306	2.14 2.07	2.01 1.95	2.27 2.19
Im8 – Made me confident in my	Female	252	2.18	2.06	2.30
ability to succeed in an engineering or technical field	Male	308	2.42	2.31	2.53
Im9 - Increased my confidence in my ability to participate in engineering projects or activities	Female Male	251 307	2.40 2.56	2.27 2.45	2.53 2.67

In general, the 'You're Hired!' program appears to have an impact with all means except one above a 2 (slightly). The only statement that resulted in a mean value less than a 2, or 'slightly' was the female mean for the statement, 'Increased my interest in studying a technical degree in college'. The significant responses when comparing female and male responses can be found in Table 5.24.

Table 5.24. Statistical Comparison of Responses to Impact Statements for Females and Males.

		Р-		Female	Male
Impact	Ν	Value	X ²	Mean	Mean
Im3 - Increased my interest in	558	< 01	22.66	2.00	2 41
studying engineering in college	558	\.01	22.00	2.00	2.41
Im4 - Increased my interest in					
studying a technical degree in	560	<.01	24.46	1.93	2.34
college					
Im8 - Made me confident in my					
ability to succeed in an engineering	560	0.03	9.06	2.18	2.42
or technical field					

For these three statements, Im3 – 'Increased my interest in studying engineering in college', Im4 – 'Increased my interest in studying a technical degree in college' and Im8 – Made me confident in my ability to succeed in an engineering or technical field' the male responses were significantly higher than the female responses.

When students were asked, 'Would you recommend that your friends participate in this activity?', overall 69.8% of students stated, 'Yes'. It is important to note that the amount of exposure to engineering and other STEM classes outside of the 'You're Hired!' activity may have impacted students as well. Table 5.25 displays the breakdown of how each school responded. There was a significant difference in how Schools responded when compared to each other, with School C and School E being the extremes.

Table 5.25.	Breakdown of	how School	A – School E	Responded	l to the Q	Juestion
'Would you	Recommend the	nat your Frien	nds Participate	in this Ac	tivity?'	

	N	Yes
School A	136	77%
School B	72	73%
School C	282	61%
School D	54	83%
School E	18	94%

CHAPTER VI

CONCLUSIONS & FUTURE WORK

6.1 Conclusions

'You're Hired!' appears to be a well-designed program that has increased students' understanding and attitudes of engineering based on data collected over two years of pre- and post-surveys. Overall the student responses showed a statistically significant, positive change in their understanding and attitudes towards engineering for the statement, E2 - 'Engineers mainly work with other people to solve problems'. This was true for the all-school experimental group data analysis as well as for both female and male responses. This initial data indicates that these positive changes in attitudes towards engineering can be most likely associated with the 'You're Hired!' program. This past year the program included a control group. The control group did not see a change in attitudes towards engineering, but the students participating in the program activities did.

Other categories of questions such as Interest and Attitudes and Skills cannot be as closely linked to the 'You're Hired!' program because the trends were not observant between experimental versus control groups like it was for the 'Engineer' category.

Pre-survey responses to the Interest statements for the experimental and control groups indicated that there was a statistical difference between the two groups of students. The experimental group responded at a greater level of agreement than the control group. When assessing the data for the Interest category by gender, female results stood out because of the increase in the level of agreement to the statement In14 - 'I like learning how things work'.

This increase in female interest to learn how things work complements the results found for the Attitude and Skills category. When looking at gender responses, males showed a statistical drop in their level of agreement to the statement, A5 - 'I know where I can find the information I need to solve problems.' While not statistically significant, the female responses to this same statement showed the opposite effect. In my observations from being present at the participating schools during the activities, I would suggest that these responses observed are due to the fact that females more times than not will take on a 'research' style of a role whereas the males choose the 'hands-on' engineer and technician roles. The individuals that conduct the research for the activities get very good at learning where to look for the information that is relevant. It is possible that in the process of researching new technologies and real-world problems, that female's interest is sparked in that they like to learn how the various 'things' work.

There was no change in the Attitude and Skills statements for the experimental versus control groups however both the all-school experimental group as well as both genders showed a statistical increase in their agreement to the statement A2 - 'I can use what I know to design and build something mechanical that works.' This also supports my conclusion that females as well as male students get engaged through the 'You're Hired!' hands-on program. It is possible that with a larger sample size for the experimental versus control groups from School A, that a similar result would have been observed.

This project achieved its goal to 1) develop a program that is designed to change students' attitudes and understanding of engineering, and 2) determine the effectiveness of the program based on data from participants. From my interpretation of this data analysis I would state that 'You're Hired!' is a well-designed program that resulted in a change in student attitudes and understanding of engineering. It is likely that other factors also contributed to student responses such as a class field trip, teachers incorporating more STEM into their daily classroom, student discussions with friends or family about possible careers or any other student experiences throughout the course of the school year.

The program developed also achieve the following goals: 1) introduce a variety of engineering careers possible with an engineering degree, 2) increase students' attitudes and understanding of engineering by incorporating features that have been previously researched and are proven to show impact, 3) introduce teachers to incorporating STEM and 21st Century Skills into their classrooms with minimal additions to their already busy workload, 4) hone students' 21st Century Skills also known as engineering mindset skills or workforce skills, and 5) engage industry professionals to participate and provide feedback as well as being a positive role model for students.

The program was successful at introducing engineering careers possible with an engineering degree that are offered at local North Dakota universities. The program also increased students attitudes and understanding that engineers work with others to collaborate and problem solve as well as the understanding that engineers can choose many different kinds of jobs. An engineering career could be anything from a process engineer working in corn milling to a new products engineer that works with inventors to design, problem solve and create new devices.

The program also showed success in introducing teachers to the idea of incorporating STEM and 21st Century Skills into their daily classrooms. Multiple teachers within the five schools have asked for additional suggestions to turn their current content into a more hands-on, collaborative project for their students. The 'You're Hired!' program as well as the teacher's new STEM/21st Century Skill classroom will help students to hone their engineering mindset skills also known as workforce skills.

Throughout the program students had multiple opportunities to receive feedback from industry professionals that attended the activities. One school in particular has also started incorporating key industry professionals into their classroom STEM activities because the impact on the student efforts is so large.

Ongoing research will continue to assist in understanding what changes can be made to make the program more effective. Future plans for the project include changing the pre- and post-surveys from the original AWE design to one that works better for the project, professional development for teachers to extend engineering design principles from 'You're Hired!' into the daily classroom, research to identify program elements that contribute to positive student changes, expansion of its implementation across a tri-state region, and development of a model for sustainability and scalability into other regions.

6.2 Future Work

Through collaboration between three higher education organizations in North Dakota, University of North Dakota, North Dakota State College of Science, and North Dakota State University it is projected that the 'You're Hired!' program will continue to play a role in schools across the immediate region and into further regions. For the future, the project will need to evolve and the 'You're Hired!' research team will need to continue to redesign to ensure it is meeting the needs of the key stakeholders as mentioned in Chapter 2. While there will be aspects of the project that are always in need of redesign, there are a couple features that should be addressed sooner rather than later. Those features include the pre- and post-surveys and teacher professional development. It will also be important to continue to revisit the program elements yearly to ensure key elements are in place as well as to ensure that the 'You're Hired!' program is designed in such a way that it can scale and evolve to meet the needs of the stakeholders.

Redesign of the Pre-Survey and Post-Survey

The pre- and post-surveys that have been used for 'You're Hired!' were adopted from AWE (Assessing Women & Men in Engineering). This resource provided a very good starting point for looking at assessing the program I was designing.

After the first year of data analysis, I looked at the survey and noticed that there were statements and wording choices that I questioned. In my effort to learn more about the assessment, I reached out to an individual who worked on the development of the preand post-surveys to gain a better understanding of why they chose to use the questions/statements and the phrases they did. In the conversation I learned that the surveys were created by a team of researchers and due to funding running out, additional research had not been done on the surveys.

In the summer of 2014, I presented an introduction of the 'You're Hired!' program with preliminary data at the 2014 Annual ASEE Conference in Indianapolis, IN. When at the conference I was able to listen to individuals and groups talk about research that while different than mine, was very similar in some of the structuring. One group presented data in which it became very clear to me that AWE had also been a survey resource for their research. When I asked about their thoughts on the survey tool provided by AWE, they stated that they have continued to do research on their own and that they use some of the questions from the original survey, but are working on creating a new survey that better fits the needs of their program. This is also what 'You're Hired!' will do for future research. It will be important to continue to stay in contact with other groups that are doing similar research to brainstorm and learn from each other.

Work was done to address current issues with the pre- and post-surveys the 'You're Hired!' program has been using after the initial proof of concept in 2012-2013. There were also changes that needed to be made to the individual activities themselves as stated in Chapter 4. To preserve the quality of research, it was decided that changes would be made to the activities for the 2013-2014 project year. The new surveys will be debuted in the 2015-2016 project year.

Currently the pre-survey is designed with categories of questions/statements in the order of that shown in Table 6.1. These categories and questions were adopted from the original AWE survey for assessing middle and high school students conducting engineering activities.

113

Table 6.1. Original Categories of Questions asked by the AWE Pre-Survey for Engineering Activities.

1) Demographics	
-----------------	--

-Name -Gender

-Ethnicity

-Education

-Name of School

-Classes currently enrolled in

-Are you currently enrolled in honors or advanced classes

-Have you been encouraged to enroll in honors or advanced classes

-Are you enrolled in a special engineering curriculum

-Do you plan to enroll in honors or advanced classes next year

-Has anyone talked to you about the importance of taking classes that will prepare you for college

-Has anyone talked to you about the importance of math in your future career

-What do you plan to do when you graduate from high school

-Have either of your parents attended college or a university after high school

2) What Do Engineers Do?

Read the following statements about what engineers might do and indicate your agreement or disagreement with each statement

-Mainly work on machines and computers

-Mainly work with other people to solve problems

-Can choose to do many different kinds of jobs

-Mainly work on things that have nothing to do with me

-I don't know what engineers do

-If you go to college, do you think you will pursue a career in an engineering-related field

-In your future, do you think you want to be an engineer

-Has anyone talked to you about becoming an engineer

-If yes, who

3) What Do Technicians Do?

Read the following statements about what technicians might do and indicate your agreement or disagreement with each statement

-Mainly work on machines and computers

-Mainly work with other people to solve problems

-Work on things that help the world

-Can choose to do many different kinds of jobs

-Mainly work on things that have nothing to do with me

-I don't know what technicians do

Table 6.1. cont.

3) What Do Technicians Do?

Read the following statements about what technicians might do and indicate your agreement or disagreement with each statement

-If you go to college, do you think you will pursue a career in a technician-related field

-In your future, do you think you want to be a technician

-Has anyone talked to you about becoming a technician

-If yes, who

4) Future Work

The following statements describe work or jobs you might do in the future. Tell us how important each of the items below is to you in your future work

-Work that makes me think

-Work that allows me to make lots of money

-Work that allows me to use math, computers, engineering, or science skills

-Work that allows me to tell other people what to do

-Work that allows me to help solve problems and create solutions

-Work that is fun to do

-Work that allows me to have time with family

-Work that allows me to help my cmomunity and/or society

-Work that makes people think highly of me

-Work that is satisfying to me

5) Interests

Here is a list of statements. Tell us what you think about them.

-I look forward to science class in school

-I look forward to math class in school

-I would rather solve a problem by doing an experiment than be told the answer

-More time should be spent on hands-on projects in science or technology activities at school

-I would like to (or already do) belong to a science or technology activities club -I get bored when I watch programs on channels like Discovery Channel, Animal Planet, Nova, Mythbusters, etc.

-I like to get science books or science experiments kits as presents

-I like learning how things work

-Science is too hard when it involves math

-Science is a difficult subject

-Doing experiments in science class is frustrating

-I feel comfortable with using a computer to make graphs and tables

-I am interested in learning more about how things work

-I like to learn to use new technology

Table 6.1. cont.

6) Math Problem
-If you encounter a math homework problem that you don't know how to solve,
what are you most likely to do
-For the above problem, check the box that best describes your ability to learn to
solve this example math problem
7) Attitudes and Skills
The table lists things you can do when you are working on school activities or
assignments.
-When I see a new math problem, I can use what I have learned to solve the problem
-I can use what I know to design and build something mechanical that works
-In lab activities, I can use what I have learned to design a solution
-I can effectively lead a team to design and build a hands-on project
-I know where I can find the information that I need to solve difficult problems
-I can explain math or science to my friends to help them understand
-I can get good grades in math
-I can get good grades in science

I propose as part of the redesign for the project that the survey categories be reorganized to the order as shown in Table 6.2. For everyone that has taken a survey, the level of interest in the process of taking the survey decreases with the number of questions. While the students are given an electronic version which in turn takes away the visual of the total number of questions, it is still a good approach to ask the questions in the order of importance of data. While this technique was not used by AWE, asking demographic questions at the end of a survey is a technique that is encouraged by many survey collectors. Even the survey tool Surveymonkey that I have used for the 'You're Hired!' program suggests this approach. This approach is done because it is possible that when a student completes demographic questions first, the questions such as gender, race, are you in honors classes, etc. can bias a student into applying typical stereotypes onto themselves and thinking, 'I am a girl that does not have honors classes offered at my school, there is no way that I will be smart enough to be an engineer' or 'I am a girl and girls are not engineers, therefore I don't want to be an engineer'.

Table 6.2. Redesign of the Categories of Questions asked by the AWE Pre-Survey for Engineering Activities.

1) What Do Engineers Do?
2) What Do Technicians Do?
4) Self – Efficacy
5) Demographics

It is important to note that the new order of categories of questions/statements

does not contain the same categories as listed in Table 6.1. That is because in addition to

the re-ordering of the categories, questions/statements were also modified in order to get

the most information out of the student responses in the least amount of questions. Table

6.3 contains the proposed questions/statements under the above categories shown in

Table 6.2.

Table 6.3. Redesigned Categories for New Pre-Survey with Revised as well as Additional Questions/Statements.

1) What Do Engineers Do?
Read the following statements about what engineers might do and indicate your
agreement or disagreement with each statement
Design and create things we use every day (from a pencil you use to write to the
road you took to get to school)
Mainly work with other people to solve problems
Use math and science principles to determine a solution to a problem
Can choose to do many different kinds of jobs
Engineers can work in dirty environments
I don't know what engineers do
In the future, do you think you want to be an engineer
Has anyone talked to you about becoming an engineer, if so who

Table 6.3. cont.

2) What Do Technicians Do?		
Read the following statements about what technicians might do and indicate your		
agreement or disagreement with each statement		
Mainly work with other people to solve problems		
Use hands-on approaches to find a solution to a problem		
Can choose to do many different kinds of jobs		
Read the following statements about what technicians might do and indicate your		
agreement or disagreement with each statement		
Technicians can work in dirty environments		
I don't know what technicians do		
In the future, do you think you want to be a technician		
Has anyone talked to you about becoming a technician, if so who		
3) Self-Efficacy		
Tell us a little about yourself		
I can lead a team to succeed in a project		
I am motivated to do my best in school		
I am able to collaborate with my classmates on class projects		
Creativity and Innovation		
I can use existing knowledge to come up with new ideas, products, or processes		
Communication and Collaboration		
I can interact with my peers in a variety of different settings		
I can work with a group to create original works or to solve problems		
I respect other's ideas even if they are different than my own		
I like to bounce ideas off of others when trying to solve a problem		
Critical Thinking, Problem Solving, and Decision Making		
When given a problem, I am able to create a plan to solve it		
I am able to collect and analyze data during an experiment to identify a solution		
If my first solution does not work, I am motivated to try again		
I can gather information on a subject and am able to organized and use only the		
appropriate information		
4) Demographics		
Name		
Gender		
Ethnicity		
Education		
Name of School		
Classes currently enrolled in		
Are you currently enrolled in honors or advanced classes		

Table 6.3. cont.

4) Demographics
Have you been encouraged to enroll in honors or advanced classes
Are you enrolled in a special engineering curriculum
Do you plan to enroll in honors or advanced classes next year
Has anyone talked to you about the importance of taking classes that will prepare you for college
Has anyone talked to you about the importance of math in your future career
What do you plan to do when you graduate from high school
Have either of your parents attended college or a university after high school

When looking to redesign the survey questions, I believe that two key items need

to be addressed; 1) The 'What do Engineers do?' question and 2) Replace the goals,

future work, attitudes questions with a hybrid that will be categorized as 'Self-Efficacy'.

Self-efficacy has to do with one's capabilities or their beliefs in their ability to reach their

goals. This new category will help researchers to understand with fewer questions the

overall sense of ability and drive that participating students have to actually reach their

goals.

The changes to the Engineer category had to do with the idea that the original survey provided statements that were either too general, or may have been difficult for the students to understand exactly what was being asked. The redesign includes slightly more detail in the statement options.

The incorporation of the Self-Efficacy category provides not only has the potential to provide very useful information regarding students' natural abilities and their abilities to work hard and reach a goal, but this section also will provide a link back to the self/peer assessments that the students take after each activity. This series of statements helps to unify the project and what the 'You're Hired!' program is about – Infusing a

change students' attitudes towards engineering and engineering careers by providing hands-on, real-world problems for the students to solve in a STEM environment while infusing 21st Century Skills that will benefit the students when they leave the classroom.

The changes to the post-survey will mirror that of the pre-survey with the inclusion of the Impact statements that help researchers to understand the 'You're Hired!' specific impact. Prior to the launch of the new surveys in for the 2015-2016 school year, researchers including myself will form a focus group of students to better understand student interpretations of the questions. I will ensure that the right questions are being asked in a way that students understand and can relate to.

Professional Development for Teachers

One of the greatest advantages that 'You're Hired! has in its program design is the continuous feedback loop. This loop is not just between the program/teachers and the students, but the program and the teachers/schools. 'You're Hired!' ensures that after each activity not only do the students receive feedback based on the activity, but the teachers and schools do as well. This constant feedback between the program and the schools creates an atmosphere that assures teachers that their opinions and suggestions matter. It is critical in designing a program like this that 1) all of the design features such as real-world problem, 3 voucher/question opportunities, presenting to community members not teachers, etc. are met and 2) that the stakeholders such as the teachers, administrators, community volunteers and students understand that they are appreciated and that their opinions matter.

After the 2012-2013 pilot year, students as well as teachers provided feedback that was then implemented into the 2013-2014 program. The early changes was

empowering for the schools and teachers as well as for the students and the program designers.

The 'You're Hired!' program received feedback from the schools after the 2012-2013 school year that a summer professional development would be beneficial for the teachers. The professional development would instruct the teachers how to incorporate the program's ideas and methodology into their classrooms between the three activities. After the 2013-2014 school year, the amount of feedback regarding professional development for teachers was greater. A couple of schools have innovative teachers that are using the outline of 'You're Hired!' in their everyday classrooms already, other teachers may need more guidance on how to conduct this type of a change.

The professional development could also provide an opportunity to train teachers on lessons/activities such as presentation styles, conceptualization of a project, collaboration activities, utilizing graphs with data, etc. that teachers could do in their classrooms to provide students with the important skills they will need for the future workplace. These lessons/activities also provide scaffolding that theoretically will make students more prepared for the next 'You're Hired!' activity. The 'You're Hired!' activities become an opportunity for teachers to assess students' progress in these areas as well as other areas in STEM and 21st Century Skills. Incorporating STEM into classrooms as well as how to infuse 21st Century Skills into what is already being taught are other topics that will be part of the professional development.

Over the summer of 2014, work was done by a team from the University of North Dakota to see how providing lessons/activities to students between 'You're Hired!' activities would affect their participation in the remaining activities. Students attended a 'You're Hired!' summer camp that was one week long. Students participated in 'You're Hired!' competition activities on Monday, Wednesday and Friday. On Tuesday and Thursday students attended a variety of sessions to provide more knowledge, scaffolding, on how to improve for the next 'You're Hired!' activity. The sessions contained information on topics such as collaboration, presentations styles, the engineering design process, how to effectively use data and graphs, and more. While the results of this summer camp have not been analyzed yet, I believe from the conversations I have had with the team from the University of North Dakota that the 'training' the students received on data graphs, teamwork, presentation styles, etc. was beneficial in preparing for the next activity challenge.

Providing teachers with professional development to support 'You're Hired!' and the incorporation of STEM and 21st Century Skills into the everyday classroom is something that teachers in the immediate area are looking for. It will be a priority for the research team from the three higher education organizations to apply for a funding source to support such an activity.

Revisiting Program Design Key Features

While the 'You're Hired!' program was designed to include key design features that contributed to various successful engineering education programs, the program is very different than any other program that is being researched right now that I am aware of. There is an abundance of research being done on summer camps, after school programs, and program insertion into specific curriculum such as physics or engineering classroom, but nothing quite like 'You're Hired!'. Because 'You're Hired!' has the ability to reach every single student at a participating school, it is important that while the program was designed with key features that encourage students to pursue engineering, that researchers ensure that additional design features are in place to support and encourage all students and their interests. The key design elements of a summer camp, after school program, and curriculum supplement are included in the program. It is now time to determine what additional design features are needed to provide all students with a successful 'You're Hired!' experience.

Preparing for Scaling, Sustaining and Evolution of the Program

The 'You're Hired!' program is spreading to new schools every year by word of mouth only, no additional solicitation has been needed. Some schools participate with one sub-section of one grade, whereas other schools participate with their entire school including grades 7-12. It is important the program continues to keep the key design features while allowing for flexibility to allow the schools to make the program their own. The key design features endorse the program by providing research data as backing, however not every feature within the original program design is required to gain similar results from students.

There were five schools included in the research of this master's study, but overall there were twenty schools that participated in the 2013-2014 school year. Of this twenty schools, 3-5 of the schools completed the activities completely on their own. I would send supplies and needed documents and the teachers at the school would complete the rest of the organizing. This type of relationship is what is desired for all participating schools. In order to meet the needs of all of the schools that are looking to take part, it is

very important that the program be designed so that schools feel comfortable conducting the activities without assistance.

It is also important in terms of sustainability that new, engaging, real-world problem statements are created to keep students engaged in the activities. These problem statements need to include a hands-on experiment/project that allows student to be innovative and to determine a solution. The hands-on kit needs to be easily shipped or contain supplies that can be purchased inexpensively from a local grocery store or gas station. As the number of schools increases, it becomes more expensive to run the program. It is important that costs are kept as low as possible while still providing meaningful experiences. The 'You're Hired!' research team is also working hard to obtain funding through grants to help support the cost of the project.

Finally, it is important to prepare for possible evolutions of the project. As mentioned prior, in the summer of 2014 there was a 'You're Hired!' summer camp. While the original design behind 'You're Hired!' included the activities being done with every student in a participating school, the summer camp's preliminary information has provided insight into the benefits of this type of program. In the week long summer camp, the students completed three separate 'You're Hired!' challenge activities. One activity was on Monday, another on Wednesday, and the final on Friday of that week. On Tuesday and Thursday the students arrived at camp and received the important scaffolding, feedback, and support needed to improve for the next day's challenge. The camp contained all other design factors as determined by research to be important. The results and feedback of the scaffolding provided between the challenges will benefit the research team when designing the professional development.

124

Final Thoughts

I believe that the 'You're Hired!' program has been designed with very important key features for effective engineering education. I also believe that including the key stakeholders such as administrators, teachers, students and community members have helped to make this a program that is needed by many schools in the immediate research area. Finally, I feel that my role in designing and analyzing this program has put me in a position to excel in a variety of situations. In any project that is designed it is important to understand 1) key elements to the project, 2) who are the stakeholders and what are their needs, 3) what are the outside constraints that may not be readily obvious, 4) understanding current funding sources and how to find additional funding, and 5) how to collaborate and bring groups of experts and professionals together to brainstorm possible solutions to the given problem. Whether an individual is working with designing a new medical product for hospitals, a new process line in a factory, or designing an educational tool that will challenge current educational approaches, these are the skills that will allow for success.

Research does not stop at with this thesis, I expect this work to continue to redesign and create an educational tool that schools not only in the North Dakota, area use, but states across the country will use in their classrooms.
APPENDICES

APPENDIX A SAMPLE OF STUDENT PRE- AND POST-SURVEYS

You're Hired! Pre-Survey

Assent Form

Project Title: You're Hired! – Infusing STEM and 21st Century Skills Investigator(s): This study is being done by: Kristin Brevik, Student at UND; Dr. Frank Bowman, UND Instructor, Dr. Brian Tande, UND Instructor, Dr. Kristi Jean, NDSCS Instructor; Dr. Bradley Bowen, NDSU Instructor

We are doing a research study; a research study is a special way to find out about something. We are trying to find out if your attitude towards engineering changes over the course of the school year. We will be coming to your school 3 times to do activities that will improve your 21st century skills. Examples of 21st century skills are: teamwork, time management, collaboration, and critical thinking. These 21st century skills are skills engineers use every day.

If you want to be in this study, we will ask you to do several things:

Before the first activity we will have you fill out a pre-survey. There are no right answers. We will then be doing 3 activities with your school throughout the school year. When we come you will spend the entire school day working with us. We will give you a problem in the morning and then you and your team of classmates will have the entire day to find a solution. At the end of the day before school gets done you will have the opportunity to present your solution to judges in a board room. When you are done presenting, you will be asked to fill out an anonymous survey which will give you the chance to rate how your teammates did. You will also be given time to look back at how you did throughout the day and comment on what you would like to get better at for next time. At the end of the school year we will have you fill out a post-survey just like the pre-survey.

We want to tell you about some things that may happen to you if you are in this study. This activity takes an entire school day which may seem like a long time, but you will not be asked to do anything different than what teachers could normally have you do.

Not everyone who is in this study will benefit. A benefit means that something good happens to you. We don't know if you will benefit. But we hope to learn something that will help other people someday.

When we are done with the study, we will write a report about what we found out. We will not use your name in the report.

You do not have to be in this study. It is up to you. If you want to be in the study, but change your mind later, you can stop being in the study.

If you do not want to be in this study, your school will still have you do the project along with the assessments and surveys, we just won't be able to talk about your accomplishments in the report.

1. By checking the box by 'Yes' you have agreed to participate in this study.



ou're Hired! Pre-Survey	
Personal Information	
*2. First Name	
* 3. Last Name	
*4. Gender	
Male	
O Female	
* 5. Ethnicity (You may check m	ore than one, as appropriate)
African/Black American	
American Indian/Alaskan Native	
Aslan & Pacific American	
Latina/Latino/Hispanic American	
White American	
Other (please specify)	

You're Hi	ired! Pre-S	Survey				
Educatio						
*6. Educ	ation (Check	the grade yo	u are in now)			
0 🖷	O 78	0 •••	O 24	O 🚥	O 100	O (20h
*7. Nam	e of your sch	ool:				
]			L)			
*8. From	the list below	w, check the	classes you a	are currently (taking in scho	ol this year:
Pre-Algel	bra			/oreign Language		
Algebre I			Ē	Remained Marths		
Algetare I			Ē	Geometry		
Cómpute	e Applicatione		10	History/Social Blud		
Córraute	r Science		10	Music		
Chailing a	er CAD (Compiden-Ala	ad Drawing)		Population		
Earth or 1	Physical Science			Pre-Delotrum		
Engineer	ling.			Technology Entron	lian	
English						
Other meth, wh	gineering, or adence	COURSES.				
		803 243 4 43				
*9. Are y	ou currently	enrolled in he	onors or adva	anced classes	s?	
0		0*		1	Not Avelable	
*10. Hay	e y ou b een e	ncouraged to	enroll in ho	nors or advan	ced classes	
0		0.	21		Not Available	
*11. Are	you enrolled	in a special e	ngineering c	uniculum?		
0		O*	na la proto con o de la desta do con El caj 17 a	j	🔿 Not. Aveilable	
*12. Do	you plan to er	nroll in honor	s or advance	d classes ner	t year?	
O 🐜 🕺	rtett (SPC) in the sets	0*			Not Avelable	
*13. Has	anyone talks	d to you abou	rt the import	ance of taking	classes that	will prepare
you for co	liege?					173 - 199
0			C	Ne		

ou're H	ired! Pre-S	Survey				
Iducatio	n					
*6. Educ	ation (Check	the grade yo	u are in now	19		
0=	O 788	0.	0=	() tim	() ttm	() 12th
*7. Nam	e of your sch	ool:				
			1			
*8. From	the list below	w, check the	asses you a	are currently t	taking in scho	ol this year:
Fre-Alge	brik			Foreign Canguage		
Agebre	8			Deteral Matt		
Agebre				Geometry		
Compute	er Applications		_	History/Soutal Blud		
Compute	e Science		-	Muso		
Chartering (In CAD (Computer Au	ad (Summid)	-	Projecte		
Entitie	inter activities			Technology Education	-	
English			23			
Other meth, er	giteering, or adjectea	CLUTHER .				
			1			
			10			
*9. Are)	you currently	enrolled in he	nors or adv	anced classes	5?	
O *=		ON			Not Available	
*10. Hay	ve vou been e	ncouraged to	enroll in ho	nors or advan	ced classes?	e.
0.		O.»		11110 FONT 1117	Not Available	
***	wan operation	in a curaial o	anii conine e	main and the other	.	
- 11. Are	you enrolled		ngineering c	HILLENING .	A	
0-		0.				
*12. Do	you plan to e	nroll in honor	s or advance	d classes ner	xt year?	
0*-		ON	8	1	Not Available	
*13. Has	s anyone talke	ed to you abou	it the import	ance of taking	classes that	will prepare
you for co	ollege?		1. D. H			
O *=			C) No.		

You're Hired! P	re-Survey	
*14. Has anyone	talked to you about the	importance of math in your future career?
0-		O M
*15. What do you	plan to do when you g	raduate from high school? (Check only one.)
O Do to a college of unit	ntally.	
	al (community college, business acto	ut, beauty extract, etc.)
Attend a factorical sch	sol then transfer to a college or unitvers	alty .
O Get a MGONA BE		
O Join the million		
O Dant Mow		
Other (please specify)		
16. Have either of	your parents attended o	college or a university after high school?
0	0 **	O Durif Stow

Engineers

*17. Read the following statements about what engineers might do and indicate your agreement or disagreement with each statement:

	Agree	Changree	Dan't Knine
Matrix work or machines and computers	0	0	0
Mainly work with other people to make problems	0	0	0
Work on Pringe that ivelp the world	0	0	0
Can shoose to do many different kinds of jobs	0	0	0
Mainly work on things that have midting to do with ris	0	0	0
l don't know what engineers do	0	0	0
Other (please wills in whole series)	•)		
	1	1	
	1	L	
*18. If you go to colle	ge, do you think you	will pursue a career in	an engineering-related
ield?		AND STANDARD CONTRACTOR STATES	AND IN TRACTOR AND AND A SECOND STREET
0	O **	0.	len'i thew

*19. In your future.	do you think you want to b	e an engineer?
O*=	0*	O Don't Know
* 20. Has anyone ta	lked to you about becomin	g an engineer?
0	() ne
21. If yes, put a chee	ck by everyone who has tal	ked to you about this:
People in after actual pro-	21/14	Outdettos pourgealor
Engineering or technology	r beauther	Math leacher
Family members	E	Ocianos teacher
Fundly Menda	Ī	Computer teacher

*22. Read the following statements about v	vhat technicians mig	ht do and indicate your
agreement or disagreement with each stater	nent:	
	and the second se	The set strength

	A.27 M	CONTRACTOR OF CONT	State States
Mainly do hands on work and experiments	0	0	0
Malety work with other people to aply problems	0	0	0
West on trings that hep the world	0	0	0
Can of some to do many offerent lights	0	0	0
Mainly work on things that have nothing to do with me	0	0	0
í dövit know whet lestrolatos de	0	0	0
Other (please with in whole sectorics	•)		
	8		
	5		
*23. If you further you	r education, do you ti	hink you will pursue a	a career in a technician
*23. If you further you related field?	r education, do you ti	hink you will pursue a	a career in a technician
*23. If you further you related field? 〇 ==	r education, do you tl O 4=	hink you will pursue a	a career in a technician
*23. If you further you related field? () *24. In your future, do	r education, do you th O 4= you think you want t	hink you will pursue a O ¤ o be a technician?	a career in a technician
*23. If you further you related field? > *24. In your future, do >	r education, do you th O % you think you want t	hink you will pursue a o be a technician?	a career in a technician unt know unt know
*23. If you further you related field? > *24. In your future, do > *25. Has anyone talks	you think you want t	o be a technician?	a career in a technician ont Now
*23. If you further you related field? ' *24. In your future, do ' *25. Has anyone talks '	r education, do you th	o be a technician?	a career in a technician ont Now ont Now
*23. If you further you related field? ' *24. In your future, do ' *25. Has anyone talke ' 26. If yes, put a check f	r education, do you th you think you want t n= ed to you about becom nc hy everyone who has	o be a technician? o be a technician? o ning a technician? o o talked to you about th	a career in a technician unt lines ont these ont these uis:
* 23. If you further you related field?	r education, do you th you think you want t ed to you about becon ue by everyone who has	nink you will pursue a be a technician? ning a technician? talked to you about th Guteros corrector	a career in a technician ont Now ont Now ont Now
*23. If you further you related field? ' *24. In your future, do ' *25. Has anyone talke ' 26. If yes, put a check f 	r education, do you th o he you think you want t o he ed to you about becom o he hy everyone who has here	o be a technician? o be a technician? o a technician? o a ning a technician? o a talked to you about th Outers consets Unit bacter	a career in a technician ont Now ont Now ont Now
 *23. If you further you related field? *= *24. In your future, do *= *25. Has anyone talks *= *6. If yes, put a check if Perple is after acted program Cryinearing or technology teat Tenty merclers 	you think you want t you think you want t ne ed to you about becom ne by everyone who has	nink you will pursue a o be a technician? o be a technician? o o ning a technician? o o talked to you about th outputs porestor talked to you about th outputs porestor	a career in a technician ont they ont they iiS:
*23. If you further you related field? '** *24. In your future, do '** *25. Has anyone talks '** 26. If yes, put a check f Proposition after acteal proposition Cogineering or technology teal frenty members Frenty members	r education, do you th you think you want t o n= ed to you about becom n= hy everyone who has n= to you about becom	hink you will pursue a o be a technician? o be a technician? o o ning a technician? o o talked to you about th o ductors corrector Units teacher o corputer teacher	a career in a technician on the on the iis:

Your Goals

*27. The following statements describe work or jobs you might do in the future. Tell us how important each of the items below is to you in your future work:

	Havi Important	Somewhat Important	Wery Interaction
Work but makes me bitk	0	0	0
Work that allows me to make lots of money	0	0	0
Work that allows the to use math, computers, engineering, or solence with	0	0	0
Work that allows the to tell other people what to do	Ø	0	0
Work that allows the to field active problems and create activities	0	0	0
Work that is full to do	0	0	0
Work that allows the to have time with family	O	0	0
Work that allows the to twic my community antition accerty	0	0	0
Work that makes people bons trighty of me	0	0	0
Work that is leaderlying to me	0	0	0

* 28. Here is a list of statements. Tell us what you think about them. Select a response that indicates your level of agreement.

	Strongly Disagree	Stimewhat Disagne	Somewhat Agree	Strongly Agree
I lists forward to activities classe (n suffect)	0	0	0	0
Laure forward to meth chase In echool	0	0	0	0
Leocald rether spike a problem by Spikg an experiment than be told the process	0	0	0	0
More the should be epert on hands on projects in science or technology activities at school	0	0	0	0
) would like to (or already dt) being to a where to technology activities data	0	0	0	0
I get bored when I webh programs on channels like Decovery Channel, Animal Planet, Nova, Wythosters, etc.	0	0	0	0
I like to get acteros books or acteros experimenta lite as presenta	0	0	0	0
I like learning how things with	0	0	0	0
Edence is the hard when it involves math	0	0	0	0
Science is a difficult subject	0	0	0	0
Doing separtments in science clean is functions	0	0	0	0
I level comfortable with using a computer to imake graphic and bables	0	0	0	0
I am interested in learning more about how things work	0	0	0	0
Little to learn to use new learnology	0	0	0	0

What You Think

* 29. If you encounter a math homework problem that you don't know how to solve, what are you most likely to do?

1.000	
	tal a parent in other family member for help with the problem
	Call a Intend who you know to good all math and say, her or him to help so you can allow it
	Context the Homework Hullins or simpler resource
	Gel help from your meth beacher on this problem
	Nork 1 but with my abuty group
	in in the feacher's each page for help
	Copy the arrawer from one of your Hande
	Search the Internet for Asia
	Take some time and try to figure out from to best approach activing this problem
Other	givene specify
This is	en exemple of a meth problem

(x)*3 - (2x)*2 + (3x) • f

What relies of a woold entry this producen? (if you are unsule about here this problem would look when writed by a teacter, please and for help.)

* 30. For the above problem, check the box below that best describes your ability to learn to solve this example math problem? Check only one.

Loan solve this problem maw
 Loan Seath myself to solve this problem
 Loan Seath myself to solve this problem
 I will be able to learn to active this problem once I take the right classese
 Even if I tools the right class, I sourcen't be able to learn to active this problem
 Toon't block I will ever take a class that fas problems this hard
 Tem not interested to learning to active this type of math problem

Other (please specify)

*31. The table lists things you can do when you are working on school activities or assignments. Check the appropriate box to tell us how often you do each of these things.

	Sterier.	Sometimes	Very Diffen	Abraya
Lon use what I have learned to least trywell have to program a computer game.	0	0	0	0
When I see a new mith problem, I can use what I have learned to solve the problem	0	0	0	0
I can use what I snow to design and build something mechanical that exits	0	0	0	0
In laty activities, I can use what I have learned to design a exturion	0	0	0	0
I can effectively lead a learn to design and build a hands-or project	0	0	0	0
How when I can find the information that I need to entry difficult problems	0	0	0	0
I can explain math of exercise to my thereas to help them understand	0	0	0	0
I can get good grades in with	0	0	0	0
I can gel good grades in science	0	0	0	0

Personal Information

I I I JL HUITE	*1	First I	ame
----------------	----	---------	-----

*2. Last Name

*3. Gender

Male

O Female

*4. Ethnicity (You may check more than one, as appropriate)

African/Black American
American Indian/Alaskan Native
Asian & Pacific American
Latina/Latino/Hispanic American
White American

Other (please specify)

ou're Hi	red! Post-S	Survey				
Education	n					
*5. Educ	ation (Check	the grade yo	u are in now)	ß		
O 6th	() 7th	O 8th	O 9th	O 10th	() 11th	() 12th
*6. Nam	e of your sch	ool:				
*7. Are y	ou currently	enrolled in ho	onors or adva	inced classe	s?	
() Yes			5		Not Available	
*8. Have	you been en	couraged to	enroll in hone	ors or advanc	ed classes?	
() Yes			2	8	Not Available	
*9. Are v	ou enrolled in	n a special en	gineering cu	rriculum?		
() Yes					Not Available	
*10. Do v	ou plan to er	nroll in honor	s or advance	d classes ne	xt year?	
() Yes			x]);	1	Not Available	
*11. Has	anyone talke	d to you abou	It the import	ance of takin	a classes that	will prepare
you for co	llege?					
() Yes			С) No		
*12. Has	anyone talke	ed to you abou	ut the import	ance of math	in your future	career?
⊖ Yes			C) No		
*13. What	at do vou plan	to do when	vou graduate	from high s	chool? (Chec	k only one.)
Go to a c	ollege or university		, j		under (enco	
Attend a t	lechnical school (con	munity college, busin	ess school, beauty s	chool, etc.)		
Attend a t	echnical school then	transfer to a college	or university	en constituin a francés (n. 2		
Get a full	-time job					
Join the r	military					
O Don't know	w					
Other (please s	pecify)					

14. Have either of your parents attended college or a university after high school?

() Yes

O No

5		111
U	Dont	Know

You're Hired! Post-Survey

Tell Us What You Think About You're Hired!

*15. Help us improve the activity by telling us what you think about the activity by checking the appropriate box:

	Strongly Disagree	Disagree	Agree	Strongly Agree
The information I received about the activity before it began helped me to participate successfully	0	0	0	0
I found the instructions and information I received during the activity to be effective and helpful	0	0	0	0
If I needed help during the activity, it was readily available	0	0	0	0
I found it easy to get to know the other participants in this activity	0	0	0	0
The leaders for this activity were prepared	0	0	0	0
This activity was well organized	0	0	0	0
My goals for participating in this activity were met	0	0	0	0

*16. How much did participating in the activity impact each of the following?

	Not at All	Slightly	Moderately	A Great Deal
Helped me understand problem solving better	0	0	0	0
Led me to a better understanding of my own career goals	0	0	0	0
Increased my interest in studying engineering in college	0	0	0	0
Increased my interest in studying a technical degree in college	0	0	0	0
Made me think more about what I will do after graduating from high school	0	0	0	0
Made me decide to work harder in school	0	0	0	0
Made me decide to take different classes in school (including college) that I had planned to	0	0	0	0
Made me confident in my ability to succeed in an engineering or technical field	0	0	0	0
Increased my confidence in my ability to participate in engineering projects or activities	0	0	0	0

*17. What did you like best about this activity? (Please write in whole sentences)

4

-

*18. If you were in charge, how would you change this activity? (Please write in whole sentences.)

*19. Would you recommend that your friends participate in this activity?

~	<u>.</u>	
	- 3	Maria
£	1.1	10
N	-	

O No

*20. In whole sentences, please explain why or why not:

Engineers

*21. Read the following statements about what engineers might do and indicate your agreement or disagreement with each statement:

	Agree	Disagree	Don't Know
Mainly work on machines and computers	0	0	0
Mainly work with other people to solve problems	0	0	0
Work on things that help the world	0	0	0
Can choose to do many different kinds of jobs	0	0	0
Mainly work on things that have nothing to do with me	0	0	0
I don't know what engineers do	0	0	0
Other (please write in whole sentence	es)		
		-	

*22. If you go to college, do you think you will pursue a career in an engineering-related field?

⊖ Yes	○ No	O Don't Know
*23. In your future	e, do you think you want	to be an engineer?
O Yes	○ No	O Don't Know
*24. Has anyone t	alked to you about beco	ming an engineer?
⊖ Yes		O №
25. If yes, put a che	eck by everyone who has	talked to you about this:
People in after school pr	ograms	Guidence counselor
Engineering or technolo	gy teacher	Math teacher
Family members		Science teacher
Family friends		Computer teacher
Other (provide kind of person	or teacher, not name)	

Technicians

*26. Read the following statements about what technicians might do and indicate your agreement or disagreement with each statement:

	Agree	Disagree	Don't Know
Mainly do hands on work and experiments	0	0	0
Mainly work with other people to solve problems	0	0	0
Work on things that help the world	0	0	0
Can choose to do many different kinds of jobs	0	0	0
Mainly work on things that have nothing to do with me	0	0	0
l don't know what technicians do	0	0	0
Other (please write in whole sentence	25)		
		2	

*27. If you further your education, do you think you will pursue a a career in a technicianrelated field?

○ Yes	◯ No	O Don't Know
*28. In your future	, do you think you want to	be a technician?
⊖ Yes	◯ No	O Don't Know
*29. Has anyone t	alked to you about becom	ing a technician?
⊖ Yes	◯ No	O Don't Know
30. If yes, put a che	ck by everyone who has t	alked to you about this:
People in after school pr	ograms	Guidence counselor
Engineering or technolo	gy teacher	Math teacher
Family members		Science teacher
Family friends		Computer teacher
Other (provide kind of person	or teacher, not name)	

Your Goals

*31. The following statements describe work or jobs you might do in the future. Tell us how important each of the items below is to you in your future work:

	Not Important	Somewhat Important	Very Important
Work that makes me think	0	0	0
Work that allows me to make lots of money	0	0	0
Work that allows me to use math, computers, engineering, or science skills	0	0	0
Work that allows me to tell other people what to do	0	0	0
Work that allows me to help solve problems and create solutions	0	0	0
Work that is fun to do	0	0	0
Work that allows me to have time with family	0	0	0
Work that allows me to help my cmomunity and/or society	0	0	0
Work that makes people think highly of me	0	0	0
Work that is satisfying to me	0	0	0

*32. Here is a list of statements. Tell us what you think about them. Select a response that indicates your level of agreement.

REALIZED AND A REAL PROVIDENCE	Strongly Disagree	Somewhat Disagree	Somewhat Agree	Strongly Agree
I look forward to science class in school	0	0	0	0
I look forward to math class in school	0	0	0	0
I would rather solve a problem by doing an experiment than be told the answer	0	0	0	0
More time should be spent on hands-on projects in science or technology activities at school	0	0	0	0
I would like to (or already do) belong to a science or technology activities club	0	0	0	0
I get bored when I watch programs on channels like Discovery Channel, Animal Planet, Nova, Mythbusters, etc.	0	0	0	0
I like to get science books or science experiments kits as presents	0	0	0	0
I like learning how things work	0	0	0	0
Science is too hard when it involves math	0	0	0	0
Science is a difficult subject	0	0	0	0
Doing experiments in science class is frustrating	0	0	0	0
I feel comfortable with using a computer to make graphs and tables	0	0	0	0
I am interested in learning more about how things work	0	0	0	0
I like to learn to use new technology	0	0	0	0

What You Think

*33. If you encounter a math homework problem that you don't know how to solve, what are you most likely to do?

	Ask a parent or other family member for help with the problem
	Call a friend who you know is good at math and ask her or him for help so you can solve it
	Contact the Homework Hotline or similar resource
	Get help from your math teacher on this problem
	Work it out with my study group
	Go to the teacher's web page for help
	Copy the answer from one of your friends
	Search the internet for help
	Take some time and try to figure out how to best approach solving this problem
Othe	r (please specify)
	2

This is an example of a math problem:

 $(x)^{n}3 - (2x)^{n}2 + (3x) = 6$

What values of x would solve this problem? (If you are unsure about how this problem would look when written by a teacher, please ask for help.)

*34. For the above problem, check the box below that best describes your ability to learn to solve this example math problem? Check only one.

I can solve this problem now

I can teach myself to solve this problem.

I will be able to learn to solve this problem once I take the right classes

O Even if I took the right class, I wouldn't be able to learn to solve this problem

O I don't think I will ever take a class that has problems this hard

I am not interested in learning to solve this type of math problem

Other (please specify)

*35. The table lists things you can do when you are working on school activities or assignments. Check the appropriate box to tell us how often you do each of these things.

	Never	Sometimes	Very Often	Always
When I see a new math problem, I can use what I have learned to solve the problem	0	0	0	0
I can use what I know to design and build something mechanical that works	0	0	0	0
In lab activities, I can use what I have learned to design a solution	0	0	0	0
I can effectively lead a team to design and build a hands-on project	0	0	0	0
I know where I can find the information that I need to solve difficult problems	0	0	0	0
I can explain math or science to my friends to help them understand	0	0	0	0
I can get good grades in math	0	0	0	0
l can get good grades in science	0	0	0	0

APPENDIX B END OF THE YEAR 'YOU'RE HIRED!' REPORT THAT SCHOOLS RECEIVED

'You're Hired!' Pilot Year Results

The pilot year was a success with more than 73.5% of the students recommending that their friends participate in this activity. Greatest project impact as assessed by the students was a better understanding of problem solving at 85.1%, ranging from 'slightly' to 'a great deal'. While the interest in pursuing engineering or technology-related careers did not statistically change during the course of the year, attitudes towards these careers did change. Recognizing the important role of faculty and staff for the positive results from this pilot project, 'You're Hired!' will continue for the 2013-2014 school year.

Sixteen participating schools were given the option of having students complete both a pre- and postsurvey to measure their awareness and perceptions about different aspects of engineering and possible STEM career choices. The survey was adapted from the Assessing Women and Men in Engineering (AWE) Project (HRD 0120642 and HRD 0607081). Evaluation highlights are included within this report, encompassing only assent-provided student responses from the seven schools that completed both the pre- and post-surveys.

The demographics breakdown from the seven schools (N=307 overall) included 52.8% (N=162) and 47.2% (N=145) of males and females, respectively. Grades and ethnicity follow:

Grade level:



A partnership between the North Dakota State College of Science and the University of North Dakota. Funded in part by the National Science Foundation NSF#0940143 and NSF#0942278 YOU'RE HIRED!

2012-2013 School Report Page 2 of 5

The project was a positive experience for a majority of the students, with 73.5% recommending that their friends participate in this activity. NDSCS considers this to be a very high approval rating due to the fact that schools can have 100% of their students participating, from high-risk to honor student.

Project impact was assessed by asking 'How much did participating in the activity impact each of the following?'



The greatest project impact was a better understanding of problem solving (85.1%). Their interest in studying a technical or engineering degree also increased. Both 'made me decide to work harder in school' and 'made me think more about what I will do after graduating from high school' received the highest 'A Great Deal' ranking.

Students' self-efficacy was also assessed by 'The table lists things you can do when you are working on school activities or assignments. Check the appropriate box to tell us how often you do each of these things:'

Attitudes and Skills	Always	Very Often	Sometimes	Never
When I see a new math problem, I can use	18.3% (55)	37.0% (111)	40.0% (120)	4.7% (14)
what I have learned to solve the problem	(-2.7%)	(-6.3%)	(7.4%)	(1.7%)
I can use what I know to design and build	14.3% (43)	27.3% (82)	45.0% (135)	13.3% (40)
something mechanical that works	(-0.2%)	(1.5%)	(4.9%)	(4.9%)
*In lab activities, I can use what I have	20.3% (61)	28.0% (84)	43.0% (129)	8.7% (26)
learned to design a solution	(2.1%)	(-12.0%)	(8.4%)	(1.5%)
I can effectively lead a team to design and	18.0% (54)	27.3% (82)	42.7% (128)	12.0% (36)
build a hands-on project	(0.0%)	(-4.0%)	(3.6%)	(0.5%)







A partnership between the North Dakota State College of Science and the University of North Dakota. Funded in part by the National Science Foundation NSF#0940143 and NSF#0942278



2012-2013 School Report Page 3 of 5

Attitudes and Skills	Always	Very Often	Sometimes	Never
I know where I can find the information	22.3% (67)	32.3% (97)	39.7% (119)	5.7% (17)
that I need to solve difficult problems	(0.4%)	(-3.3%)	(0.8%)	(2.1%)
I can explain math or science to my friends	20.4% (61)	37.1% (111)	34.4% (103)	8.0% (24)
to help them understand	(-1.6%)	(4.8%)	(-3.2%)	(0.0%)
	36.6% (109)	31.2% (93)	27.2% (81)	5.0% (15)
I can get good grades in math	(1.1%)	(-6.4%)	(2.3%)	(3.0%)
	37.9% (106)	35.7% (100)	25.0% (70)	1.4% (4)
I can get good grades in science	(2.5%)	(-5.8%)	(3.3%)	(-0.1%)

'You're Hired!' is designed to put the students in charge of their learning (self-directed learner) and encourage collaboration, instead of the student just raising their hand and asking a teacher for the answers. Students' behavior did change, with 11.8% more students recognizing that they can search the internet for help and a 9.0% decrease in the number that would go to their teacher for help. The change in students' behavior was assessed by 'If you encounter a math homework problem that you don't know how to solve, what are you most likely to do?'









A partnership between the North Dakota State College of Science and the University of North Dakota. Funded in part by the National Science Foundation NSF#0940143 and NSF#0942278

YOU'RE HIRED!

2012-2013 School Report Page 4 of 5

When asked 'If you go to college, do you think you will pursue a career in an engineering-related field?", post-survey results indicated:

'yes'	'no'	'don't know'		
27.2%	40.1%	32.8%		

There is not a statistical difference between the overall pre- and post-survey responses to this question, however there is an association of gender and school. The 'yes' response probability reaches 20.8% for females from three of the schools, while only 10.1% for females from the remaining schools. The 'yes' response probability is also higher for males from two of the schools (63.8%) than the remaining five schools (27.3%).

Student attitude towards engineering was assessed with 'Read the following statements about what engineers might do and indicate your level of agreement or disagreement with each statement – Engineering is: '



Students' attitude towards engineering careers changed over the course of this project, with the biggest change in students agreeing that engineers work with other people to solve problems (increase of 20.8%). Students also agreed that engineers can choose to do many different jobs (81.0%) and work on things that help the world (84.4%).







A partnership between the North Dakota State College of Science and the University of North Dakota. Funded in part by the National Science Foundation NSF#0940143 and NSF#0942278



2012-2013 School Report Page 5 of 5

When asked 'If you go to college, do you think you will pursue a career in a technician-related field?', post-survey results indicated:

'yes'	'no'	'don't know'		
17.6%	43.5%	38.9%		

There is not a statistcal difference between the overall pre- and post-survey responses to this question.

Student attitude towards technicians was assessed with 'Read the following statements about what technicians might do and indicate your level of agreement or disagreement with each statement:'



Students' attitude towards technician careers also changed over the course of this project, with the biggest change in students agreeing that technicians can choose to do many different kinds of jobs (increase of 13.8%).

With the positive student feedback and promising changes in students' attitude, behavior and selfefficacy, NDSCS will be working with the University of North Dakota and North Dakota State University to move the 'You're HiredI' pilot project forward.

Kristin Brevik UND Chemical Engineering Graduate Student Kristin.Brevik@ndscs.edu Dr. Kristi Jean NDSCS Associate Professor Kristi.Jean@ndscs.edu







A partnership between the North Dakota State College of Science and the University of North Dakota. Funded in part by the National Science Foundation NSF#0940143 and NSF#0942278

APPENDIX C ANALYSIS OF EVERY QUESTION/STATEMENT INCLUDED IN THE PRE-AND POST-SURVEY FOR THE EXPERIMENTAL AND CONTROL GROUPS

Engineer Questions

Engineers	Group	N	P-Value	X ²	Pre-Survey Mean	Post-Survey Mean
E1 - Mainly work on machines and computers	C E	134 135	0.74 0.30	0.60 2.42	2.20 2.54	2.29 2.42
E2 - Mainly work with other people to solve problems	C *E	132 133	0.30 <0.01	2.40 17.52	2.31 2.36	2.39 2.74
E3 - Work on things that help the world	C E	133 133	0.58 0.21	1.09 3.13	2.51 2.73	2.59 2.81
E4 - Can choose to do many different kinds of jobs	*C *E	130 134	0.04 <0.01	6.21 13.18	2.37 2.43	2.57 2.74
E5 - Mainly work on things that have nothing to do with me	*C E	130 133	0.01 0.12	9.87 4.20	1.75 1.70	1.60 1.63
E6 - I don't know what engineers do	C E	126 132	0.40 0.62	1.86 0.95	1.57 1.40	1.46 1.32
E7 - Pursue a career in an engineering-related field?	C E	136 136	0.32 0.20	2.27 3.23	1.66 1.95	1.74 2.06
E8 - Do you think you want to be an engineer?	C E	136 136	0.16 0.24	3.65 2.83	1.46 1.81	1.61 1.90

*Significance: p-value less than or equal to 0.05 C: 'Control Group' and E: 'Experimental Group' Degrees of Freedom (DF) =6

Future Work Questions

					Pre-Survey	Post-Survey
Future Work	Group	Ν	P-Value	X ²	Mean	Mean
F1 - Work that makes	С	136	0.51	1.35	2.46	2.47
me think	E	133	0.46	1.56	2.49	2.56
F2 - Work that allows	C	124	0.97	0.20	2.50	2.56
me to make lots of	C F	134	0.87	0.29	2.39	2.50
money	L	155	0.20	5.20	2.39	2.31
F3 - Work that allows						
me to use math,	С	136	0.90	0.22	2.19	2.15
computers, engineering,	E	133	0.87	0.28	2.32	2.37
or science skills						
F4 - Work that allows	C	13/	0.60	0.73	1 78	1.80
me to tell other people	E E	134	0.09	0.73 2.71	1.78	1.80
what to do	L	155	0.20	2.71	1.00	1.00
F5 - Work that allows						
me to help solve	С	135	0.21	3.12	2.16	2.31
problems and create	Е	133	0.52	1.29	2.43	2.48
solutions						
F6 - Work that is fun to	С	135	0.84	0.34	2.73	2.76
do	Е	135	0.23	2.95	2.79	2.85
F7 - Work that allows	C	125	0.14	2.05	2 72	2 77
me to have time with	C F	135	0.14	2.95	2.73	2.77
family	Ľ	155	0.35	2.09	2.74	2.05
F8 - Work that allows						
me to help my	С	135	0.40	1.81	2.49	2.40
community and/or	E	134	0.93	0.15	2.51	2.49
society						
F9 - Work that makes	С	135	0.43	1.68	2.26	2.23
people think highly of	Ē	134	0.34	2.16	2.31	2.28
me	_					
F10 - Work that is	С	135	0.23	2.95	2.84	2.76
satisfying to me	Ē	133	0.17	3.59	2.77	2.83
<i>J8</i>	—					

*Significance: p-value less than or equal to 0.05 C: 'Control Group' and E: 'Experimental Group' Degrees of Freedom (DF) = 6

Interest Questions

					Pre-Survey	Post-Survey
Interest	Group	Ν	P-Value	X^2	Mean	Mean
In1 - I look forward to science	С	136	0.46	2.56	2.90	2.78
class in school	Е	134	0.54	2.16	3.41	3.43
In2 - I look forward to math	С	134	0.51	2.30	2.97	2.82
class in school	Е	135	0.56	2.08	2.92	2.86
In3 - I would rather solve a						
problem by doing an	С	136	0.66	1.62	3.13	3.05
experiment than be told the	Е	135	0.12	5.83	3.36	3.25
answer						
In4 - More time should be						
spent on hands-on projects in	С	136	0.84	0.83	3.29	3.23
science or technology	E	135	а	а	3.63	3.64
activities at school						
In5 - I would like to (or						
already do) belong to a	С	134	0.69	1.45	2.02	1.90
science or technology	E	133	0.82	0.93	2.71	2.64
activities club						
In6 - I get bored when I						
watch programs on channels	С	136	0.44	2 72	2.00	2.09
like Discovery Channel,	F	130	0.44	3.86	1.99	1.03
Animal Planet, Nova,	Ľ	155	0.20	5.00	1.99	1.95
Mythbusters, etc.						
In7 - I like to get science	С	136	0.33	3 40	2 21	2.04
books or science experiments	F	130	0.90	0.58	2.21	2.04
kits as presents	Ľ	131	0.90	0.50	2.31	2.12
In8 - I like learning how	С	135	0.22	4.39	3.26	3.07
things work	Е	132	0.32	3.54	3.32	3.39
In9 - Science is too hard when	С	133	0.88	0.69	2.21	2.17
it involves math	*E	135	0.01	11.52	1.98	2.33
In10 - Science is a difficult	С	133	0.44	2.73	2.20	2.11
subject	Е	134	0.13	5.59	1.78	2.01
In11 - Doing experiments in	С	133	0.70	1 43	1 84	1.88
science class is frustrating	*F	135	0.70	7 74	1.04	1.80
	Ľ	155	0.05	7.74	1.57	1.07
In12 - I feel comfortable with	С	135	0.20	4.66	3.02	2.82
using a computer to make	Е	134	0.97	0.25	3.03	3.06
graphs and tables						
In13 - I am interested in	С	134	0.95	0.37	3.10	3.05
learning more about how	Ĕ	133	0.28	3.86	3.35	3.26
things work		100		5.00	5.55	5.20
In14 - I like to learn to use	С	135	0.15	5.27	3.30	3.20
new technology	*Ē	135	< 0.01	14.17	3.59	3.44
Bj	-	100	0.01	1	5.67	5

*Significance: p-value less than or equal to 0.05

C: 'Control Group' and E: 'Experimental Group'

DF = 9

 a - A reliable chi² and p-value could not be calculated for In4 – 'More time should be spent on hands-on projects in science or technology activities at school' because one of the answer options received less than five responses.

Attitude and Skills Questions

Attitude and Skills	Group	Ν	P-Value	X ²	Pre-Survey Mean	Post-Survey Mean
A1 - When I see a new math problem, I can use what I have learned to solve the problem	*C E	133 134	<0.01 0.14	13.32 5.52	2.88 2.92	2.71 2.78
A2 - I can use what I know to design and build something mechanical that works	C *E	132 134	0.16 0.02	5.23 9.59	2.30 2.48	2.55 2.55
A3 - In lab activities, I can use what I have learned to design a solution	C E	133 134	0.58 0.44	1.98 2.68	2.47 2.89	2.57 2.81
A4 - I can effectively lead a team to design and build a hands-on project	C E	133 133	0.45 0.15	2.66 5.35	2.39 2.85	2.53 2.85
A5 - I know where I can find the information that I need to solve difficult problems	C E	132 134	0.94 0.26	0.38 3.97	2.81 2.89	2.83 2.72
A6 - I can explain math or science to my friends to help them understand	C E	132 134	0.97 0.19	0.27 4.77	2.56 2.82	2.55 2.67
A7 - I can get good grades in math	C E	133 134	0.79 0.66	1.03 1.59	3.01 3.14	2.91 3.01
A8 - I can get good grades in science	C E	132 132	0.71 0.73	1.36 1.30	2.92 3.35	2.89 3.25

*Significance: p-value less than or equal to 0.05 C: 'Control Group' and E: 'Experimental Group' DF = 9

APPENDIX D ANALYSIS OF THE DIFFERENCES BETWEEN PRE-SURVEY RESPONSES FOR EACH OF THE FIVE PARTICIPATING SCHOOLS.

			Pre-Survey Mean				
Engineers	P-Value	X ²	Α	В	С	D	Ε
*E1 - Mainly work on machines and computers	< 0.01	25.48	2.54	2.46	2.34	2.43	1.89
*E2 - Mainly work with other people to solve problems	<0.01	35.49	2.36	2.55	2.20	2.74	2.89
E3 - Work on things that help the world	0.10	13.25	2.73	2.82	2.65	2.83	3.00
E4 - Can choose to do many different kinds of jobs	0.20	11.11	2.43	2.55	2.44	2.68	2.78
E5 - Mainly work on things that have nothing to do with me	0.09	13.65	1.70	1.58	1.57	1.66	1.22
*E6 - I don't know what engineers do	< 0.01	62.17	1.40	1.23	1.75	1.08	1.11
*E7 - Pursue a career in an engineering-related field?	< 0.01	25.72	1.95	1.85	1.77	2.19	2.00
*E8 - Do you think you want to be an engineer?	<0.01	22.17	1.81	1.72	1.64	2.07	1.83

Engineer Questions

Future Work Questions

			Pre-Survey Mean				
Future Work	P-Value	X ²	Α	В	С	D	Ε
F1 - Work that makes me think	< 0.01	33.00	2.49	2.57	2.31	2.65	2.72
F2 - Work that allows me to make lots of money	0.01	19.21	2.59	2.37	2.58	2.44	2.33
F3 - Work that allows me to use math, computers, engineering, or science skills	<0.01	36.91	2.32	2.38	2.11	2.65	2.50
F4 - Work that allows me to tell other people what to do	0.44	7.92	1.80	1.74	1.69	1.83	1.83
F5 - Work that allows me to help solve problems and create solutions	<0.01	41.02	2.43	2.53	2.20	2.63	2.83
F6 - Work that is fun to do	0.02	18.45	2.79	2.84	2.69	2.76	3.00
F7 - Work that allows me to have time with family	0.09	13.57	2.74	2.79	2.74	2.65	3.00
F8 - Work that allows me to help my community and/or society	< 0.01	31.91	2.51	2.53	2.34	2.79	2.78
F9 - Work that makes people think highly of me	0.07	14.45	2.31	2.10	2.29	2.25	2.72
F10 - Work that is satisfying to me	0.29	9.59	2.77	2.78	2.76	2.81	2.94

Interest Questions

				Pre-Survey Mean				
Interest	P-Value	X ²	Α	B	С	D	Ε	
*In1 - I look forward to science class in school	< 0.01	45.52	3.41	3.19	3.01	3.63	3.50	
In2 - I look forward to math class in school	0.06	20.46	2.92	2.84	2.73	3.17	3.39	
*In3 - I would rather solve a problem by doing an experiment than be told the answer	<0.01	66.08	3.36	3.27	2.87	3.46	3.83	
*In4 - More time should be spent on hands-on projects in science or technology activities at school	<0.01	45.03	3.63	3.46	3.23	3.69	3.89	
*In5 - I would like to (or already do) belong to a science or technology activities club	<0.01	61.84	2.71	2.47	2.19	3.17	3.06	
*In6 - I get bored when I watch programs on channels like Discovery Channel, Animal Planet, Nova, Mythbusters, etc.	0.01	26.39	1.99	1.77	2.00	2.26	1.50	
*In7 - I like to get science books or science experiments kits as presents	<0.01	38.76	2.51	2.23	2.05	2.75	2.00	
*In8 - I like learning how things work	< 0.01	35.42	3.32	3.42	3.11	3.50	3.89	
*In9 - Science is too hard when it involves math	< 0.01	29.56	1.98	2.07	2.29	2.17	1.78	
In10 - Science is a difficult subject	0.15	16.93	1.78	1.96	2.10	2.06	1.72	
*In11 - Doing experiments in science class is frustrating	0.01	26.41	1.59	1.66	1.89	1.92	1.44	
*In12 - I feel comfortable with using a computer to make graphs and tables	<0.01	30.16	3.03	2.92	2.81	3.36	3.39	
*In13 - I am interested in learning more about how things work	<0.01	82.80	3.35	3.28	2.80	3.60	3.89	
*In14 - I like to learn to use new technology	<0.01	88.46	3.59	3.48	2.92	3.75	3.72	

Attitude and Skills

			Pre-Survey Mean				
Attitude and Skills	P-Value	X ²	Α	В	С	D	Ε
*A1 - When I see a new math problem, I can use what I have learned to solve the problem	0.03	22.26	2.92	2.85	2.61	2.92	3.06
*A2 - I can use what I know to design and build something mechanical that works	<0.01	57.65	2.48	2.24	2.08	2.81	3.06
*A3 - In lab activities, I can use what I have learned to design a solution	<0.01	49.59	2.89	2.81	2.49	3.09	3.17
*A4 - I can effectively lead a team to design and build a hands-on project	<0.01	55.52	2.85	2.72	2.33	3.04	3.06
*A5 - I know where I can find the information that I need to solve difficult problems	0.01	25.46	2.89	2.92	2.60	2.94	3.22
A6 - I can explain math or science to my friends to help them understand	0.16	16.81	2.82	2.56	2.57	2.79	2.94
A7 - I can get good grades in math	0.05	20.91	3.14	3.07	2.91	3.15	3.39
*A8 - I can get good grades in science	0.01	26.36	3.35	3.19	3.00	3.23	3.44

APPENDIX E

ANALYSIS OF EVERY QUESTION/STATEMENT INCLUDED IN THE PRE-AND POST-SURVEY FOR THE ALL-SCHOOL EXPERIMENTAL GROUP Engineer Questions

Engineers	N	P-Value	X ²	Pre-Survey Mean	Post-Survey Mean
*E1 - Mainly work on machines and computers	558	0.01	9.93	2.40	2.23
*E2 - Mainly work with other people to solve problems	556	< 0.01	36.22	2.36	2.62
E3 - Work on things that help the world	554	0.33	2.24	2.72	2.75
*E4 - Can choose to do many different kinds of jobs	556	< 0.01	25.11	2.49	2.69
*E5 - Mainly work on things that have nothing to do with me	553	<0.01	11.01	1.60	1.66
E6 - I don't know what engineers do	551	0.61	0.98	1.51	1.47
E7 - Pursue a career in an engineering-related field?	559	0.73	0.62	1.87	1.90
E8 - Do you think you want to be an engineer?	559	0.77	0.53	1.74	1.77

*Significance: p-value less than or equal to 0.05

DF = 2
Future Work Questions

Future Work	Ν	P-Value	X ²	Pre-Survey Mean	Post-Survey Mean
F1 - Work that makes me think	553	0.78	0.50	2.43	2.46
F2 - Work that allows me to make lots of money	553	0.21	3.15	2.53	2.47
F3 - Work that allows me to use math, computers, engineering, or science skills	551	0.71	0.68	2.26	2.25
*F4 - Work that allows me to tell other people what to do	552	0.05	5.92	1.74	1.83
F5 - Work that allows me to help solve problems and create solutions	553	0.97	0.05	2.36	2.37
F6 - Work that is fun to do	553	0.38	1.93	2.75	2.78
F7 - Work that allows me to have time with family	554	0.77	0.52	2.74	2.72
F8- Work that allows me to help my community and/or society	551	0.29	2.45	2.47	2.42
F9 - Work that makes people think highly of me	552	0.78	0.49	2.28	2.24
F10 - Work that is satisfying to me	549	0.28	2.51	2.78	2.75

Interest Questions

Interest	Ν	P-Value	X ²	Pre-Survey Mean	Post-Survey Mean
In1 - I look forward to science	551	0.20	267	2 21	2 1 1
class in school	554	0.30	5.07	5.21	5.11
In2 - I look forward to math	550	0.76	1 16	2.86	2 97
class in school	330	0.76	1.10	2.80	2.87
In3 - I would rather solve a					
problem by doing an	550	0.54	2.15	2.12	2 1 1
experiment than be told the	330	0.34	2.13	5.15	5.11
answer					
In4 - More time should be					
spent on hands-on projects in	516	0.11	5.00	2 42	2 40
science or technology	540	0.11	5.99	5.42	5.40
activities at school					
In5 - I would like to (or					
already do) belong to a	552	0.26	4.00	2 47	2 42
science or technology	555	0.20	4.00	2.47	2.45
activities club					
In6 - I get bored when I					
watch programs on channels					
like Discovery Channel,	551	0.34	3.37	1.97	2.07
Animal Planet, Nova,					
Mythbusters, etc.					
In7 - I like to get science					
books or science experiments	550	0.91	0.55	2.25	2.24
kits as presents					
In8 - I like learning how	550	0.06	7 30	3.26	3 27
things work	550	0.00	1.59	5.20	5.27
In9 - Science is too hard when	550	0.11	6.11	2 15	2 27
it involves math	550	0.11	0.11	2.15	2.27
*In10 - Science is a difficult	550	0.04	8 46	1 98	2 13
subject	550	0.04	0.40	1.90	2.15
In11 - Doing experiments in	549	0.10	6 25	1 77	1 91
science class is frustrating	547	0.10	0.25	1.//	1.91
In12 - I feel comfortable with					
using a computer to make	550	0.97	0.22	2.95	2.97
graphs and tables					
In13 - I am interested in					
learning more about how	552	0.44	2.69	3.11	3.14
things work					
*In14 - I like to learn to use	549	0.03	9.26	3 27	3 30
new technology	549	0.05	1.20	5.21	5.50

Attitude and Skills

				Pre-Survey	Post-Survey
Attitude and Skills	Ν	P-Value	X ²	Mean	Mean
A1 - When I see a new math problem, I can use what I have learned to solve the problem	545	0.53	2.21	2.76	2.69
*A2 - I can use what I know to design and build something mechanical that works	544	<0.01	17.24	2.30	2.43
A3 - In lab activities, I can use what I have learned to design a solution	544	0.44	2.69	2.71	2.67
A4 - I can effectively lead a team to design and build a hands-on project	544	0.46	2.58	2.60	2.61
A5 - I know where I can find the information that I need to solve difficult problems	545	0.27	3.92	2.77	2.69
A6 - I can explain math or science to my friends to help them understand	544	0.71	1.37	2.66	2.65
A7 - I can get good grades in math	545	0.22	4.43	3.03	3.03
A8 - I can get good grades in science	542	0.59	1.90	3.15	3.15

APPENDIX F ANALYSIS OF EVERY QUESTION/STATEMENT INCLUDED IN THE PRE-AND POST-SURVEY BROKEN DOWN BY SCHOOL

Engineer Questions

					Pre-Survey	Post-Survey
Engineers	School	Ν	P-Value	X ²	Mean	Mean
	А	136	0.30	2.42	2.54	2.42
E1 - Mainly work on	В	68	0.56	1.15	2.46	2.31
machines and	*C	282	0.04	6.33	2.34	2.15
computers	D	54	0.13	4.05	2.43	2.07
•	Е	18	0.18	1.80	1.89	2.33
	*A	136	< 0.01	17.52	2.36	2.74
E2 - Mainly work	В	67	0.13	4.15	2.55	2.78
with other people to	*C	281	< 0.01	18.15	2.20	2.46
solve problems	D	54	0.13	4.13	2.74	2.81
	Е	18	0.31	1.03	2.89	3.00
	А	133	0.21	3.13	2.73	2.81
F3 - Work on things	В	68	0.57	1.11	2.82	2.90
that help the world	С	281	0.90	0.20	2.65	2.65
that help the world	D	54	0.18	3.38	2.83	2.83
	E	18	0.31	1.03	3.00	2.94
	*A	136	< 0.01	13.18	2.43	2.74
E4 - Can choose to do	*B	68	0.02	8.26	2.55	2.85
many different kinds	*C	282	0.04	6.59	2.44	2.58
ofjobs	D	52	0.10	4.60	2.68	2.88
	Е	18	0.55	0.36	2.78	2.89
	А	136	0.12	4.20	1.70	1.63
E5 - Mainly work on	В	67	0.72	0.66	1.58	1.48
things that have	*C	278	0.01	9.61	1.57	1.73
nothing to do with me	D	54	0.13	4.08	1.66	1.67
	Е	18	0.37	0.80	1.22	1.44
	А	134	0.62	0.95	1.40	1.32
F6 - I don't know	В	67	0.46	1.56	1.23	1.28
what engineers do	С	278	0.11	4.44	1.75	1.60
what engineers do	*D	54	0.04	6.53	1.08	1.33
	E	18	0.19	3.33	1.11	1.50
	А	136	0.20	3.23	1.95	2.06
E7 - Pursue a career in	В	69	0.58	1.08	1.85	1.97
an engineering-related	С	282	0.94	0.13	1.77	1.77
field?	D	54	0.93	0.15	2.19	2.13
	Е	18	0.53	1.28	2.00	1.78
	А	136	0.24	2.83	1.81	1.90
E8 - Do you think you	В	69	0.20	3.21	1.72	1.83
want to be an	С	282	0.94	0.12	1.64	1.62
engineer?	D	54	0.81	0.43	2.07	2.15
	E	18	0.75	0.58	1.83	1.72

Future Work Questions

					Pre-Survey	Post-Survey
Future Work	School	Ν	P-Value	X^2	Mean	Mean
	А	135	0.46	1.56	2.49	2.56
	В	66	0.16	3.69	2.57	2.67
F1 - Work that makes	Ē	281	0.45	1.60	2 31	2 30
me think	D	54	0.15	2.13	2.51	2.50
	E	17	0.91	0.01	2.03	2.09
	L	125	0.01	2.20	2.72	2.71
F2 - Work that allows	A R	65	0.20	5.20 0.22	2.39	2.31
ma to make lots of	Б С	203	0.89	0.22	2.57	2.42
me to make lots of		202	0.10	4.03	2.38	2.40
money	D F	17	0.38	1.07	2.44	2.41
	Е	1/	0.47	1.50	2.33	2.55
F3 - Work that allows	A	134	0.87	0.28	2.32	2.37
me to use math,	В	65	0.05	5.89	2.38	2.29
computers,	С	281	0.97	0.05	2.11	2.11
engineering, or	D	54	0.72	0.66	2.65	2.63
science skills	Е	17	0.80	0.43	2.50	2.41
	٨	135	0.26	2 71	1.80	1.80
E4 - Work that allows	R	65	0.20	0.59	1.00	1.80
ma to tall other poopla	в *С	281	0.74	6.39	1.74	1.80
me to ten omer people	·C	201	0.04	0.24	1.09	1.05
what to do	D	54	0.85	0.30	1.83	1.81
	E	1 /	0.58	1.07	1.83	1.94
F5 - Work that allows	Α	135	0.52	1.29	2.43	2.48
me to help solve	В	65	0.86	0.29	2.53	2.51
problems and create	С	282	0.75	0.57	2.20	2.17
solutions	D	54	0.12	4.20	2.63	2.78
solutions	Е	17	0.37	1.97	2.83	2.82
	А	135	0.23	2.95	2.79	2.85
	В	66	0.09	4.86	2.84	2.83
FO - WORK that is iun	С	281	0.92	0.17	2.69	2.70
10 00	D	54	0.80	0.44	2.76	2.81
	Е	17		0.00	3.00	3.00
	А	135	0.35	2.09	2.74	2.83
F7 - Work that allows	В	66	0.53	1.27	2.79	2.70
me to have time with	Ē	282	0.19	3.35	2.74	2.66
family	D	54	0.50	1.37	2.65	2.76
	Ē	17	0.30	1.09	3.00	2.94
	А	134	0.93	0.15	2.51	2.49
F8- Work that allows	B	65	0.14	3.89	2 53	2 35
me to help my	Č	281	0.78	0.50	2.33	2.33
community and/or	D	54	0.70	1.80	2.51	2.55
society	E	17	0.41	2.81	2.75	2.07
	A	124	0.24	2.01	2.76	2.02
EQ. Work that makes	A	154	0.34	2.10	2.31	2.28
F9 - WORK that makes	D C	200	0.47	1.32	2.10	2.08
people think highly of	C	282	0.64	0.88	2.29	2.25
me	D E	54 17	0.96	0.09	2.25	2.24
	E	1 /	0.28	2.31	2.12	2.39
	Α	133	0.17	3.59	2.77	2.83
F10 - Work that is	В	65	0.11	4.46	2.78	2.83
satisfying to me	С	280	0.13	4.09	2.76	2.67
substyling to file	D	54	0.58	1.08	2.81	2.76
	E	17	0.31	1.03	2.94	3.00

Interest Questions

Interest	School	Ν	P-Value	\mathbf{X}^2	Pre-Survey Mean	Post-Survey Mean
	А	135	0.54	2 16	3 41	3 43
	R	66	0.17	5.09	3.19	2 91
In1 - I look forward to	C D	282	0.17	2.02	3.01	2.91
science class in school	D D	202	0.40	2.94	3.61	2.94
	D E	17	0.14	0.03	3.03	3.55
		17	0.80	0.03	3.30	3.55
	A	134	0.56	2.08	2.92	2.86
In2 - I look forward	В	65	0.71	1.39	2.84	2.80
to math class in school	C	280	0.62	1.80	2.73	2.79
	D	54	0.24	4.19	3.17	3.31
	E	17	0.57	1.99	3.39	3.24
In3 - I would rather	А	134	0.12	5.83	3.36	3.25
solve a problem by	В	64	0.27	3.93	3.27	3.13
doing an experiment	С	281	0.38	3.05	2.87	2.90
than be told the	D	54	0.42	2.82	3.46	3.61
answer	E	17	0.94	0.01	3.83	3.82
In4 - More time	*A	133	0.01	10.47	3.63	3.64
should be spent on	В	66	0.54	2.16	3.46	3.33
hands-on projects in	С	277	0.63	1.75	3.23	3.20
science or technology	D	53	0.58	1.94	3.69	3.77
activities at school	Е	17	0.58	0.31	3.89	3.82
In5 - I would like to	А	135	0.82	0.93	2.71	2.64
(or already do) belong	B	65	0.68	1.50	2.71	2.37
to a science or	Č	282	0.59	1.90	2.19	2.37
technology activities	D	54	0.40	2.96	3.17	3.02
club	E	17	0.52	2.25	3.06	2.82
Inf I get hored when	2	17	0102	2.20	2100	2:02
I watch programs on	А	135	0.28	3.86	1.99	1.93
channels like	В	66	0.21	4.55	1.77	2.02
Discovery Channel	С	279	0.17	5.08	2.00	2.19
Animal Planet Nova	D	54	0.65	1.65	2.26	2.00
Mythbusters etc	E	17	0.26	3.97	1.50	1.76
nij tilo ustolis, etc.	Δ	134	0.90	0.58	2 51	2 42
In7 - I like to get	B	66	0.93	0.44	2.31	2.12
science books or	C	279	0.95	0.97	2.25	2.20
science experiments	D	54	0.38	3.07	2.05	2.69
kits as presents	F	17	0.58	2 46	2.75	2.05
	L 	122	0.40	2.40	2.00	2.12
	A	155	0.32	3.54	5.52 2.42	2.39
In8 - I like learning	ы С	280	0.47	2.33	J.4∠ 2 11	5.52 2.12
how things work		20U 51	0.31	0.20	3.11	3.12 2.46
	D F	34 17	0.90	1.24	2.00	5.40 2.92
	止 * ٨	125	0.31	11.54	1.09	2.02
In Q Saianas is tas	·A P	133	0.01	11.32	1.98	2.33
hand when it ince 15 too	р С	201	0.05	1.00	2.07	2.23
math		201	0.11	0.08	2.29	2.29
main	D F	3Z	0.85	0.80	2.1/ 1.79	2.10
	E	16	0.39	3.04	1./8	1.94
	A	134	0.13	5.59	1.78	2.01
In10 - Science is a	В	65	0.32	3.54	1.96	2.23
difficult subject	C	280	0.15	5.34	2.10	2.21
unneun subject	D	54	0.81	0.97	2.06	1.89
	-		o	0.0-	4	

Interest Questions cont.

	А	134	0.05	7.74	1.59	1.87
In11 - Doing	*B	66	0.01	10.66	1.66	2.18
experiments in science	С	278	0.30	3.68	1.89	1.92
class is frustrating	D	54	0.35	3.25	1.92	1.67
	Е	17	0.80	0.44	1.44	1.53
In12 - I feel	А	135	0.97	0.25	3.03	3.06
comfortable with	В	65	0.45	2.63	2.92	3.08
using a computer to	С	280	0.99	0.09	2.81	2.80
make graphs and	D	53	0.74	1.24	3.36	3.40
tables	Е	17	0.26	4.03	3.39	3.24
	А	135	0.28	3.86	3.35	3.26
In13 - I am interested	В	64	0.61	1.81	3.28	3.25
in learning more about	С	282	0.18	4.85	2.80	2.96
how things work	D	54	0.67	1.56	3.60	3.46
	Е	17	0.48	1.48	3.89	3.71
	*A	133	< 0.01	14.17	3.59	3.44
In 14 I like to loom	В	65	0.85	0.79	3.48	3.37
to use new technology	*C	280	0.02	10.16	2.92	3.16
to use new technology	D	54	0.48	2.47	3.75	3.59
	Е	17	0.48	1.45	3.72	3.47

*Significance: p-value less than or equal to 0.05

Attitude and Skills

Attitude and Skills	School	Ν	P-Value	X ²	Pre-Survey Mean	Post-Survey Mean
A1 - When I see a new	А	134	0.14	5.52	2.92	2.78
math problem, I can	В	61	0.64	1.70	2.85	2.85
use what I have	С	279	0.43	2.76	2.61	2.53
learned to solve the	D	54	0.73	1.29	2.92	2.96
problem	Е	17	0.69	1.47	3.06	3.06
A2 - I can use what I	*A	134	0.02	9.59	2.48	2.55
know to design and	В	60	0.06	7.35	2.24	2.57
build something	С	279	0.07	7.13	2.08	2.23
mechanical that works	D	54	0.60	1.89	2.81	2.87
mechanical that works	Е	17	0.86	0.31	3.06	2.94
A2 In the pativities I	А	134	0.44	2.68	2.89	2.81
A3 - III lab activities, I	В	61	0.73	1.28	2.81	2.85
learned to design a	С	278	0.48	2.47	2.49	2.42
solution	D	54	0.46	2.60	3.09	3.07
solution	E	17	0.31	2.34	3.17	3.47
1. Lean officiatively	А	133	0.15	5.35	2.85	2.85
load a team to design	В	61	0.08	6.74	2.72	2.95
and build a hands on	С	279	0.98	0.21	2.33	2.34
and build a mands-on	D	54	0.38	3.05	3.04	2.93
project	Е	17	0.86	0.31	3.06	3.12
A5 - I know where I	А	134	0.26	3.97	2.89	2.72
can find the	В	61	0.81	0.98	2.92	2.95
information that I need	С	279	0.31	3.62	2.60	2.49
to solve difficult	D	54	0.32	3.50	2.94	3.13
problems	E	17	0.97	0.06	3.22	3.24

Attitudes and Skills cont.

A6 Loop overlain	А	134	0.19	4.77	2.82	2.67
Ao - I can explain	В	61	0.17	5.06	2.56	2.85
finant of science to my	С	278	0.65	1.63	2.57	2.53
Intends to help them	D	54	0.24	4.24	2.79	2.94
understand	Е	17	0.71	1.37	2.94	2.94
	А	134	0.66	1.59	3.14	3.01
A7 Law act and 1	В	61	0.26	4.01	3.07	3.31
A/ - I can get good	С	279	0.06	7.23	2.91	2.92
grades in math	D	54	0.71	1.40	3.15	3.19
	Е	17	0.64	1.69	3.39	3.41
	А	132	0.73	1.30	3.35	3.25
	В	61	0.62	0.95	3.19	3.31
A8 - I can get good	С	278	0.36	3.24	3.00	3.04
grades in science	D	54	0.43	2.74	3.23	3.20
	Е	17	0.78	1.10	3.44	3.29

APPENDIX G ANALYSIS OF EVERY QUESTION/STATEMENT INCLUDED IN THE PRE-AND POST-SURVEY BY GENDER

					Pre-Survey	Post-Survey
Engineers	Gender	Ν	P-Value	X ²	Mean	Mean
E1 - Mainly work on machines and computers	Female *Male	251 309	0.06 0.03	5.56 6.75	2.32 2.46	2.17 2.29
E2 - Mainly work with other people to solve problems	*Female *Male	250 304	<0.01 <0.01	18.23 19.59	2.39 2.33	2.62 2.61
E3 - Work on things that help the world	Female Male	250 304	0.39 0.67	1.91 0.81	2.75 2.70	2.80 2.70
E4 - Can choose to do many different kinds of jobs	*Female *Male	250 304	<0.01 <0.01	15.49 14.30	2.52 2.45	2.71 2.68
E5 - Mainly work on things that have nothing to do with me	Female *Male	250 302	0.13 0.02	4.10 7.97	1.62 1.59	1.61 1.70
E6 - I don't know what engineers do	Female Male	242 293	0.94 0.54	0.12 1.22	1.58 1.45	1.55 1.39
E7 - Pursue a career in an engineering- related field?	Female Male	253 311	0.68 0.89	0.77 0.23	1.61 2.08	1.63 2.11
E8 - Do you think you want to be an engineer?	Female Male	253 311	0.46 0.92	1.56 0.16	1.45 1.97	1.50 1.99

Engineer Ouestions

Future Work Questions

					Pre-Survey	Post-Survey
Future Work	Gender	Ν	P-Value	X ²	Mean	Mean
F1 - Work that makes me think	Female Male	251 309	0.93 0.84	0.15 0.36	2.48 2.39	2.50 2.42
F2 - Work that allows me to make lots of money	*Female Male	252 308	0.05 0.78	6.06 0.51	2.52 2.55	2.39 2.54
F3 - Work that allows me to use math, computers, engineering, or science skills	Female Male	253 307	0.59 0.83	1.06 0.37	2.19 2.32	2.21 2.29
F4 - Work that allows me to tell other people what to do	Female Male	252 306	0.58 0.06	1.10 5.76	1.72 1.76	1.76 1.89
F5 - Work that allows me to help solve problems and create solutions	Female Male	252 307	0.81 0.95	0.43 0.10	2.37 2.36	2.37 2.36
F6 - Work that is fun to do	Female Male	253 309	0.50 0.55	1.39 1.20	2.77 2.73	2.78 2.78
F7 - Work that allows me to have time with family	Female Male	250 306	0.57 0.99	1.13 0.03	2.81 2.69	2.77 2.68
F8 - Work that allows me to help my community and/or society	Female Male	250 306	0.30 0.64	2.39 0.88	2.56 2.39	2.48 2.38
F9 - Work that makes people think highly of me	Female Male	249 306	0.27 0.78	2.63 0.49	2.26 2.29	2.16 2.31
F10 - Work that is satisfying to me	Female Male	250 305	0.21 0.59	3.08 1.06	2.82 2.74	2.75 2.74

Interest Questions

T 4 4		NT	D X 1	x 7?	Pre-Survey	Post-Survey
Interests	Gender	N	P-Value	X2	Mean	Mean
In1 - I look forward to	Female	253	0.62	1.76	3.14	3.06
science class in school	Male	308	0.21	4.53	3.26	3.16
In2 - I look forward	Female	251	0.44	2.68	2.78	2.92
to math class in school	Male	304	0.45	2.66	2.92	2.84
In3 - I would rather solve a problem by doing an experiment	Female	252	1.00	0.06	3.11	3.12
than be told the answer	Male	303	0.29	3.78	3.15	3.10
In4 - More time should be spent on						
hands-on projects in	Female	250	0.62	1.79	3.42	3.44
science or technology activities at school	Male	303	0.16	5.12	3.43	3.36
In5 - I would like to (or already do) belong	Famala	249	0.11	6 11	2.25	2 20
to a science or	Male	240	0.11	0.11	2.55	2.20
technology activities club	Wate	505	0.74	1.20	2.36	2.02
In6 - I get bored when I watch programs on						
channels like	Female	251	0.82	0.93	2.20	2.19
Discovery Channel,	*Male	304	0.05	7.62	1.79	1.97
Animal Planet, Nova,						
Mythbusters, etc.						
In7 - I like to get	F 1	240	0.00	0.01	2.10	0.10
science books or	Female	249	0.98	0.21	2.10	2.13
kits as presents	Male	304	0.62	1.79	2.38	2.33
	F 1	247	0.20	2.02	2 1 1	2.14
In8 - I like learning	Female	247	0.39	3.03	3.11	3.14
	Male	300	0.14	5.50	5.59	5.57
In9 - Science is too	Female	246	0.66	1.61	2.28	2.27
math	*Male	302	0.04	8.54	2.05	2.26
In10 - Science is a	Female	247	0.34	3 39	2.00	2.15
difficult subject	Male	304	0.11	5.94	1.97	2.11
In11 - Doing			0.47			1.0.6
experiments in science	Female	245	0.47	2.52	1.75	1.86
class is frustrating	Male	303	0.26	3.98	1./9	1.94
In12 - I feel						
comfortable with	Female	247	0.28	3.81	2.86	2.89
using a computer to	Male	303	0.56	2.06	3.02	3.03
make graphs and		200	0.00	2.00	2.02	2.00
tables						
In13 - I am interested	Female	245	0.49	2.41	2.90	3.01
how things work	Male	303	0.69	1.47	3.28	3.25
In14 - I like to learn	*Female	246	< 0.01	13.22	3.00	3.1/
to use new technology	Male	303	0.90	0.60	3.09	3 44
		202	0.70	0.00	5.11	2.11

Attitude and Skills

Attitude and Skills	Gender	N	P-Value	X ²	Pre-Survey Mean	Post-Survey Mean
A1 - When I see a new math problem, I can use what I have learned to solve the problem	Female Male	250 305	0.87 0.25	0.71 4.13	2.70 2.81	2.69 2.69
A2 - I can use what I know to design and build something mechanical that works	Female Male	250 304	0.01 0.04	10.50 8.32	2.05 2.50	2.20 2.63
A3 - In lab activities, I can use what I have learned to design a solution	Female Male	251 306	0.45 0.51	2.63 2.32	2.55 2.84	2.55 2.76
A4 - I can effectively lead a team to design and build a hands-on project	Female Male	251 304	0.34 0.19	3.34 4.79	2.59 2.61	2.56 2.66
A5 - I know where I can find the information that I need to solve difficult problems	Female Male	250 304	0.17 <0.01	4.97 13.03	2.61 2.89	2.72 2.66
A6 - I can explain math or science to my friends to help them understand	Female Male	251 303	0.50 0.98	2.37 0.20	2.66 2.66	2.67 2.64
A7 - I can get good grades in math	Female Male	251 305	0.45 0.21	2.64 4.51	3.07 2.99	3.15 2.93
A8 - I can get good grades in science	Female Male	250 304	0.68 0.85	1.53 0.82	3.19 3.11	3.22 3.08

BIBLIOGRAPHY

AWE instrument user guide - pre-college surveys. (Accessed 2011). Retrieved from Assessing Women and Men in Engineering (AWE):

https://www.engr.psu.edu/awe/secured/ director/AWEhome.aspx.

- Bandura, A. (1977). Self-Efficacy: Toward a Unifying Theory of Behavioral Change. Psychological Reveiw, 191 - 215.
- Clarke, J., & Dede, C. (2009). Design for Scalability: A Case Study of the River City Curriculum. Journal of Science Education Technology, 353-365.
- Coburn, C. (2003). Rethinking Scale: Moving Beyond Numbers to Deep and Lasting Change. Educational Researcher, 3-12.
- Committee on Public Understanding of Engineering Messages and National Academy of Engineering. (2008). Changing the Conversation: Messages for Improving Public Understanding of Engineering. National Academies Press.
- Cunningham, C., & Lachapelle, C. (2010). The Impact of Engineering is Elementary (EiE) on Students' Attitudes towards Engineering and Science. American Society for Engineering Education.
- Cunningham, C., & Lachapelle, C. (2012). Engaging ALL Students in Engineering. American Society for Engineering Education.

- Davis, C., Yeary, M., & Sluss, J. (2012). Reversing the Trend of Engineering Enrollment Declines with Innovative Outreach, Recruiting, and Retention Programs. IEEE, 55(2), 157-163.
- Dawes, L., & Rasmussen, G. (2006). Activity and engagement Keys in connecting engineering with secondary school students.
- Edelson, D. (2001). Learning-for-use: A framework for the design of technologysupported inquiry activities. Journal of Research in Science Teaching, 38(3), 355-385.
- Eniola-Adefeso, O. (Fall 2010). Engaging K-12 Students in the Engineering Classroom: A Creative Use of Undergraduate Self-Directed Projects. Chemical Engineering Education, 280 - 286.
- Ertmer, P. e. (2007). Using Peer Feedback to Enhance the Quality of Student Online Postings: An Exploratory Study. Journal of Computer Mediated Communication, 412-433.
- Fantz, T., Siller, T., & DeMiranda, M. (2011). Pre-Collegiate Factors Influencing the Self-Efficacy of Engineering Students. Journal of Engineering Education, 604-623.
- Franklin, B. (n.d.).
- Fussell Policastro, E. (2009, April). Engineers can change the world. (InTech) Retrieved January 03, 2014, from http://www.isa.org/InTechTemplate.cfm? template=/ContentManagement/ ContentDisplay.cfm&ContentID=75381
- Genalo, L., Bruning, M., & Adams, B. (2000). Creating a K-12 engineering educational outreach center. American Society for Engineering Education. Washington, DC.

- Grade 5 Measurement & Data. (2014). (Common Core State Standards Initiative -Preparing America's Students for College & Career) Retrieved March 22, 2014, from http://www.corestandards.org/Math/Content/5/MD/
- Habash, R. W., & Suurtamm, C. (2010). Engaging High School and EngineeringStudents: A Multifaceted Outreach Program Based on a Mechatronics Platform.IEEE, 53(1), 136-143.
- Hill, C., Corbett, C., & St. Rose, A. (2010). Why So Few? Women in Science,Technology, Engineering, and Mathematics. Washington, DC: AmericanAssociation of University Women.
- Hmelo-Silver, C., Golan Duncan, R., & Chinn, C. (2007). Scaffolding and Achievement in Problem-Based and Inquiry Learning: A Response to Kirschner, Sweller, adn Clark (2006). Educational Psychologist, 42(2), 99-107.
- Jeffers, A., Safferman, A., & Safferman, S. (2004). Understanding K-12 Engineering Outreach Programs. Journal of Professional Issues In Engineering Education And Practice, 130, 95-108.
- Keengwe, J., & Onchwari, G. (2008). Computer Technology Integration and Student Learning: Barriers and Promise. Journal of Science Education Technology, 560-565.
- Kimrey, J. (2013, July 26). Engineering futures are always bright. (Chron) Retrieved January 03, 2013, from http://www.chron.com/jobs/article/Engineering-futuresare-always-bright-4688904.php

- Knight, M., & Cunningham, C. (2004). Draw an Engineer Test (DAET): Development of a Tool to Investigate Students Ideas about Engineers and Engineering. ASEE Annual Conference and Exposition.
- Lachapelle, C., Phadnis, P., Hertel, J., & Cunningham, C. (2012). What is Engineering? A Survey of Elementary Students.
- Marshall, H., McClymont, L., & Joyce, L. (2007). Public Attitudes to and Perceptions of Engineering and Engineers 2007.
- Marzano, R., & Heflebower, T. (2012). Teaching & Assessing 21st Century Skills. Bloomington, INI: Marzano Research Laboratory.
- Mooney, M., & Laubach, T. (2002). Adventure Engineering: A Design Centered, Inquiry Based Approach to Middle Grade Science and Mathematics Education. Journal of Engineering Education, 91(3), 309-318.
- Ololube, N. P. (2009). Instructional technology in higher education: A case of selected universities in the Niger Delta. Retrieved from The Hong Kong Institute of Education: http://www.ied.edu.hk/apfslt/v10_issue2/ololube/index.htm
- Patterson, K., Grenny, J., Maxfield, D., McMillan, R., & Switzler, A. (2007). The Power to Change Anything INFLUENCER. New York, NY: McGraw-Hill Professional Publishing.
- Price, A. (2000). Recommendations for establishing small scale K, 1 outreach. American Society for Engineering Education. Washington, DC.
- Safar, A., & AlKhezzi, F. (2013). Beyond Computer Literacy: Technology Integration and Curriculum Tranformation. College Student Journal, 614-626.

Science and Engineering Degrees: 1966-2010. (2013, June). Retrieved from National Science Foundation:

http://www.nsf.gov/statistics/nsf13327/content.cfm?pub_id=4266&id=2

[Accessed 03 January 2014]

Shute, V. (2008). Focus on Formative Feedback. American Educational Research Association.

(2014). The Condition of STEM 2013 National. ACT.

- The President's Council of Advisors on Science and Technology. (2010). Prepare and Inspire: K-12 Education in Science, Technology, Engineering, and Math (STEM) For America's Future.
- Wagner, T. (2008). The Global Achievement gap: Why even our best schools don't teach the new survival skills our children need - and what we need to do about it. New York, NY: Basic Books.

Wulf, W. (2006). Diversity in Engineering. Women in Engineering ProActive Network.

- Young, K. (2007). Recruiting Future Engineers through Effective Guest Speaking in Elementary School Classrooms. Meeting the Growing Demand for Engineers and Their Educators 2010-2020 International Summit. Munich.
- Yurtseven, H. O. (2002). How Does the Image of Engineering Affect Student Recruitment and Retention? A Perspective from the USA*. Global Journal of Engineering Education, 6, 17-23.